University of Wisconsin Milwaukee

Theses and Dissertations

August 2024

THE IMPACT OF PRENATAL EXPOSURE TO ENDOCRINE DISRUPTING CHEMICALS ON LANGUAGE DEVELOPMENT TRAJECTORIES

Justin Yu University of Wisconsin-Milwaukee

Follow this and additional works at: https://dc.uwm.edu/etd

Part of the Epidemiology Commons

Recommended Citation

Yu, Justin, "THE IMPACT OF PRENATAL EXPOSURE TO ENDOCRINE DISRUPTING CHEMICALS ON LANGUAGE DEVELOPMENT TRAJECTORIES" (2024). *Theses and Dissertations*. 3643. https://dc.uwm.edu/etd/3643

This Dissertation is brought to you for free and open access by UWM Digital Commons. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of UWM Digital Commons. For more information, please contact scholarlycommunicationteam-group@uwm.edu.

THE IMPACT OF PRENATAL EXPOSURE TO ENDOCRINE DISRUPTING CHEMICALS ON LANGUAGE DEVELOPMENT TRAJECTORIES

by

Justin Yu

A Dissertation Submitted in

Partial Fulfillment of the

Requirements for the Degree of

Doctor of Philosophy

in Environmental Health Sciences

at

The University of Wisconsin-Milwaukee

August 2024

ABSTRACT

THE IMPACT OF PRENATAL EXPOSURE TO ENDOCRINE DISRUPTING CHEMICALS ON LANGUAGE DEVELOPMENT TRAJECTORIES

by

Justin Yu

The University of Wisconsin-Milwaukee, 2024 Under the Supervision of Professor Amy Kalkbrenner

Environmental contaminants that can impact and disrupt the endocrine system, known as endocrine disrupting chemicals (EDCs), are of growing concern, found in water, air, and common household goods. The disruption to the endocrine system can have severe impacts on the development of the fetus, particularly with regards to neurodevelopment. Understanding and elucidating the effects of prenatal exposure to EDCs on neurodevelopment will help us develop policies and interventions that minimize exposure and risk. In this research, we examined the effects of prenatal exposure to EDCs and their impact on language development trajectories using two enhanced risk autism cohorts. Chapter 1 introduces the concepts of neurodevelopment, endocrine disruption, the EDCs we intend to examine, and the statistical methods we intend to use in this research. Chapter 2 examines the effect that autism diagnostic status has on language development trajectories, with the hypothesis that those with a diagnosis of autism or that present with sub-clinical symptoms are at greater risk of abnormal development. Our results indicated that children with a diagnosis of autism, or who present with sub-clinical symptoms are at a greater risk of falling into an abnormal

ii

language trajectory. Chapter 3 examines the effects of prenatal exposure to phthalates on language development trajectories, with the hypothesis that increased exposure results in greater risk of abnormal development. Our results indicated that low molecular weight phthalates tended to increase risk, while high molecular weight phthalate metabolites tended to decrease risk, though only a few reached statistical significance. Chapter 4 examines the effects of prenatal exposure to air toxics on language development trajectories, with the hypothesis that increased exposure will lead to greater risk of abnormal development. Our results indicated that nearly all air toxics did not have significant effect on risk of abnormal language development, with only acetaldehyde showing a decreased risk. Chapter 5 discusses the previous chapters and the implications of our findings. This research highlights the need for further research in a larger and more representative population, and that the effects of EDCs need to be more thoroughly explored to better elucidate their effects on neurodevelopment.

TABLE OF CONTENTS

Abstract	
List of Figures	
List of Tables	
Abbreviations	
Acknowledgements	
Chapter 1 – General Introduction	1
Introduction	2
Language Development	3
The Endocrine System	6
Phthalates	8
Air Toxics	11
Trajectory Analysis	13
Chapter 2 – Language Trajectories in Siblings of Children with Autism	14
Abstract	16
Introduction	18
Methods	20
Results	24
Discussion	26
Table and Figures	33
Chapter 3 – The Impact of Prenatal Phthalate Exposure on	
Language Development Trajectories in Siblings of Children with Autism	62
Abstract	63
Introduction	65
Methods	69
Results	75

Discussion	77
Tables and Figures	88
Chapter 4 – The Impact of Prenatal Air Toxic Exposure on	
Language Development Trajectories in Siblings of Children with Autism	154
Abstract	155
Introduction	157
Methods	160
Results	165
Discussion	167
Tables and Figures	173
Chapter 5 – Discussion, Conclusions, and Future Directions	208
Mixture Analysis	209
Use of High/Enhanced Risk Cohorts	210
Use of Trajectory Analysis (Latent Class Growth Analysis)	211
Sex Differences in the Effect of Phthalates	212
Discussion on Environmental Exposures	213
Future Examination of Language Development as an Outcome	217
Conclusions	218
References	220
Appendix	245

LIST OF FIGURES

Figure 2.1 Expressive Language Trajectories	56
Figure 2.2 Receptive Language Trajectories	57
Figure 2.3 Proportions of trajectory class assignment for expressive language	58
Figure 2.4 Proportions of trajectory class assignment for receptive language	59
Figure 2.5: Distributions of MSEL Expressive and Receptive Language	
Raw Scores at 6, 12, 24, and 36 Months	60
Figure 2.6: Scatterplots of MSEL Expressive vs Receptive Language	
Raw Scores at Each Time of Data Collection	61
Figure 3.1 Two-Step Latent Class Growth Analysis Plot for	
Expressive Language	122
Figure 3.2 Two-Step Latent Class Growth Analysis Plot for	
Receptive Language	123
Figure 3.3 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for mBP2	124
Figure 3.4 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for mBzP2	125
Figure 3.5 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for mCNP	126
Figure 3.6 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for mCOP	127
Figure 3.7 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for mCPP	128
Figure 3.8 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for mECPP	129
Figure 3.9 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for mEHHP	130
Figure 3.10 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for mEHP	131

Figure 3.11 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for mEOHP	132
Figure 3.12 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for mEP2	133
Figure 3.13 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for MHBP	134
Figure 3.14 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for MHiBP	135
Figure 3.15 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for miBP	136
Figure 3.16 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for mNP2	137
Figure 3.17 One-Step Latent Class Growth Analysis Plot for	
Expressive Language for DEHP	138
Figure 3.18 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for mBP2	139
Figure 3.19 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for mBzP2	140
Figure 3.20 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for mCNP	141
Figure 3.21 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for mCOP	142
Figure 3.22 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for mCPP	143
Figure 3.23 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for mECPP	144
Figure 3.24 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for mEHHP	145
Figure 3.25 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for mEHP	146

Figure 3.26 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for mEOHP	147
Figure 3.27 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for mEP2	148
Figure 3.28 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for MHBP	149
Figure 3.29 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for MHiBP	150
Figure 3.30 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for miBP	151
Figure 3.31 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for mNP2	152
Figure 3.32 One-Step Latent Class Growth Analysis Plot for	
Receptive Language for DEHP	153
Figure 4.1: Language Development Trajectories from Latent Class	
Growth Analysis for Expressive Language	198
Figure 4.2: Language Development Trajectories from Latent Class	
Growth Analysis for Receptive Language	199
Figure 4.3: Cluster Dendrogram for Air Toxics	100
Figure 4.4: Spearman Correlation between Air Toxics Heat Map	201
Figure 4.5: Bayesian Kernel Machine Regression Overall Risk	
– Expressive Language	202
Figure 4.6: Bayesian Kernel Machine Regression Single Exposure Effects	
– Expressive Language	203
Figure 4.7: Bayesian Kernel Machine Regression Interactive Effects	
- Expressive Language	204
Figure 4.8: Bayesian Kernel Machine Regression Overall Risk	
- Receptive Language	205
Figure 4.9: Bayesian Kernel Machine Regression Single Exposure Effects	
 Receptive Language 	206

Figure 4.10: Bayesian Kernel Machine Regression Interactive Effects

- Receptive Language

LIST OF TABLES

Table 2.1 Characteristics of Participants in Early Autism Risk Longitudinal	
Investigation (EARLI) and Markers of Autism Risk in Babies – Learning	
Early Signs (MARBLES) by Study Neurodevelopmental Classification	33
Table 2.2 Relative Risk Ratios of Language Development Trajectory	
by Neurodevelopmental Classification, Participants in Early Autism	
Risk Longitudinal Investigation (EARLI) and Markers of Autism Risk	
in Babies – Learning Early Signs (MARBLES)	38
Table 2.3: Characteristics of Participants Included and Excluded from Analyses	39
Table 2.4: Characteristics of Included Study Sample Before and After	
Imputing Missing Data	44
Table 2.5: Measures of Model Fit for Expressive Language	
Latent Class Growth Analysis	47
Table 2.6: Measures of Model Fit for Receptive Language	
Latent Class Growth Analysis	48
Table 2.7: Characteristics of Study Population Stratified by	
Assigned Expressive Language Trajectory Classes	49
Table 2.8: Characteristics of Study Population Stratified by	
Assigned Receptive Language Trajectory Classes	52
Table 2.9: Cross-classification of study population by	
assigned language trajectories	54
Table 2.10: Total number of children falling into trajectories based on Neurodevelopmental Classification	55
Table 3.1 Mean and Standard Deviation of 36 Month Mullen Scores	
for Included Population by Child Characteristics	88
Table 3.2 Comparison of characteristics of included versus excluded individuals	; 91
Table 3.3 Comparison of Imputed versus Non-Imputed Covariates	95
Table 3.4: Relative Risk Ratios and 95% Confidence Intervals for	
Phthalate Metabolites and Expressive and Receptive Language Trajectories	99

Table 3.5 Measures of Model Fit for Expressive Language Latent Class	
Growth Analysis, Two-Step Model	101
Table 3.6 Measures of Model Fit for Receptive Language Latent Class	
Growth Analysis, Two-Step Model	102
Table 3.7 Measure of Model Fit for Expressive Language Latent Class	
Growth Analysis, One-Step Model	103
Table 3.8 Measure of Model Fit for Receptive Language Latent Class	
Growth Analysis, One-Step Model	107
Table 3.9 Comparison of Class Assignment between One-Step and	
Two-Step Models for Expressive Language	111
Table 3.10 Comparison of Class Assignment between One-Step and	
Two-Step Models for Receptive Language	114
Table 3.11 Sex Modification for Expressive Language	117
Table 3.12 Sex Modification for Receptive Language	119
Table 3.13 Comparison of Urinary Phthalate Concentrations	
between NHANES and Included Population	121
Table 4.1 Mean and Standard Deviation of 36 Month Mullen Scores	
for Included Population by Child Characteristics	173
Table 4.2 Relative Risk Ratios for each IQR increase of Air Toxics	176
Table 4.3: False Discovery Rate Adjusted P-Values	178
Table 4.4: Posterior Inclusion Probabilities for Expressive Language	182
Table 4.5: Posterior Inclusion Probabilities for Receptive Language	185
Table 4.6: Air Toxics with Evidence of Endocrine Disrupting Activity	188
Table 4.7: Comparison of characteristics of included versus excluded individuals	s189
Table 4.8: Comparison of Imputed versus Non-Imputed Covariates	193
Table 4.9: Comparison of Transformation of Air Toxics	197

LIST OF ABBREVIATIONS

ADHD	Attention Deficit Hyperactive Disorder
ADOS	Autism Diagnostic Observation Schedule
AIC	Akaike Information Criteria
ASD	Autism Spectrum Disorder
BIC	Bavesian Information Criteria
BKMR	Bayesian Kernel Machine Regression
BzBP	Butyl benzyl phthalate
CDC	Centers for Disease Control and Prevention
DAG	Directed Acyclic Graph
DBP	Di-n-butyl phthalate
DEP	Di-ethyl phthalate
DEHP	Di(2-ehtylhexyl) phthalate
DiBP	Di-isobutyl phthalate
DiDP	Di-isodecyl phthalate
DiNP	Di-isononyl phthalate
DOP	Dioctvl phthalate. Di-n-octvl phthalate
DSM-V	Diagnostic and Statistical Manual of Mental Disorders. Fifth Edition
EARLI	Early Autism Risk Longitudinal Investigation
EDC	Endocrine Disrupting Chemical/Compound
EPA	Environmental Protection Agency
ICL	Integrated Complete Likelihood
LCGA	Latent Class Growth Analysis
LI	Language Impairment
MARBLES	Markers of Autism Risk in Babies – Learning Early Signs
mBP2	Mono-n-butyl phthalate
mBzP2	Monobenzylphthalate
mCNP	Mono-carboxyisononyl phthalate
mCOP	Mono-carboxyisooctyl phthalate
mCPP	Mono(3-carboxypropyl) phthalate
mEP2	Monoethyl phthalate
mEHP	Mono(2-ethylhexyl) phthalate
mECPP	Mono(2-ethyl-5-carboxypentyl) phthalate
mEHHP	Mono(2-ethyl-5-hydroxyhexyl) phthalate
mEOHP	Mono(2-ethyl-5-oxohexyl) phthalate
MHBP	Monohydroxy-n-butyl phthalate
MHiBP	Monohydroxy-isobutyl phthalate
miBP	Mono-isobutyl phthalate
mNP2	Mono-isononyl phthalate
MSEL	Mullen Scales of Early Learning
NATA	National Air Toxics Assessment
Non-TD	Non-Typically Developing
RRR	Relative Risk Ratio
SABIC	Sample Size Adjusted Bayesian Information Criteria
TD	Typically Developing

ACKNOWLEDGEMENTS

I would like to thank several individuals who have provided advice, support, and mentorship throughout the duration of my degree. Thank you to my advisor Dr. Amy Kalkbrenner and my committee members Dr. Bonita Klein-Tasman, Dr. Kurt Svoboda, Dr. Cheng Zheng, and Dr. Heather Volk for their advice, support, and guidance throughout the duration of this dissertation. Without your help and support, I do not think I'd be able to get to this point in my academic career.

Thank you to the members of Kalkbrenner Labs, Tara Jensen, Victor Garcia Florez, and Mozhgan Nagi. The discussions we had during our weekly lab sessions, along with your feedback and support have proven invaluable to the writing of this dissertation.

Thank you to the staff and faculty of Zilber, who ignited a passion for teaching and public health within me. Thank you for all the administrative and technical support throughout the years, without which I'd likely still be lost in reams of paperwork and confusion.

Thank you to my friends, family, and loved ones, especially to Nathan, Samantha, and Cobie. You kept my spirits high during my lowest moments, and provided support and care when I needed it most. The nights spent together in laughter and fun kept my motivation and morale high, and I certainly wouldn't have made it this far with you all.

Chapter 1 – General Introduction

I. Introduction

The establishment of language is an important milestone in early childhood development, as it plays a role in almost every aspect of daily life. The effects of delays and impairments in language development may continue into adulthood, and include atypical social and emotional development, poor academic performance, and increased risk for diagnosis with neurodevelopmental disorders (1–7). There are many factors that can influence language development, including exposure to certain environmental pollutants. It is well known and documented that exposure to environmental chemicals during pregnancy can lead to abnormal neurodevelopment. Understanding the relationship between language development, its overall trajectories, and prenatal exposure to environmental chemicals, particularly within high-risk populations, may provide additional insight in identifying specific environmental hazards that may negatively impact language development. Once identified, targeted interventions can be implemented by policy makers in order to better mitigate the impact of these environmental exposures.

This research focuses on prenatal exposure to chemicals that act as endocrine disruptors (EDCs), which may play a role in fetal neurodevelopment. Exposure to these chemicals may negatively affect fetal neurodevelopment, which may result in abnormal language development. Rather than focusing on a single time point outcome, this research aims to examine language development over a longitudinal time period. By developing and using these language development trajectories as our outcome of interest, we are able to perform a more nuanced examination on the effects of EDCs on language development and gain several advantages over more traditional

epidemiological methods that examine single time point outcomes. By using multiple measures of language to develop these trajectories, we are able to improve our measurement of language development, reducing the chance of outcome misclassification. In addition, statistical analysis of trajectory groups allows for the examination of both within and between group variation, allowing researchers to identify risk factors unique to a specific trajectory class. This, in turn, may allow for more targeted interventions for those who fall within that trajectory class.

II. Language Development

The establishment of language is an important milestone in early childhood development. Delays or impairment of language development may lead to adverse effects later in life. The Diagnostic and Statistical Manual of Mental Disorders, Fifth Edition (DSM-V) lists the following as criteria for diagnosis of language disorder: persistent difficulties in the acquisition and use of language across modalities due to deficits in comprehension or production, language abilities that are substantially and quantifiably below those expected for age, onset during an early developmental period, and the difficulties are not attributable to hearing or other sensory impairment, motor dysfunction, or another medical or neurological condition and are not better explained by intellectual disability or global developmental delay (8). Language impairment (LI) has also been linked with and often co-occurs with several neurodevelopmental disorders, including attention deficit hyperactive disorder (ADHD) and autism (9,10,11, p.). As a result, understanding how, when, and why language impairment occur are important in order to prevent detrimental outcomes in the future.

Due to the multi-faceted aspects of language development, the suspected causes of LI, and the impacts of those causes, are varied; causes that result in a smaller vocabulary may not be the same causes that cause delays in verbal communications, for example. From a biological perspective, there have been several studies that have indicated that the basal ganglia may play an important role in language development (12,13). In addition, a region of the brain known as Broca's area or Broca's region has widely been regarded as an important structure for language development (14–16). Disruption of the development of these areas may result in language impairment. Thus, anything that may impact neurodevelopment has a possibility of directly impacting language development.

Although the study of environmental pollutants and LI is currently limited, some environmental pollutants have been linked to increased risk of LI including polychlorinated biphenyls (17), certain heavy metals (18), air pollutants (19,20), and polycyclic aromatic hydrocarbons (21,22). Many other environmental pollutants (e.g. air toxics, pesticides, and perfluoroalkyl substances) are likely candidates for study because they have been linked to neurodevelopmental disorders generally, such as autism (23–25), but have not yet been explicitly explored for links with language development. Beyond biological risk factors, consistently supported demographic and behavioral risk factors include: lower maternal education, having a family history of LI, higher birth order, male sex, preterm birth, lower 5 minute Apgar score, maternal smoking, and maternal alcohol use (26–30).

Human language development is not static but unfolds over time, following different trajectories that vary from person to person. Even among children without

diagnoses of developmental disorders or language impairment, subtle differences in language development trajectories can be detected and measured. These differences result in several trajectories of language development (31–33). However, impaired language development follows distinct and different trajectories than typical development. Studies of individuals with language impairment have found several different trajectories (34,35, p.,36,37). In addition, a large portion of existing research that examines language development trajectories has been in the context of autism spectrum disorder (ASD). For example, Landa et al. identified four groups using latent class growth analysis based on scores from the Mullen Scales of Early Learning on 204 siblings of children with autism (38). For the classes identified in these studies, they can be generalized into two broad categories: typical development and delayed or slowed improvement. Typical development (TD) is the trajectory defined by control individuals; that is, individuals with no clinical diagnosis of autism or other neurodevelopmental disorder, including LI. However, the definition of TD varies by study due to the nature of their cohorts, along with the method used to measure language. Those with delayed or slowed improvement may show improvement but either at a slower pace or begin at a less developed point than TD individuals, though they may still develop at an equal pace. When compared to children without a diagnosis of autism, children with autism tend to score lower on psychometric tests and have trajectories that are significantly different (39,40), though the general trend of the trajectory may be similar. However, both studies did not examine a third group: individuals who show "sub-clinical" symptoms. A portion of this research dissertation includes this third group, which we call

non-typically developing (non-TD), for trajectory comparison, which may provide more insight into possible trajectory differences between these diagnostic groups.

III. The Endocrine System

The endocrine system refers to the system of organs within the body that produce hormones, chemical messengers that regulate bodily functions, and the effect these hormones have on the body. These organs include the thyroid, gonads, and pituitary glands, which produce hormones such as sex hormones (e.g. estrogen), thyroid hormones (e.g. T3, T4), and peptide hormones (e.g. insulin, oxytocin). Because hormones are responsible for cell-to-cell communication during development, disruption of the endocrine system during critical developmental windows can lead to severe detriments in development. Pollutants that interfere with proper endocrine functioning are called endocrine disrupting compounds (EDCs), and are commonly linked to many different developmental outcomes, including neurodevelopmental deficits. For example, thyroid hormones play important roles during fetal neurodevelopment, acting on processes such as cell migration and differentiation, myelination, and synaptogenesis (41,42). As a result, prenatal thyroid disruption can lead to severe neurodevelopmental defects, including cretinism, lower IQ, and an increased risk of developing neurodevelopmental disorders (43–47).

More than 1800 EDCs have been identified by the US Food and Drug Administration (FDA) that disrupt at least one endocrine pathway (48). Some examples of common EDCs include bisphenol A, organophosphate pesticides, phthalates, polybrominated diphenyl ethers (PBDEs), polycyclic aromatic hydrocarbons (PAHs), polychlorinated biphenyls (PCBs), and per/polyfluoroalkyl substances (PFAS).

Disruption of the endocrine system can occur through several different methods, including disrupting hormone metabolism by antagonizing or activating hormone receptors, modulating receptor coactivators, and influencing DNA methylation (49). EDCs can be found in various mediums, including groundwater, air pollution, and household products, and have been linked with many different health effects, including obesity, diabetes, reproductive health, certain types of cancer, and neurodevelopment disorders (50–53). Because of the large number of EDCs, combined with the multiple number of possible exposure routes, there has been a large emphasis on identifying exposure-outcome relationships and promoting the removal of EDCs with known health effects from general use.

As mentioned previously, hormones play a key role in infant development, particularly with neurodevelopment. Many EDCs have documented associations with neurodevelopmental disorders such as autism and ADHD. Polychlorinated biphenyls (PCBs) have been associated with ADHD (54), along with more general neurodevelopmental impairments, such as lower IQ, mental development, and psychomotor development (17,55). Polybrominated diphenyl ethers (PBDEs) have consistently shown a connection with neurodevelopment and neurotoxicity (56–58), along with associations with certain neurodevelopmental detriments in children related to attention, fine motor coordination, and cognition (59,60). Polycyclic aromatic hydrocarbons (PAHs) have been associated with adverse neurodevelopmental effects, such as cognitive delays and autism (21,61–63). Certain pesticides have been linked to lower IQ, lower motor speed, lower motor coordination, lower visuospatial performance, worse visual memory, and an increased risk for ADHD (64–71).

While research into the role of hormones on language development has been sparser, there is still evidence that both thyroid and sex hormones may play a role in language development. Postnatal treatment of thyroid deficiencies led to improvements in language deficits in children with hypothyroidism (72). Quast et al. found that sex hormone levels were linked with infant babbling, an important marker of articulatory skills and vocal development (73). Given the important role of hormones in neurodevelopment, it is reasonable to suspect that some EDCs that impact neurodevelopment may also interfere with language development. Indeed, some studies have found that certain EDCs have been associated with delays in language development. Bornehag et al. found that certain phthalates were associated with language delays, and both carbamate and organochlorine pesticides have been found to decrease language abilities in children (74–76).

IV. Phthalates

Phthalates, or phthalate esters, are a class of chemicals that are primarily used as plasticizers. They are used in a wide variety of products, including cosmetics, food packaging, cleaning materials, and pharmaceuticals. Phthalates have been shown to be endocrine disruptors through both animal and human studies, affecting estrogen and thyroid hormone pathways (77–79). These effects have been linked to several health effects, including neurotoxicity (80–82), hepatotoxicity (82,83), metabolic syndromes (84,85), and reproductive effects (83,86). Phthalates impair neurodevelopment via several pathophysiologies, which are varied, due to the wide range and types of phthalates and the possibility of different biological effects from different phthalate compounds. Several of the mechanisms of action of phthalates that have been

observed in animal studies include cell apoptosis (80,87), affecting neurotransmitter pathways (80,88), changing gene expression (87,89), and altering hormone levels (89,90). These effects have been connected to potential neurodevelopmental issues, which indicates that there may be more than one mechanism of action taking place that is resulting in an association with neurodevelopmental disorders and delays.

Currently, there is a significant amount of literature that indicates that phthalates have a negative impact on neurodevelopment. Phthalates have been found to be associated with neurodevelopmental disorders such as autism and ADHD (78,91,92) along with deficits in other neurological domains such as cognition (93,94), motor effects (95), and behavioral outcomes (96–98). Disruption of functionality and plasticity in the hippocampus, alterations in gene regulation and DNA damage, and disruption of several hormonal pathways are possible explanations for the neurodevelopmental effects of phthalates (99–102). There have been several studies that examined prenatal phthalate exposure and its effects on language development, four finding associations with delays in language development (74,103–105), and four finding no association (97,106–108). These studies generally excelled in prospective designs where phthalates were primarily measured via biomarkers. They used psychometric tests to measure language development, but only examined a single time point during development. Although the reasons for their differing results are not yet wellestablished, possible reasons for the contradicting findings include exposure misclassification, precision or sampling error (particularly for those studies with smaller sample sizes), and confounding. In addition, there were little to no examinations on multi-phthalate models, which are important to consider given that exposure to

phthalates often occurs in a mixture setting. One weakness that has been noted in prior studies involve the possibility of exposure misclassification, perhaps due to single measurements of phthalate exposures (97,107,108), which may result in no association being observed between phthalate exposure and delayed language development.

It is also important to note that when discussing phthalates and their possible association with language development, individual phthalates may have differing results, which may help to explain some of the inconsistencies. When we reviewed multiple studies that covered several phthalates and their metabolites, we found that results were inconsistent across phthalates, with some studies showing some phthalates with an increased risk for abnormal language development with others showing protective or null effects (74,97,103–108). Part of this difference may be in part due to the chemical composition of each phthalate and its metabolites, since it is thought that phthalate toxicity may be due in part to their overall molecular weight. Phthalates are classified into low and high molecular weight classes. The difference in weight may account for the differences between each individual phthalate ester. Each ester has different solubilities, with lower weight phthalates being more volatile in their pure state, but have low volatility in aqueous solutions, while high weight esters are hydrophobic in comparison, whose solubility is dependent on the length of the aryl chains of the phthalate (109).

Currently, evidence points to a possible sex difference in phthalates' various effects (110–112); this is to be expected as many endocrine disruptors impact female health differently than male health. This may be due to phthalate's impact on different hormonal pathways, some of which may be active during different periods of

development between the different sexes. Prior studies have found that sex differences vary based on phthalates, with two studies showing greater risk for boys (74,105) and one showing greater risk for girls (113). This may also be further compounded by the effect sex may have on certain neurodevelopmental outcomes. A literature review performed by Etchell et al., for instance, found inconsistent evidence for sex differences in brain function and structure with respect to language development (114). There is also a sex difference in prevalence in some neurodevelopmental disorders, such as autism and ADHD, with males being diagnosed at a higher rate than females. This may be due to interactions between sex and other factors, such as genetics, hormones, and environment (115).

V. Air Toxics

Air toxics, as defined by the United States Environmental Protection Agency (EPA), are toxic or hazardous air pollutants that cause or may cause serious health effects such as cancer, reproductive effects, and/or adverse environmental and ecological effects (116,117). Because of this broad definition, there are several classes of chemicals that are considered to be air toxics; the Clean Air Act identifies a total of 187 air toxics the EPA is required to control, though there are many more hazardous air pollutant (117). Pesticides, polychlorinated biphenyls (PCBs), particulate matter (e.g. PM₁₀ and PM_{2.5}), polycyclic aromatic hydrocarbons (PAHs), various volatile organic compounds (VOCs, e.g. benzene, formaldehyde), and heavy metals (e.g. lead, mercury) are just a few examples. With the wide variety of chemicals, air toxics have been linked to a vast array of health effects, including neurodevelopmental disorders (54,57,118–120), cognitive decline (121), and cancer (122).

Of the numerous air toxics, PCBs, phthalates, PAHs, polybrominated diphenyl ethers (PBDEs), and certain heavy metals are a few examples with known endocrine disrupting effects (61,123–142). The mechanisms by which these air toxics cause these effects are varied, with some mimicking sex hormones (123,143–145), while others disrupt thyroid hormone pathways (123,124,146–149). Air toxics that act as EDCs have been linked to several health outcomes, including insulin resistance, cancers, fetal development, and reproductive issues (150,151, p.2,152–156). The routes of exposure to these air toxics vary, depending on the specific EDC. For instance, exposure to PAHs is due to combustion of organic materials, while exposure to phthalates is due to exposure to plastics or plasticizers (157). Exposure to these toxics also varies depending on environment, with some toxics being more prevalent within indoor environments compared to outdoor environments. In addition to individual effects, it is also important to note that exposure to these chemicals often do not occur in isolation; that is, an individual is often exposed to more than one class of chemicals at a time. Exposure to such mixtures may result in a compounding effect on the body, resulting in a stronger effect than if an individual were exposed to an air toxic individually.

A large number of studies focusing on the relationship between chemicals considered to be air toxics and neurodevelopmental disorders examined all routes of exposure, rather than focusing on solely airborne exposure. Of the studies that focused on airborne exposure, several air toxics have been linked with an increased risk for autism (24,119,158–160), and airborne exposure to PAHs have been linked to an increased risk of ADHD (161), as well as cognitive delays and autism (21,61,62). Furthermore, there have been few studies done on these chemicals and language

development. A total of 5 studies have examined exposure to these chemicals and language development, with all but one finding associations between exposure and abnormal language development (162–166). However, these studies did not exclusively examine airborne exposure to these air toxics; all but one examined all routes of exposure and one focused on dietary exposure. Other issues include the fact that all of these studies have examined single time point outcomes, rather than a trajectory outcome, and that they did not adjust for possible co-pollutant confounding due to limited exposure assessment.

Perhaps more importantly than their individual effects, exposure to air toxics often means all individuals are exposed to multiple chemicals. This arises because air toxics share common sources, creating similar geographic gradients after originating from vehicle exhaust, power plants, and industrial activity. Within a mixture of chemicals, health effects may be magnified, resulting in a more potent effect than exposure to a single chemical alone (e.g. pollutant-pollutant interactions). Furthermore, due to these exposures, there is a high risk of confounding due to the high correlation between the various pollutants and the correlation of the pollutants with the outcome of interest, along with a greater risk of Type I and Type II errors (167). While there are several statistical methods that can deal with pollutant-pollutant confounding, there is no single accepted method for examining pollutant-pollutant confounding. Rather, each method has its own strengths and weaknesses, along with purpose of use, so determining which statistical approach is most appropriate relies on the question being asked and what information is desired (168).

VI. Trajectory Analysis

When evaluating developmental outcomes, there are two common approaches researchers can use: a single time point approach or a longitudinal approach. In the former, researchers focus on a developmental outcome measurement at a single point in time, often resulting in a cross-sectional view of the outcome. In the latter, researchers use multiple outcome measurements over a period of time, resulting in an analysis that takes into consideration an individual's growth over time. While there are many forms of longitudinal analysis, this research focuses on trajectory analysis, which aims to take multiple outcome measurements in order to build and compare growth patterns between individuals.

Trajectory analysis can be performed using several different statistical techniques. These include growth mixture modeling, group based trajectory modelling, latent class analysis, and latent transition analysis (169). However, despite the different methods one can employ, there are multiple strengths of trajectory analysis. Perhaps the clearest strength of studying language trajectories is that trajectories are more representative of development than examining diagnostic outcomes at a single time point (170), given that development by its nature unfolds over time, and is not well-represented by a single-time snapshot. This benefit can be framed as improved measurement of the outcome of interest; that is, reduced outcome misclassification. In addition, by creating trajectory groups, it allows researchers to examine between-person variation within each group, along with between-group variation as well. This may permit researchers to identify unique risk factors within each group, allowing for more targeted interventions to aid those specific groups in preventing detrimental health outcomes.

Chapter 2 - Language Trajectories in Siblings of Children with Autism

<u>Abstract</u>

Background: Language development is a critical part of human development that unfolds over time. We aimed to examine and characterize the trajectories of language development within children with an older sibling with autism, as these children are more likely to have neurodevelopmental delays, thus allowing more detailed exploration of suboptimal language acquisition trajectories.

Methods: Participants were drawn from the Early Autism Risk Longitudinal Investigation (EARLI) (n=251) and the Markers of Autism Risk in Babies – Learning Early Signs (MARBLES) (n=393) cohorts that recruited pregnant mothers who previously had a child with autism (ASD). Expressive and receptive language development was measured using the Mullen Scales of Early Learning (MSEL) at ages 6,12, 24, and 36 months of age. Each child was classified into one of 3 neurodevelopmental classifications: Autism Spectrum Disorder (ASD) (n=93), nontypically developing with no ASD (non-TD) (n=79), or typically developing (TD) (n=250). We used latent class growth analysis (LCGA) to determine unique language trajectories based on MSEL receptive or expressive language raw scores, the child's study-derived neurodevelopmental classification, child sex, maternal education, and maternal race.

Results: We determined four language trajectories for expressive (High Growth, Moderate Growth, Low Growth, and Tracking) and two trajectories for receptive language (High Growth and Late Growth). Overall, we observed that suboptimal language trajectories were most common for children classified as ASD and intermediate for non-TD children versus TD. Non-TD and ASD children were more likely to belong to the Low Growth, Late Growth, and Tracking trajectories.

Conclusion: While previous studies have shown that children with ASD are more likely to have sub-optimal language development, we newly showed that children with non-typical development not meeting ASD criteria also exhibit higher proportions with suboptimal trajectories, even when symptoms associated with neurodevelopmental delays are sub-clinical and do not meet diagnostic criteria for ASD.

Keywords: language development, language trajectory, autism, autism spectrum disorder, latent class growth analysis

Abbreviations: autism spectrum disorder (ASD), latent class growth analysis (LCGA), typically developing (TD), non-typically developing (non-TD), Mullen Scales of Early Learning (MSEL), neurodevelopmental delays (NDDs)

Introduction

Language abilities play a role in almost every aspect of daily life and early childhood is a time of rapid language acquisition. Language development delay or impairment can contribute to atypical social and emotional development, poor academic performance, and increased risk for diagnosis with neurodevelopmental disorders (1–7). Understanding the developmental trajectory of language in populations enriched for likelihood of neurodevelopmental disorders may aid the early identification of those conditions, as well as better target language focused interventions among at risk groups.

Human language development unfolds over time, following trajectories that vary from person to person, but that also cluster in classes. In addition to an optimal language acquisition trajectory defined by the most growth without observed degradation, prior research has identified between 4 to 7 different atypical trajectories or patterns in children from grade school to adolescence (34,35, p.,36,37). Law et al., for example, found evidence for three atypical trajectories: one where individuals start at a lower point but track with normally developing individuals ("tracking"), one where individuals develop at a similar rate but plateau at a certain time point ("plateau"), and one where individuals start at a similar point to normally developing individuals but with slower growth rates ("deterioration") (34). Pickles et al. labeled some of their trajectories as "delay" or "catch-up", both of which appear to follow the proposed "tracking" trajectory from Law et al. Tambyraja et al. likewise found several trajectories, two of which following the proposed "tracking" as described above and one that followed

"deterioration". The patterns identified will vary by the sample size, age of children, and characteristics of the sample, including the prevalence of developmental disabilities.

A substantive portion of existing research that examines language development trajectories has been in the context of autism spectrum disorder (ASD). Individuals diagnosed with ASD, compared to those without, tend to score lower on language tests and have trajectories with a decrease or a slower increase in language abilities (39,40). Although the presence of language impairment is common in individuals with ASD, language development among those with ASD is highly heterogenous (171,172).

Less studied are children who have "sub-clinical" symptoms of ASD or who may otherwise be identified as neuroatypical among other cognitive and behavioral dimensions. Studies including sufficient numbers of these children may be beneficial in more fully elucidating the relationship between neurodevelopment and language impairment trajectories, because the study will have better statistical ability to resolve language patterns in this group. An example of one such study that was enhanced for atypical development and sub-threshold autism-like traits was Landa et al.'s investigation of 204 siblings of children with ASD. This study identified four trajectory groups using latent class growth analysis based on scores from the Mullen Scales of Early Learning (38).

To add to the understanding of language development trajectories among children, especially those with atypical development, we conducted a study of similar design based on data collected from two ASD sibling cohorts, using latent class growth analysis (LCGA). We evaluated how language development varied across children in

these cohorts meeting diagnostic criteria for ASD, showing evidence of atypical development, and appearing to be typically developing.

<u>Methods</u>

Population

We included individuals in the Early Autism Risk Longitudinal Investigation (EARLI) and the Markers of Autism Risk in Babies – Learning Early Signs (MARBLES) studies (173,174). Both studies recruited pregnant women who already had a child with a diagnosis of ASD, or where the biological father had a child with ASD and followed both the mother and the expected child longitudinally. EARLI recruited from several sites, spanning across northeast Maryland, southeast Pennsylvania, and northern California, while MARBLES recruited primarily from northern California. Children in EARLI were born between 2009 and 2013, while children in MARBLES were born between 2006 and 2023. Demographic information was obtained via in-person interviews and questionnaires (Table 2.1).

From 621 children in both cohorts, we excluded persons with 1 or no recorded language scores or who had no study-based classification as ASD, non-TD, or TD, because these were key study variables and to be consistent with prior literature, yielding a sample size of 493 individuals (Table 2.3).

Language Development and Neurodevelopmental Classification

Trained staff administered the Mullen Scales of Early Learning (MSEL, 20) to assess cognitive development (including language), in person, at 6, 12, 24, and 36 months of age. The MSEL is a standardized psychometric test that is used to measure cognitive development in children ages 3 to 60 months (175). The MSEL generates five subscores (gross motor, fine motor, expressive language, receptive language, and visual reception). We used expressive language and receptive language raw subscores to generate our language development trajectories. Both expressive and receptive language involve the ability to process visual, auditory, and written language, though in different ways. Expressive language involves the ability to communicate ideas and thoughts using words and gestures to convey messages accurately and appropriately to others. Receptive language involves the ability to understand and process the meaning of language directed towards an individual.

All study children were assigned a research-based neurodevelopmental classification. When study children were 3 years old, licensed clinical psychologists evaluated them using the Autism Diagnostic Observation Schedules (ADOS, 19). These ADOS scores, together with MSEL composite standardized scores and clinical best estimate of an ASD diagnosis, were used to categorize study children as meeting criteria for autism (ASD), non-typical development (non-TD), and typical development (TD) based on a previously reported algorithm (177). Children classified as ASD had scores over the ADOS cutoff and met the Diagnostic and Statistical Manual of Mental Disorders (DSM) criteria for ASD. Non-TD children had ADOS scores within three points of the diagnostic cutoff or had either one MSEL subscore 2.0 standard deviations below average or two MSEL age adjusted subscores 1.5 standard deviations below average. Thus, non-TD children may have autism-related symptoms subthreshold of diagnostic criteria and/or poorer performance on any 1-2 of 5 MSEL subscales. Of the 96 children who were classified as non-TD, 71 of them were categorized as such due to low scores

on receptive or expressive language. Children who did not meet these criteria were classified as TD.

Statistical Analysis

The most frequently missing variables were expressive and receptive language scores at the middle time point: 24 months: 29% missing (Table 2.4). This may be explained due to a funding cut in EARLI, where this collection point was then dropped. A smaller number were missing other covariates, including expressive and receptive MSEL scores at other time points (6% for 6 months, 2% for 12 months, and 7% for 36 months), maternal educational attainment (1%), homeownership (3%), and maternal race (1%). To maximize our statistical precision and avoid a potential bias by excluding children missing data, we used multiple imputation by chained equations using the R package *mice* to impute missing values for variables listed above, including MSEL scores (178). Imputed distribution of these variables can be seen in Table 2.4.

We used latent class growth analysis (LCGA) to characterize language development trajectories, separately modeling expressive vs. receptive language. LCGA is a method to identify unmeasured (or latent) class membership using observed variables, with the goal of creating classes (or trajectories) so that individuals within a class are more similar than individuals between classes (179,180). We used a "one-step" approach where the same LCGA model was used to find the trajectories that best fit the data and estimate associations between neurodevelopmental classification (ASD, non-TD, TD) with trajectories. In addition to MSEL scores and neurodevelopmental classification (approximately the following a priori potential predictors of language for more accurate prediction for language trajectory assignments: child sex, race/ethnicity,
and maternal education (as a proxy for socioeconomic status) (27–30). Parameters from the LCGA models were used to generate relative risk ratios and 95% confidence intervals, and we performed an overall chi-squared test of the association between neurodevelopmental classification and language trajectories.

We ran LCGA models with 2-7 classes, assessing each model for goodness of fit, as indicated by lower Bayesian information criteria (BIC), sample size adjusted BIC (SABIC), and integrated complete likelihood (ICL), and higher entropy values. These fit measures do not always agree with each other, often pointing to different models as the best fitting model. Because our models generally had low entropy (>0.8), we selected the best-fitting model (number of classes) using the BIC, based on recommendations in Diallo et al. (181), along with favoring fewer over more classes for interpretability, and avoiding models with classes with <5% of the study population. While the complete models use probabilities of language trajectory membership (where a child has probabilities for each trajectory), we assigned each child to the trajectory class for which they had the highest probability for figures.

We selected language trajectory class names based on the overall shape of the class in the plot and guided by designations used in prior literature. We examined the patterns of expressive versus receptive language, examining both MSEL raw score correlations at each time point and concordance between our assigned language trajectories.

All analyses were conducted using Rstudio version 2022.02.1 Build 461 with R Version 4.1.3 "One Push-Up". The LCGA was conducted using the *lcmm* package (182).

<u>Results</u>

Of the included children from EARLI and MARBLES, 114 were designated as ASD, 96 as non-TD, and 283 as TD. Children classified as having ASD were more likely to be male and from families with lower-SES (e.g., lower educational attainment) and of non-white race compared to the other neurodevelopmental classifications (Table 2.1). Children classified as TD were from families with higher SES (higher maternal education and more homeownership).

Mean expressive and receptive MSEL language raw scores increased across age (Figure 2.5). While the mean language scores in infancy (6 and 12 months) were similar across ASD, non-TD, and TD, the scores began to diverge at 24 and 36 months. During those later time points, the mean expressive and receptive MSEL scores were lowest for ASD, intermediate for non-TD, and highest for children classified as TD (Table 2.1). Expressive and receptive language raw scores were correlated at all ages, with higher correlations with later developmental periods, with no individuals exhibiting highly discordant scores (e.g., high expressive with low receptive) (Figure 2.6).

For expressive language, 4 trajectory classes were optimal (Table 2.5). The 4 assigned trajectory classes started at a similar level of language ability at 6 months, diverged slightly at 12 months, and then increasingly diverged at 24 and 36 months of age, resulting in classes that we named High Growth, Tracking, Moderate Growth, and Low Growth (Figure 2.1). The Tracking class mostly followed the High Growth class, albeit with lower scores at 24 months, along with a noticeable increase at 36 months. The majority of children, overall and for all neurodevelopmental classifications, were assigned to the intermediate Tracking trajectory for expressive language (273 of 493,

Figure 2.3). Moderate growth had consistent growth, but mean scores were lower than both Tracking and High Growth classes. The Low Growth class had almost no change in scores between 12 and 24 months, with a slight increase in scores at 36 months.

For receptive language, both the BIC and ICL indicated that the 4-class model was the best fit (Table 2.6). However, the 3 and 4-class models all had at least one class that had less than 5 percent of the population, and so we selected the next best fitting, 2-class model, labeling them as High Growth and Late Growth. Here, the Late Growth trajectory saw slight changes in raw scores between 12 and 24 months, but scores at 36 months jumped considerably (Figure 2.2). A majority of children, overall and for all neurodevelopmental classifications, were assigned to the High Growth trajectory for receptive language (373 of 493, Figure 2.4).

For both receptive and expressive language, we observed that our results had low entropy (entropy > 0.8), which indicated that the class separation was good for our LCGA. There were no individuals who had clearly discordant language trajectory assignments between expressive vs. receptive language (e.g., being assigned to High Growth for expressive but Late Growth for receptive). All individuals were either in the same class or in the adjacent class (Table 2.9).

For both expressive and receptive language, individuals in the Low or Late Growth trajectory were more likely to be male, have mothers who were less educated, and were less likely to identify as non-Hispanic white than individuals who fell in the High Growth trajectory (Tables 2.7 and 2.8). Individuals in the Tracking or Moderate Growth trajectory were less likely to identify as non-Hispanic white and more likely to have mothers who were less educated compared to the High Growth trajectory.

Both expressive and receptive language trajectories were highly associated with a child's neurodevelopmental classification, as shown by very low p-values from our overall chi square tests (Table 2.2 and Figures 2.3 and 2.4).

For expressive language, children with ASD at 36 months were substantially more likely to be in the Low Growth trajectory and were more likely to fall into the Tracking trajectory when compared to children classified as TD (Table 2.2 and Figures 2.3 and 2.4). Language trajectories for children classified as non-TD showed intermediate patterns of suboptimal language – between the more severely impacted ASD distribution which was largely of Moderate and Low Growth language trajectories, and the TD distribution which was largely made up of children with High Growth language trajectories. Non-TD children compared to TD children were more likely to be classified in the Low Growth and Tracking language trajectories, though the confidence bands include the null (Table 2.2). No TD children were classified as Moderate Growth, leading to an inability to calculate relative risk ratios (RRR) for these comparisons.

For receptive language, children with ASD were more likely to fall into the Late Growth trajectory when compared to both TD and non-TD children (Table 2.2 and Figures 2.3 and 2.4). When compared to TD children, non-TD children were more likely to fall into the Late Growth category.

Discussion

The goal of this study was to characterize the trajectory of language development in two cohorts of children with enhanced risk for neurodevelopmental disorders because they had a sibling with ASD, including the distribution of trajectories across

neurodevelopmental classification. Our results supported the hypothesis that a child's neurodevelopmental outcome classification was highly related to their language development trajectory. More specifically, children who were classified as ASD were more likely to belong trajectories that indicate a slower rate of language development and children with non-TD (but without ASD) were generally intermediate in trajectories of language acquisition compared to children classified as having typical development (TD).

Children classified as having ASD were more likely to belong to the Low Growth and Late Growth trajectories for expressive and receptive language, respectively, when compared to both non-TD and TD classified children. The Low Growth trajectory is distinct from other trajectories by having lower overall scores at each time point. Each of the other expressive language trajectories showed a marked increase in mean score at 24 months, which was not reflected in the Low Growth trajectory. Notably, for the Late Growth trajectory, we observed a large jump in score in receptive language at 36 months, an indication that there may be some "catch up" to the High Growth trajectory scores with certain developmental trajectories, though our analysis was not able to observe these other hypothetical trajectories in receptive language. Children classified as non-TD were also more likely to belong to both the Low Growth and Tracking trajectory classes for expressive and the Late Growth class for receptive language when compared to TD children. These results are consistent with prior literature, which shows that children with ASD generally score lower on language tests than those who do not have ASD, although there is a high amount of heterogeneity with language growth (37,39,40). While Landa and Pickles were in comparable cohorts (Landa included

children with a sibling with ASD while Pickles used referrals for possible autism), Brignell pulled from a broader community-based population.

Similar to the "catch up" effect observed in receptive language for the Late Growth trajectory, a smaller jump in score was seen in the Low Growth class for expressive language. Prior literature has indicated that such trajectory shapes are possible to observe in language development (37), and as such, it cannot be discounted that the Late Growth trajectory may be closer to a "tracking" trajectory, or that the Low Growth trajectory may be closer to a "late growth" trajectory. However, the LCGA algorithm did not determine that this was the case, though models that have greater number of classes may have been able to resolve more distinct and nuanced trajectory patterns. These trends highlight that the selection of the number of classes has a significant impact on the findings of all LCGA studies, including ours.

The use of the MSEL is important to note, as the MSEL is one of several inputs used in determining child neurodevelopmental classification as having non-TD, in addition to being the primary determinant of language trajectories. The distinction between a non-TD and TD classification involves criteria across the 5 MSEL subscores, of which expressive and receptive language are 2 of the 5. From our sample, a total of 82 of 96 individuals did not meet these language subscore thresholds. As a result, there is some level of correlation between our neurodevelopmental classifications and independent variables (language trajectories), at least for individuals with a non-TD classification. However, the neurodevelopmental classification algorithm does not only rely solely on the expressive and receptive language subscores of the MSEL, but also uses the fine motor and visual receptive subscores, so there may still be some level of

independence there. Further, the algorithm utilizes the standard score, which reflects distance from the normative mean, while our analysis uses the raw score, which reflects absolute level of language, as our dependent variable.

Our findings are in concert with previous studies and extend the understanding of language trajectories among children with suboptimal neurodevelopment. Landa et al. placed 235 children into three groups: Early-ASD (diagnostic impression of ASD at 14 months), Late-ASD (did not receive a diagnostic impression of ASD at 14 months), and non-ASD (40). They used generalized estimating equations (GEE) to determine differences in mean language scores using the MSEL between the three groups, finding significant differences for receptive and expressive language between the Early-ASD group and the Non-ASD group at all ages, whereas separation between the Late-ASD group and the Non-ASD group occurred only at later developmental periods: 24 and 36 months. Our approach in contrast to that of Landa et al. was able to shed light on a group that had non-typical development but did not meet ASD study criteria. Furthermore, it is only with our LCGA/trajectory approach that we could go describe the shape of language acquisition trajectories (such as we found with our Late Growth group), in contrast to only determining mean differences with a GEE approach. Tek et al. examined 35 children using Brown's 14 grammatical morphemes, dividing them into ASD and TD groups, using individual growth curve analysis to examine the differences (183). They found two distinct language development profiles, which they labeled as ASD-high verbal and ASD-low verbal. Our larger sample size (493) allowed us to detect a greater number of language trajectories (such as the four trajectories we resolved for expressive language) and their links with neurodevelopmental classification. Visser et

al. examined 203 children using LCGA to generate five different trajectory groups based on their Autism Diagnostic Observation Schedule (ADOS) scores, including the ADOS language score, along with other domains such as IQ (184). While Visser et al. was able to describe ASD phenotypic subgroups, the subgroups were more holistic and included other behavioral and cognitive domains beyond just language. Furthermore, all children were referred for ASD, thus preventing the evaluation of language in relation to more typical developmental classifications.

Strengths of our study include the use of language development trajectories, rather than single time point outcomes, which allowed for a more detailed and accurate examination of language acquisition across development. In addition, we had 4 repeated, robust measures of both expressive and receptive language, reducing the possibility of incorrectly assigning an individual to a wrong trajectory class. Second, the use of cohorts which prospectively follow siblings of children with autism allowed for a greater ability to detect abnormal language development trajectories that may be less common in unselected cohorts. Our included cohorts have a heightened prevalence for ASD compared to the general population and have increased prevalence of other neurodevelopmental delays. Third, the use of a continuous neurodevelopmental outcome measurement (i.e., MSEL raw scores), rather than a dichotomous neurodevelopmental outcome, allowed for the examination of individuals who present with "sub-clinical" symptoms (non-TD). Few studies have included these individuals as a separate group, with some including this group as part of the controls or not including them in the analysis altogether. The inclusion of this non-TD group allows a more

complete picture of language development in early childhood across a range of neurodevelopmental outcomes.

Our measurement of language using the MSEL was robust in that it is considered to be a suitable psychometric test for measuring expressive and receptive language (185) and has been shown to have good construct validity (186). Yet the MSEL is limited to expressive and receptive language domains and may miss other aspects of language - a complex phenotype. Future research should explore other facets of language development, such as vocabulary, grammatical development, or semantics. Our analysis was limited to the first 36 months of age, so we are unable to extrapolate language trajectories after this period. Studies that have examined language development patterns in children with ASD past the age of 36 months have largely found that, while children with ASD generally start with lower scores, their growth patterns are similar to those without ASD, with some instances of children with ASD scoring near identically to those without ASD (37,39). Due to the nature of LCGA analysis, our results are sensitive to the number of classes selected and our total sample size. While our sample size is sufficiently large enough to utilize LCGA, selecting for a different number of classes or changing the sample size may alter the results. However, we are confident that the overall direction of effect would remain consistent between various class choices.

In conclusion, we identified unique language trajectories for expressive and receptive language development in a population of children more likely to develop ASD or a developmental delay. Children with an ASD exhibited the most impairment; while children with non-typical development were intermediate, with both of these groups

more likely to belong to the Low or Late Growth category as opposed to the High Growth category when compared to children with typical development. Further exploration of these results should be conducted in larger and more diverse populations and/or with more in-depth characterization of language development measures.

Tables and Figures

Table 2.1: Characteristics of Participants in Early Autism Risk Longitudinal Investigation (EARLI) and Markers of Autism Risk in

Babies – Learning Early Signs (MARBLES) by Study Neurodevelopmental Classification

	ASD (N=114)	Non-TD (N=96)	TD (N=283)	Overall (N=493)
Child Gender				
Female	28 (24.6%)	43 (44.8%)	145 (51.2%)	216 (43.8%)
Male	86 (75.4%)	53 (55.2%)	138 (48.8%)	277 (56.2%)
Maternal Educational Attainment				
Less than high school	5 (4.5%)	3 (3.1%)	6 (2.1%)	14 (2.9%)
High school diploma/GED	8 (7.1%)	9 (9.4%)	12 (4.3%)	29 (5.9%)
Some college	49 (43.8%)	41 (42.7%)	92 (32.6%)	182 (37.1%)
Bachelor's degree	25 (22.3%)	25 (26.0%)	101 (35.8%)	151 (30.8%)
Graduate or professional degree	25 (22.3%)	18 (18.8%)	71 (25.2%)	114 (23.3%)
Missing	2	0	1	3
Homeownership				
Rent	52 (50.5%)	40 (43.0%)	95 (34.2%)	187 (39.5%)
Own	51 (49.5%)	53 (57.0%)	183 (65.8%)	287 (60.5%)
Missing	11	3	5	19
Maternal Race/Ethnicity				
Non-Hispanic White	56 (50.5%)	43 (45.7%)	165 (58.3%)	264 (54.1%)

	ASD (N=114)	Non-TD (N=96)	TD (N=283)	Overall (N=493)
Black/African American	11 (9.9%)	13 (13.9%)	7 (2.5%)	31 (6.4%)
Hispanic	24 (21.6%)	19 (20.2%)	57 (20.1%)	100 (20.5%)
Other/Multiracial	20 (18.0%)	19 (20.2%)	54 (19.1%)	93 (19.0%)
Missing	3	2	0	5
Expressive language raw score (6 months)				
Mean (SD)	6.49 (1.29)	6.11 (1.12)	6.33 (1.27)	6.32 (1.25)
Median [Min, Max]	6.00 [4.00, 11.0]	6.00 [3.00, 10.0]	6.00 [3.00, 12.0]	6.00 [3.00, 12.0]
Missing	18	12	37	67
Receptive language raw score (6 months)				
Mean (SD)	7.85 (1.51)	7.73 (1.56)	8.04 (1.48)	7.94 (1.51)
Median [Min, Max]	8.00 [4.00, 12.0]	8.00 [2.00, 12.0]	8.00 [3.00, 13.0]	8.00 [2.00, 13.0]
Missing	18	12	37	67
Expressive language raw score (12 months)				
Mean (SD)	11.0 (2.70)	11.4 (2.46)	12.5 (2.45)	12.0 (2.60)
Median [Min, Max]	11.0 [5.00, 16.0]	12.0 [6.00, 16.0]	13.0 [5.00, 19.0]	12.0 [5.00, 19.0]
Missing	9	10	10	29

	ASD (N=114)	Non-TD (N=96)	TD (N=283)	Overall (N=493)
Receptive language raw score (12 months)				
Mean (SD)	11.2 (2.14)	12.2 (1.84)	13.0 (1.83)	12.5 (2.04)
Median [Min, Max]	11.0 [5.00, 17.0]	12.0 [7.00, 16.0]	13.0 [9.00, 26.0]	13.0 [5.00, 26.0]
Missing	9	10	10	29
Expressive language raw score (24 months)				
Mean (SD)	16.3 (5.04)	19.4 (3.65)	22.4 (3.84)	20.6 (4.83)
Median [Min, Max]	16.0 [6.00, 28.0]	20.0 [9.00, 26.0]	22.0 [12.0, 33.0]	21.0 [6.00, 33.0]
Missing	42	57	83	182
Receptive language raw score (24 months)				
Mean (SD)	17.3 (5.98)	21.7 (4.62)	25.8 (3.24)	23.3 (5.47)
Median [Min, Max]	15.0 [2.00, 28.0]	22.0 [13.0, 30.0]	26.0 [14.0, 33.0]	25.0 [2.00, 33.0]
Missing	43	57	83	183
Expressive language raw score (36 months)				
Mean (SD)	25.2 (7.18)	29.7 (5.98)	33.8 (3.77)	31.0 (6.27)
Median [Min, Max]	26.0 [6.00, 41.0]	31.0 [15.0, 42.0]	34.0 [23.0, 45.0]	32.0 [6.00, 45.0]

	ASD (N=114)	Non-TD (N=96)	TD (N=283)	Overall (N=493)
Missing	14	5	29	48
Receptive language raw score (36 months)				
Mean (SD)	25.6 (6.89)	30.1 (4.28)	32.6 (3.81)	30.5 (5.54)
Median [Min, Max]	28.0 [2.00, 41.0]	30.0 [22.0, 42.0]	32.0 [24.0, 46.0]	31.0 [2.00, 46.0]
Missing	13	5	31	49

Footnotes for Table 2.1: TD = Typically Developing, Non-TD = Non-typically Developing; Percentages were calculated after removing missing individuals from the sample; We present maternal education here as a categorical variable, but because it is coded as ordinal, we elected to use it as a continuous variable in our analysis to avoid problems with model convergence

Table 2.2: Relative Risk Ratios of Language Development Trajectory by Neurodevelopmental Classification, Participants in Early Autism Risk Longitudinal Investigation (EARLI) and Markers of Autism Risk in Babies – Learning Early Signs (MARBLES)

		Relative Risk Ratios and 95% Confidence Intervals			
	Total (ASD/Non-TD/TD)	ASD vs TD	ASD vs Non-TD	Non-TD vs TD	
		Expressive Language (overall chi-square p value < 2.2e-16			
High Growth (n=106)	106 (5/13/88)	Referent	Referent	Referent	
Tracking (n=273)	273 (40/56/177)	3.7 (1.1, 12.3)	1.5 (0.4, 5.9)	2.4 (0.6, 10.6)	
Moderate Growth (n=34)	34 (29/5/0)	*	12.3 (1.9, 81.3)	*	
Low Growth (n=80)	80 (40/22/18)	32.1 (7.7, 133.0)	3.5 (0.9, 13.5)	9.2 (0.7, 115.8)	
		Receptive Language (overall chi-square p value < 2.2e-16)			
High Growth (n=373)	373 (43/63/267)	Referent	Referent	Referent	
Late Growth (n=120)	120 (71/33/16)	25.9 (12.8, 52.4)	2.8 (1.4, 5.5)	9.0 (2.5, 32.1)	

38

Footnote for Table 2.2: TD = Typically Developing, Non-TD = Non-typically Developing; n=114 for ASD, n=96 for non-TD, n=283 for TD; Multinomial logistic regression models with neurodevelopmental classification status as the main predictor and the trajectory classes as the outcome; All models were adjusted for child sex, maternal race (categorical), and maternal education (continuous, higher is more educated); * = because there were no TD individuals with that fell into the Moderate Growth trajectory class, those RRR were unable to be calculated

	Included (N=493)	Excluded (N=128)	Overall (N=621)
Child Gender			
Female	216 (43.8%)	64 (54.7%)	280 (45.9%)
Male	277 (56.2%)	53 (45.3%)	330 (54.1%)
Missing	0	11	11
Maternal Educational Attainment			
Less than high school	14 (2.9%)	4 (5.8%)	18 (3.2%)
High school diploma/GED	29 (5.9%)	7 (10.2%)	36 (6.4%)
Some college	182 (37.1%)	19 (27.5%)	201 (36.0%)
Bachelor's degree	151 (30.8%)	20 (29.0%)	171 (30.6%)
Graduate or professional degree	114 (23.3%)	19 (27.5%)	133 (23.8%)
Missing	3	59	62
Homeownership			
Rent	187 (39.5%)	31 (44.3%)	218 (40.1%)
Own	287 (60.5%)	39 (55.7%)	326 (59.9%)
Missing	19	58	77

Table 2.3: Characteristics of Participants Included and Excluded from Analyses

	Included (N=493)	Excluded (N=128)	Overall (N=621)
Neurodevelopmental Classification Status			
Typically Developing	283 (59.3%)	3 (33.3%)	286 (57.0%)
Non-typically Developing	96 (18.7%)	3 (33.4%)	99 (19.7%)
ASD	114 (22.0%)	3 (33.3%)	117 (23.3%)
Missing	0	119	119
Maternal Race/Ethnicity			
Non-Hispanic White	264 (54.1%)	39 (52.7%)	303 (53.9%)
Black/African American	31 (6.4%)	8 (10.8%)	39 (6.9%)
Hispanic	100 (20.5%)	12 (16.2%)	112 (19.9%)
Other/Multiracial	93 (19.0%)	15 (20.3%)	108 (19.3%)
Missing	5	54	59
Expressive language raw score (6 months)			
Mean (SD)	6.32 (1.25)	6.33 (1.54)	6.32 (1.28)
Median [Min, Max]	6.00 [3.00, 12.0]	6.00 [3.00, 11.0]	6.00 [3.00, 12.0]
Missing	67	88	155
Receptive language raw score (6 months)			

	Included (N=493)	Excluded (N=128)	Overall (N=621)
Mean (SD)	7.94 (1.51)	8.03 (1.93)	7.95 (1.55)
Median [Min, Max]	8.00 [2.00, 13.0]	8.00 [1.00, 11.0]	8.00 [1.00, 13.0]
Missing	67	88	155
Expressive language raw score (12 months)			
Mean (SD)	12.0 (2.60)	12.0 (2.91)	12.0 (2.62)
Median [Min, Max]	12.0 [5.00, 19.0]	12.0 [6.00, 17.0]	12.0 [5.00, 19.0]
Missing	29	91	120
Receptive language raw score (12 months)			
Mean (SD)	12.5 (2.04)	12.4 (2.49)	12.5 (2.07)
Median [Min, Max]	13.0 [5.00, 26.0]	13.0 [3.00, 16.0]	13.0 [3.00, 26.0]
Missing	29	91	120
Expressive language raw score (24 months)			
Mean (SD)	20.6 (4.83)	20.1 (5.45)	20.6 (4.85)
Median [Min, Max]	21.0 [6.00, 33.0]	20.0 [11.0, 28.0]	21.0 [6.00, 33.0]
Missing	182	116	298

	Included (N=493)	Excluded (N=128)	Overall (N=621)
Receptive language raw score (24 months)			
Mean (SD)	23.3 (5.47)	19.6 (5.62)	23.2 (5.51)
Median [Min, Max]	25.0 [2.00, 33.0]	18.0 [13.0, 29.0]	25.0 [2.00, 33.0]
Missing	183	116	299
Expressive language raw score (36 months)			
Mean (SD)	31.0 (6.27)	26.8 (9.82)	31.0 (6.36)
Median [Min, Max]	32.0 [6.00, 45.0]	25.0 [12.0, 39.0]	32.0 [6.00, 45.0]
Missing	48	120	168
Receptive language raw score (36 months)			
Mean (SD)	30.5 (5.54)	26.5 (9.80)	30.4 (5.65)
Median [Min, Max]	31.0 [2.00, 46.0]	28.0 [5.00, 38.0]	31.0 [2.00, 46.0]
Missing	49	120	169

Footnote for Table 2.3: To be removed from the analysis, an individual needed to be missing 3 or more MSEL scores for either expressive or receptive language or be missing a study classification (ASD, non-TD, or TD)

	Original Data (N=493)	Imputed (N=493)
Child Gender		
Female	216 (43.8%)	216 (43.8%)
Male	277 (56.2%)	277 (56.2%)
Maternal Educational Attainment		
Less than high school	14 (2.9%)	14 (2.8%)
High school diploma/GED	29 (5.9%)	29 (5.9%)
Some college	182 (37.1%)	184 (37.3%)
Bachelor's degree	151 (30.8%)	151 (30.6%)
Graduate or professional degree	114 (23.3%)	115 (23.3%)
Missing	3	
Homeownership		
Rent	187 (39.5%)	194 (39.4%)
Own	287 (60.5%)	299 (60.6%)
Missing	19	
Maternal Race/Ethnicity		

Table 2.4: Characteristics of Included Study Sample Before and After Imputing Missing Data

	Original Data (N=493)	Imputed (N=493)
Non-Hispanic White	264 (54.1%)	265 (53.8%)
Black/African American	31 (6.4%)	31 (6.3%)
Hispanic	100 (20.5%)	100 (20.3%)
Other/Multiracial	93 (19.0%)	97 (19.7%)
Missing	5	
Expressive language raw score (6 months)		
Mean (SD)	6.32 (1.25)	6.27 (1.25)
Median [Min, Max]	6.00 [3.00, 12.0]	6.00 [3.00, 12.0]
Missing	67	
Receptive language raw score (6 months)		
Mean (SD)	7.94 (1.51)	7.86 (1.54)
Median [Min, Max]	8.00 [2.00, 13.0]	8.00 [2.00, 13.0]
Missing	67	
Expressive language raw score (12 months)		
Mean (SD)	12.0 (2.60)	12.0 (2.61)

	Original Data (N=493)	Imputed (N=493)
Median [Min, Max]	12.0 [5.00, 19.0]	12.0 [5.00, 19.0]
Missing	29	
Receptive language raw score (12 months)		
Mean (SD)	12.5 (2.04)	12.5 (2.00)
Median [Min, Max]	13.0 [5.00, 26.0]	13.0 [5.00, 26.0]
Missing	29	
Expressive language raw score (24 months)		
Mean (SD)	20.6 (4.83)	20.4 (4.93)
Median [Min, Max]	21.0 [6.00, 33.0]	20.0 [6.00, 33.0]
Missing	182	
Receptive language raw score (24 months)		
Mean (SD)	23.3 (5.47)	22.9 (5.62)
Median [Min, Max]	25.0 [2.00, 33.0]	25.0 [2.00, 33.0]
Missing	183	

	Original Data (N=493)	Imputed (N=493)
Expressive language raw score (36 months)		
Mean (SD)	31.0 (6.27)	31.1 (6.12)
Median [Min, Max]	32.0 [6.00, 45.0]	32.0 [6.00, 45.0]
Missing	48	
Receptive language raw score (36 months)		
Mean (SD)	30.5 (5.54)	30.6 (5.43)
Median [Min, Max]	31.0 [2.00, 46.0]	31.0 [2.00, 46.0]
Missing	49	

Footnote for Table 2.4: To be removed from the analysis, an individual needed to be missing 3 or more MSEL scores for either expressive or receptive language or be missing a neurodevelopmental classification; Percentages were calculated after removing missing individuals from the sample; We present maternal education here as a categorical variable, but because it is coded as ordinal, we elected to use it as a continuous variable in our analysis to avoid problems with model convergence

Number of classes	Log- likelihood	SABIC	BIC	Entropy	ICL
1	-5440.39	10917.1	10955.19	1	10955.19
2	-5127.54	10336.8	10422.5	0.915236*	10451.47
3	-4979.2	10085.51	10218.82	0.844211	10303.2
4	-4908.17	9988.841	10169.76*	0.851522	10271.24*
5	-4887.5	9992.901	10221.43	0.788803	10389
6	-4843.89*	9951.077*	10227.22	0.87692	10335.94
7	-1.00E+09	2E+09	2E+09	1	2E+09

Table 2.5: Measures of Model Fit for Expressive Language Latent Class Growth Analysis

Footnote for Table 2.5: Higher log-likelihood, lower SABIC (sample adjusted BIC), lower BIC (Bayesian Inclusion Criterion), higher entropy, and lower ICL (integrated classification likelihood) indicates an overall better fit; * indicates the best model based on the above criteria (single class will always have an entropy of 1 so we select the next highest entropy instead); we prioritized BIC and

ICL based on the findings in Diallo et al., 2016

Number of classes	Log- likelihood	SABIC	BIC	Entropy	ICL
 1	-5346.83	10729.97	10768.06	1	10768.06
2	-5003.98	10089.68	10175.38	0.900346	10209.44
3	-4860.16	9847.437	9980.746	0.95275*	10006.34
4	-4658.26	9489.037	9669.955*	0.920754	9724.116*
5	-1.00E+09	2E+09	2E+09	1	2E+09
6	-4576.24*	9415.777*	9691.915	0.83438	9838.213
7	-4581.78	9472.269	9796.017	0.911532	9880.888

Table 2.6: Measures of Model Fit for Receptive Language Latent Class Growth Analysis

Footnote for Table 2.6: Higher log-likelihood, lower SABIC (sample adjusted BIC), lower BIC (Bayesian Inclusion Criterion), higher entropy, and lower ICL (integrated classification likelihood) indicates an overall better fit; * indicates the best model based on the above criteria (single class will always have an entropy of 1 so we select the next highest entropy instead); 5 and 6 class models failed to converge; we prioritized BIC and ICL based on the findings in Diallo et al., 2016; classes had <5% of the population in the 3 and 4 class models, so we selected the next best fit model instead

	High Growth (N=106)	Tracking (N=273)	Moderate Growth (N=34)	Low Growth (N=80)	Overall (N=493)
Child Gender					
Female	51 (48.1%)	128 (46.9%)	7 (20.6%)	30 (37.5%)	216 (43.8%)
Male	55 (51.9%)	145 (53.1%)	27 (79.4%)	50 (62.5%)	277 (56.2%)
Maternal Educational Attainment					
Less than high school	44 (41.5%)	82 (30.0%)	5 (14.7%)	20 (25.0%)	151 (30.6%)
High school diploma/GED	30 (28.3%)	65 (23.8%)	8 (23.5%)	12 (15.0%)	115 (23.3%)
Some college	3 (2.8%)	17 (6.2%)	3 (8.8%)	6 (7.5%)	29 (5.9%)
Bachelor's degree	2 (1.9%)	5 (1.8%)	3 (8.8%)	4 (5.0%)	14 (2.8%)
Graduate or professional degree	27 (25.5%)	104 (38.1%)	15 (44.1%)	38 (47.5%)	184 (37.3%)
Homeownership					
Rent	35 (33.0%)	105 (38.5%)	19 (55.9%)	35 (43.8%)	194 (39.4%)
Own	71 (67.0%)	168 (61.5%)	15 (44.1%)	45 (56.3%)	299 (60.6%)
Maternal Race/Ethnicity					
Non-Hispanic White	60 (56.6%)	155 (56.8%)	14 (41.2%)	36 (45.0%)	265 (53.8%)

Table 2.7: Characteristics of Study Population Stratified by Assigned Expressive Language Trajectory Classes

	High Growth (N=106)	Tracking (N=273)	Moderate Growth (N=34)	Low Growth (N=80)	Overall (N=493)
Black/African American	6 (5.7%)	12 (4.4%)	4 (11.8%)	9 (11.3%)	31 (6.3%)
Hispanic	14 (13.2%)	56 (20.5%)	10 (29.4%)	20 (25.0%)	100 (20.3%)
Other/Multiracial	26 (24.5%)	50 (18.3%)	6 (17.6%)	15 (18.8%)	97 (19.7%)
Expressive language raw score (6 months)					
Mean (SD)	6.33 (1.34)	6.25 (1.14)	6.62 (1.71)	6.14 (1.25)	6.27 (1.25)
Median [Min, Max]	6.00 [3.00, 10.0]	6.00 [4.00, 12.0]	6.00 [3.00, 11.0]	6.00 [3.00, 11.0]	6.00 [3.00, 12.0]
Expressive language raw score (12 months)					
Mean (SD)	13.0 (2.58)	12.0 (2.45)	11.3 (2.41)	11.0 (2.79)	12.0 (2.61)
Median [Min, Max]	13.0 [5.00, 18.0]	12.0 [5.00, 19.0]	12.0 [6.00, 15.0]	11.5 [6.00, 15.0]	12.0 [5.00, 19.0]
Expressive language raw score (24 months)					
Mean (SD)	27.0 (2.74)	19.8 (2.80)	11.9 (3.47)	17.5 (3.17)	20.4 (4.93)
Median [Min, Max]	26.0 [21.0, 33.0]	20.0 [6.00, 26.0]	12.0 [6.00, 19.0]	18.0 [11.0, 27.0]	20.0 [6.00, 33.0]
Expressive language raw score (36 months)					
Mean (SD)	36.8 (2.76)	32.6 (2.49)	16.7 (4.50)	24.3 (2.48)	31.1 (6.12)

	High Growth	Tracking	Moderate Growth	Low Growth	Overall
	(N=106)	(N=273)	(N=34)	(N=80)	(N=493)
Median [Min, Max]	37.0 [31.0, 45.0]	32.0 [26.0, 42.0]	18.0 [6.00, 23.0]	25.0 [18.0, 29.0]	32.0 [6.00, 45.0]

	High Growth (N=373)	Late Growth (N=120)	Overall (N=493)
Child Gender			
Female	179 (48.0%)	37 (30.8%)	216 (43.8%)
Male	194 (52.0%)	83 (69.2%)	277 (56.2%)
Maternal Educational Attainment			
Less than high school	10 (2.7%)	4 (3.3%)	14 (2.8%)
High School/GED	18 (4.8%)	11 (9.2%)	29 (5.9%)
Some college	121 (32.4%)	63 (52.5%)	184 (37.3%)
Bachelor's degree	127 (34.0%)	24 (20.0%)	151 (30.6%)
Graduate or Professional degree	97 (26.0%)	18 (15.0%)	115 (23.3%)
Homeownership			
Rent	127 (34.0%)	67 (55.8%)	194 (39.4%)
Own	246 (66.0%)	53 (44.2%)	299 (60.6%)
Maternal Race/Ethnicity			
Non-Hispanic White	212 (56.8%)	53 (44.2%)	265 (53.8%)

Table 2.8: Characteristics of Study Population Stratified by Assigned Receptive Language Trajectory Classes

	High Growth (N=373)	Late Growth (N=120)	Overall (N=493)
Black/African American	18 (4.8%)	13 (10.8%)	31 (6.3%)
Hispanic	67 (18.0%)	33 (27.5%)	100 (20.3%)
Other/Multiracial	76 (20.4%)	21 (17.5%)	97 (19.7%)
Receptive language raw score (6 months)			
Mean (SD)	7.95 (1.52)	7.59 (1.56)	7.86 (1.54)
Median [Min, Max]	8.00 [2.00, 13.0]	8.00 [3.00, 11.0]	8.00 [2.00, 13.0]
Receptive language raw score (12 months)			
Mean (SD)	12.8 (2.00)	11.6 (1.75)	12.5 (2.00)
Median [Min, Max]	13.0 [5.00, 26.0]	12.0 [7.00, 15.0]	13.0 [5.00, 26.0]
Receptive language raw score (24 months)			
Mean (SD)	25.6 (2.69)	14.4 (3.42)	22.9 (5.62)
Median [Min, Max]	26.0 [16.0, 33.0]	14.0 [2.00, 21.0]	25.0 [2.00, 33.0]
Receptive language raw score (36 months)			
Mean (SD)	32.4 (3.83)	25.1 (6.02)	30.6 (5.43)
Median [Min, Max]	32.0 [24.0, 46.0]	27.0 [2.00, 35.0]	31.0 [2.00, 46.0]

Expressive Language <u>Trajectories</u>	Receptive Language Trajectories		
	High Growth	Late Growth	
High Growth	105	1	
Tracking	232	41	
Moderate Growth	3	31	
Low Growth	33	47	

Table 2.9: Cross-classification of study population by assigned language trajectories

Neurodevelopmental Classification					
Expressive Language	ASD	Non-TD	TD		
Trajectories					
High Growth	5	13	88		
Tracking	40	56	177		
Moderate Growth	29	5	0		
Low Growth	40	22	18		
Receptive Language					
Trajectories					
High Growth	43	63	267		
Late Growth	71	33	16		

Table 2.10: Total number of children falling into trajectories based on Neurodevelopmental Classification



Figure 2.1: Expressive Language Trajectories

Footnotes for Figure 2.1: Bars indicate 95 percent confidence interval centered around the mean at each time point



Figure 2.2: Receptive Language Trajectories





Figure 2.3: Proportions of trajectory class assignment for expressive language

Footnotes for Figure 2.3: Trend shows that those with a diagnosis of ASD or Non-TD are more likely to fall into Low and Moderate Growth compared to Tracking and High Growth


Figure 2.4: Proportions of trajectory class assignment for receptive language

Footnotes for Figure 2.4: Trend shows that those with a diagnosis of ASD or Non-TD are more likely to fall into Late Growth



Figure 2.5: Distributions of MSEL Expressive and Receptive Language Raw Scores at 6, 12, 24, and 36 Months

Footnotes for Figure 2.5: Bars indicate 95 percent confidence intervals centered on mean



Figure 2.6: Scatterplots of MSEL Expressive vs Receptive Language Raw Scores at Each Time of Data Collection

Footnote for Figure 2.6: Comparison of expressive and receptive language scores; disjoint scores would be present in the upper left and lower right corners; we observe a mostly linear relationship between the two scores

Chapter 3 - The Impact of Prenatal Phthalate Exposure on Language Development Trajectories in Siblings of Children with Autism

<u>Abstract</u>

Background: Language development is a critical part of human development that unfolds across time. We aimed to examine how prenatal phthalate exposure affects early childhood language development, utilizing a robust longitudinal analysis methodology.

Methods: Participants were drawn from the Early Autism Risk Longitudinal Investigation (EARLI) (n=251) and the Markers of Autism Risk in Babies – Learning Early Signs (MARBLES) (n=393) cohorts that recruited pregnant mothers who previously had a child with autism (ASD). Expressive and receptive language development was measured using the Mullen Scales of Early Learning (MSEL) at ages 6,12, 24, and 36 months of age. A total of 14 phthalate metabolites were assessed using multiple first morning voids at each trimester of pregnancy. We used latent class growth analysis (LCGA) to determine language trajectories based on MSEL receptive or expressive language raw scores, prenatal phthalate exposure, cohort, child sex, maternal age, maternal race/ethnicity, homeownership, and maternal educational attainment.

Results: We found 3 trajectories for both expressive and receptive language using both a one-step and two-step LCGA approach. The general direction of risk did not change between the one and two step approaches. Most phthalates were not statistically significant. Certain metabolites of di(2-ethylhexyl) phthalate decreased risk of falling into an abnormal trajectory for receptive language, and metabolites of diisodecyl phthalate decreased risk of falling into the Tracking trajectory for expressive language using the one-step approach. The metabolites of di-n-butyl phthalate

increased risk for falling into the Low Growth trajectory for expressive language using the one-step approach.

Conclusion: Most of our phthalates were not statistically significant, though some trends were observed among low and high molecular weight phthalates. These trends were largely consistent with prior literature.

Introduction

The establishment of language is an important milestone in early childhood development, as it plays a role in almost every aspect of daily life. Language development delay or impairment may continue into adulthood, and can contribute to atypical social and emotional development, poor academic performance, and increased risk for diagnosis with neurodevelopmental disorders (Beitchman & Brownlie, 2012; Howlin & Udwin, 2002; Irwin et al., 2002; Johnson et al., 1999; Roulstone et al., 2011; Snowling et al., 2006; Whitehouse et al., 2009). The causes of language delay are multifaceted and varied, with some common social factors including family history, child sex, maternal education, and home environment (28,187,188). There are also varied environmental chemical exposures that are suspected to be linked with language delays, including certain heavy metals (e.g. lead, mercury), polychlorinated biphenyls, polybrominated diphenyl ethers, phthalates, and organophosphate insecticides (74,189–192), although, for some of these exposures, both the direction and magnitude of causality with language have proven to be inconsistent.

Certain biological systems are critical to early neurodevelopment and language development, such as the endocrine system. Environmental pollutants that disrupt this system, known as endocrine disrupting chemicals (EDCs) are therefore of vital interest, having a higher *a priori* suspicion of causing language impairment. As such, understanding environmental causes that may act as EDCs will not only provide a better understanding behind the etiology of language disorders, but also provide stakeholders with vital knowledge to make impactful changes to reduce overall risk.

Phthalates are a class of environmental exposures with known impacts on the endocrine system and neurodevelopment. They are plasticizers used in a wide variety of products, including cosmetics, food packaging, cleaning materials, and pharmaceuticals. Phthalates may impair neurodevelopment via several proposed pathophysiologies, including cell apoptosis (80,87), affecting neurotransmitter pathways (80,88), and altering gene expression and hormone levels (87,89). These effects are thought to be linked to potential neurodevelopmental issues, including disorders such as autism and ADHD (78,91,92) along with deficits in other neurological domains such as cognition (93,94), motor effects (95), and behavioral outcomes (96–98).

Of the multiple dozens of phthalates used in products, they are grouped broadly by low molecular weight (LMW) and high molecular weight (HMW) (193). HMW phthalates, such as di-(2-ethylhexyl) phthalate (DEHP), di-isononyl phthalate (DiNP), and di-isodecyl phthalate (DiDP) are typically used to add flexibility to polyvinyl chloride plastic (PVC) and are used in plastic tubing and food packaging. LMW ph thalates, such as diethyl phthalate (DEP), dimethyl phthalate (DMP) and dibutyl phthalate (DBP) tend to be used as solvents, and are more present in household products such as adhesives and cosmetics (193,194). The difference in metabolism of the different types of phthalates are thought to result in differing biological pathways. HMW phthalates tend to form oxidized metabolites, while LMW phthalates tend to form monoesters. These monoesters are theorized to be more relevant to androgen insufficiency (193). However, toxicity profiles can vary between various phthalates, and not all phthalates have been thoroughly investigated for their mechanisms of action.

Phthalates have been studied in relation to language impairment, including both HMW and LMW phthalate groups. Results of impacts on language have been inconsistent between molecular weight classes and individual phthalates.

Findings of LMW phthalates (as a class) and language vary widely to include increased and decreased risk (protective associations). One study found increased risk of language impairment for all LMW phthalates examined (106), three studies found increased or decreased risk depending on the individual phthalate compound examined (74,107,108), and one study found protective associations for all LMW phthalates examined (97). Olesen et al. found various results depending on the method of language measurement, with increased risk for complexity scores on the MacArthur-Bates Communicative Development Inventories for all LMW phthalates, but individual LMW phthalates had increased or decreased risk for vocabulary scores on the MacArthur-Bates Communicative Development Inventories. In addition, Olesen found sex differences, where males generally exhibited risk associations that leaned more protective, though none reached statistical significance, for both measures of language (105).

For HMW phthalates, the direction of association has been more consistent with increased risk across several studies, suggesting that HMW phthalates may be more harmful to the developing nervous system, although results were still dependent on sex and the language measurement tool. For instance, Bornehag et al., Hyland et al, and Polanska et al., all found increased risk for all HMW phthalates observed (74,106,108). Olesen et al. found increased risk in males for both vocabulary and complexity with

HMW phthalates, but females had a decreased risk, except for the metabolite mEHP which had an increased risk (105). Likewise, Huang et al. found decreased risk for all HMW phthalates when examining verbal IQ, but an increased risk when examining verbal comprehension index on the Wechsler Intelligence Scale for Children-IV (103).

Possible reasons for the high variability in findings may reflect true nuance where impairment only follows from exposure to a particular phthalate compound, pertains to only a certain subtle attribute of language that is only detectable by certain psychometric tests, or is only evident in one sex. Alternately, heterogenous findings may be influenced by errors such as exposure misclassification, precision or sampling error (particularly for those studies with smaller sample sizes), and confounding. These studies of phthalates and language generally excelled in prospective designs where phthalates were primarily measured via biomarkers and used psychometric tests to measure language development, but only examined a single time point during development. One weakness that has been noted in prior studies involve the possibility of exposure misclassification, perhaps due to measurements of phthalate exposures at a single point during development (97,107,108), which may result in no association being observed between phthalate exposure and delayed language development.

To clarify the possibility that certain phthalates, and phthalate molecular classes, may exert deleterious impacts on language, we conducted a study based on data collected from two autism sibling cohorts. Phthalates were measured using urinary biomarkers including more than one point of collection during pregnancy. We utilized latent class growth analysis (LCGA) to capture different trajectories of language acquisition measured 4 times in early childhood. We examined the association of 15

prenatal phthalate exposures and their potential impact on language development trajectories and examined whether results differed by child sex.

<u>Methods</u>

Population

We included individuals in the Early Autism Risk Longitudinal Investigation (EARLI) and the Markers of Autism Risk in Babies – Learning Early Signs (MARBLES) studies (173,174). Both studies recruited pregnant women who already had a child with a diagnosis of ASD or whose fathers who had a previous biological child with ASD. Both the mother and the expected child were followed longitudinally. EARLI recruited from several sites, spanning across northeast Maryland, southeast Pennsylvania, and northern California, while MARBLES recruited primarily from northern California. Children in EARLI were born between 2009 and 2013, while children in MARBLES were born between 2006 and 2023. Demographic information was obtained via in-person interviews and questionnaires.

Language Development and Neurodevelopment Measurement

The Mullen Scales of Early Learning (MSEL, 20) was used to assess language development at 6, 12, 24, and 36 months of age. The MSEL is a standardized psychometric test that is used to measure cognitive development in children ages 3 to 60 months (175). Trained staff administered the MSEL in person at each of the time points above for both cohorts, having the child complete MSEL tasks and grading them based on tasks completed. The MSEL generates five subscores (gross motor, fine motor, expressive language, receptive language, and visual reception). We used

expressive language and receptive language raw subscores to generate our language development trajectories. Both expressive and receptive language involve the ability to process visual, auditory, and written language, though in different ways. Expressive language involves the ability to communicate ideas and thoughts through the use of words and gestures to convey messages accurately and appropriately to others. Receptive language involves the ability to understand and process the meaning of language directed towards an individual. At the 36 month visit, children were assigned a neurodevelopmental classification from licensed clinical psychologists using the Autism Diagnostic Observation Schedules (ADOS, 19). The ADOS, together with MSEL composite standardized scores and clinical best estimate of an ASD diagnosis, were used to categorize study children as meeting criteria for autism (ASD), non-typical development (non-TD), and typical development (TD) based on a previously reported algorithm (177).

Phthalate Measurements

Mothers in EARLI were instructed to provide up to two first morning void (FMV) urine samples during their 1st, 2nd, or 3rd trimester. Further details on the collection and analysis for the EARLI phthalate metabolites are detailed in a previous paper (195). For MARBLES participants, mothers were instructed to collect three FMVs and one 24-hour urine sample in each trimester. Further details on the collection and analysis for the MARBLES phthalate metabolites are detailed in a previous paper (196). A total of 14 metabolites were quantified in both studies: monoethyl phthalate (mEP2), mono-isobutyl phthalate (mBP2), monohydroxy-n-butyl phthalate (MHBP), monobenzyl phthalate (mBzP2),

mono(2-ethylhexyl) phthalate (mEHP), mono(2-ethyl-5-hydroxyhexyl) phthalate (mEHHP), mono(2-ethyl-5-oxohexyl) phthalate (mEOHP), mono(2-ethyl-5carboxypentyl) phthalate (mECPP), mono(3-carboxypropyl) phthalate (mCPP), monoisononyl phthalate (mNP2), mono-carboxyisooctyl phthalate (mCOP), and monocarboxyisononyl phthalate (mCNP). We also constructed ΣDEHP due to the high correlation between the metabolites of DEHP (mEHP, mEHHP, mEOHP, and mECPP) by using the molar sum of those metabolites: we divided their concentrations by their molecular weight and then summed the results.

To control for urinary dilution, we used two measurements. For EARLI, creatinine concentration was collected, and for MARBLES, specific gravity was collected. For each measurement, a site-specific Z-score was created and included in all models along with the raw unadjusted reported phthalate concentrations. Creatinine concentrations were log-base2 transformed prior to the calculations of Z-scores. This is consistent with methods described in Chiu et al. (197).

For phthalate concentrations below the limit of detection (LOD), we used machine-observed concentrations without substitution or transformation. Because some women had multiple phthalate measurements, we created an average exposure for each individual for each phthalate metabolite by summing all phthalate measurements taken, then dividing by the total number of samples taken during the prenatal period. Because MARBLES also had pooled samples that used multiple samples for a single phthalate concentration, we generated a weighted average metabolite concentration for that mother using the following formula: $C_{avg} = \frac{C_i + C_{pool} * N_{pool}}{N_{mool} + N_i}$, where C_i is the concentration for any individual sample (i.e. non-pooled samples), C_{pool} is the concentration of the pooled sample, N_{pool} is the number of samples used in the pool, and N_i is the number of individual samples. For individuals that did not have pooled samples, we averaged the individual samples together to obtain the average metabolite concentration. To best model phthalates numerically in relation to language scores, we examined the AIC for simple models using phthalates and the expressive and receptive language scores at 36 months (Supplemental Table 2). We found that transformations of phthalates were not necessary. However, we winsorized phthalate metabolite measures at the 99th percentile to reduce the effects of extreme measurements. We divided all phthalates by their interquartile range for all statistical analyses.

Statistical Analysis

A total of 621 individuals were in the initial study populations. We excluded persons missing the study-derived diagnostic classification (n=119) or for whom we weren't confident in their language trajectories (those with 1 or no recorded language scores out of the 4 data collections points, n=98), or who were missing all phthalate measurements (n=209), yielding a sample size of 362 individuals (Table 3.2). The most frequently missing variables were expressive and receptive language scores at 24 months: 39.8% and 40.1% missing, respectively (Table 3.3). A total of 209 persons were missing phthalate values, and when one was missing, the entire suite of phthalate metabolites was missing. A smaller number were missing other covariates, including expressive and receptive Mullen scores at other time points (15.7% for 6 months, 6.1% for 12 months, and 0.8 and 1.1% for 36 months respectively), maternal educational attainment (0.3%), homeownership (3.5%), and maternal race (1.0%). To include

persons missing covariates or 1 or 2 Mullen scores, we used multiple imputation by chained equations using the R package *mice* to impute missing values for variables listed above, including Mullen scores (178). Imputed distribution of these variables can be seen in Table 3.3.

We used a latent class growth analysis (LCGA) approach to analyze the relationship between individual phthalates and language development trajectories, separately modeling expressive and receptive language. We used LCGA to both create language development trajectories that best fit the data and to assign children to the trajectory most appropriate to their language development. LCGA is a method to identify unmeasured (or latent) class membership using observed variables, with the goal of creating classes (or trajectories) so that individuals within a class are more similar than individuals between classes (179,180). We performed two types of LCGA analysis: a two-step and a one-step process.

Our primary analyses consisted of a two-step LCGA approach, where we first assigned language trajectories, and then in a separate model regressed these on phthalate exposures. We ran several LCGA models with different numbers of classes, ranging from 2 classes to 7 classes, using only Mullen scores as our inputs. We selected the best number of classes using the following measures of fitness: Bayesian information criteria (BIC), sample size adjusted BIC (SABIC), integrated complete likelihood (ICL), and entropy, selecting the model that had the best measures of fit (i.e. low BIC, ICL, and SABIC with low entropy [>0.8]). We prioritized using the BIC based on recommendations observed in Diallo et al. (181), along with two additional considerations: we favored models with fewer classes to aid in interpretability, and we

avoided models with low number of individuals in a class (<10% of the study population). After individuals were assigned, a multinomial logistic regression was performed using the assigned classes as the outcome and phthalate exposures as our main predictors, along with site-specific Z-score for urine dilution. We adjusted all models for cohort (EARLI vs. MARBLES) and also accounted for the following a priori potential predictors of language: child sex, maternal age, maternal race/ethnicity, maternal education, and home-ownership (as proxies for socioeconomic status) (27– 30).

In a sensitivity analysis, we used a one-step LCGA, where the language trajectories were established in one model simultaneously with the same covariates as above and phthalate exposures. The inclusion of these predictors in the LCGA algorithm holds several advantages: misclassification of individuals into the wrong trajectory class is irrelevant (198), measurement errors of trajectory membership are incorporated into the analysis (198), and the inclusion of covariates can improve class separation and reduces standard errors (199). We ran several LCGA models with different numbers of classes, ranging from 2 classes to 5 classes, using the same method described above to select the optimum number of trajectory classes.

For both the two-step and one-step processes, we assessed the coefficients and confidence intervals for each phthalate metabolite as a predictor of language trajectory class. Following each process, each person was assigned a probability of belonging to each trajectory class (separately for two and one step), and we assigned each person to their highest probability class for tables and figures. We selected class names based on the overall shape of the class in the plot, guided by designations used in prior literature.

All analyses were conducted using Rstudio version 2022.02.1 Build 461 with R Version 4.1.3 "One Push-Up". The LCGA was conducted using the *lcmm* package (182).

<u>Results</u>

Males had slightly lower MSEL raw scores than females at 36 months of age (Table 3.1). Children whose mothers were 30 or younger or who did not receive at least a college degree had lower MSEL raw scores than children who had mothers older than 30 or who received at least a college degree. Children of African American or Hispanic mothers also had lower MSEL raw scores than children from non-Hispanic white or multiracial mothers. Children whose parents rented had lower MSEL scores than children of homeowners. There was no appreciable difference in MSEL scores between birth seasons.

There was a tendency for many of the phthalate metabolites to be higher among females (Supplemental Table 1). Many of the HMW phthalates were increasingly higher for older maternal ages and higher levels of maternal education. Most phthalates did not seem consistently patterned by race, except for mEP2 and mCOP where concentrations were several-fold higher among mothers who identified as African American.

With our two-step LCGA approach, for expressive language, 3 trajectory classes were optimal for capturing patterns in language development. These trajectory classes were named High Growth, Tracking, and Low Growth (Figure 3.1). None of the phthalates were statistically significant for associations with expressive language trajectories (Table 3.4). In general, we observed that most metabolites of LMW

phthalates saw an increase in the likelihood for falling into the Tracking and Low Growth versus the more optimal High Growth trajectory categories. DBP was the LMW phthalate with the strongest risk ratios with the suboptimal trajectories. Higher levels of the HMW butyl benzyl phthalate (BzBP) exhibited increased likelihood of being in the suboptimal expressive language trajectories. Other HMW phthalates saw a decrease in likelihood of falling into Tracking and Low Growth categories (e.g. protective direction), and this pattern was most notable for metabolites of di(2-ethylhexyl) phthalate.

With our two-step LCGA approach, for receptive language, 3 trajectory classes were optimal for capturing patterns in language development, which we named High Growth, Tracking, and Late Growth (Figure 3.2). Metabolites of LMW phthalates generally had near-null associations, except for those of DEP, which exhibited somewhat elevated associations with the Tracking and Low Growth trajectories (but without being statistically significant). Associations with HMW phthalate metabolites were also largely near-null, except for BzBP, which exhibited elevated (risk) associations, and DEHP, for which several metabolites exhibited statistically significant protective associations for both Tracking and Late Growth trajectories.

We examined whether sex was a modifier by including a sex by phthalate metabolite interaction term. For expressive language, based on likelihood ratio testing, sex did not appear to be a modifier for any phthalate metabolites (Table 3.11). For receptive language, sex was suggestive as a possible effect modifier for the metabolites of DiNP (mCOP and mNP2) and was most notable and consistent for metabolites of DEHP (Table 3.12). Especially for the Late Growth Trajectory, the association with these metabolites exerted a stronger protective effect among females versus males.

In sensitivity analyses, we also conducted a one-step LCGA. Like our two-step approach, we used goodness-of-fit measures to determine the total number of classes from between 1 and 5 total classes (Tables 3.9 and 3.10). While goodness-of-fit measures sometimes indicated more classes, an examination of the distribution of population showed that at least one class would have fewer than 10 percent of the study population. In general, the one-step and two-step class distributions were broadly overlapping, with only a few individuals being classified as a different class (Table 3.9 and 3.12).

With our one-step LCGA approach, for expressive language, 3 trajectory classes were found to be optimal for all phthalates, which we named High Growth, Tracking, and Low Growth. The shape of the trajectories generally followed those observed in our two-step approach (Figures 3.3 to 3.16). In general, trends observed in the two-step process were observed in the one-step approach. However, two metabolites were statistically significant where they were not in the two-step process.

With our one-step LCGA approach, for receptive language, 3 trajectory classes were found to be optimal for all metabolites, which we labeled as High Growth, Tracking, and Late Growth. The shape of the trajectories generally followed those observed in our two-step approach (Figures 3.17 to 3.31). In general, trends observed in the two-step process were observed in the one-step approach.

Discussion

The goal of this study was to characterize language development trajectories in two cohorts of children with enhanced familial risk for autism and evaluate if prenatal

phthalate exposure was associated with delayed or abnormal language development. Results were mixed depending on the individual phthalate. Most phthalates, whether HMW or LMW, were not associated with impaired language trajectories. The LMW phthalates DBP and DEP exhibited a trend toward impaired expressive language and DEP with impaired receptive language. Among the HMW phthalates, BzBP exhibited a trend with suboptimal language development. Metabolites of the HMW DiDP, DiNP, DEHP, and the metabolite mCPP consistently exhibited a trend towards protective effects on both expressive and receptive language.

While most phthalate exposures were consistent with the direction of their risk between the various trajectory classes, there were a handful of instances where a metabolite was protective for one class but raised risk or had a null effect for another class. For example, in our two-step approach for expressive language, miBP and mCNP saw a decrease in risk for belonging to the Low Growth class but a null effect for belonging to the Tracking class. This may indicate a true difference between the risk of belonging to those two trajectories. It may also be possible that in this sample which is enhanced for genetic liability for neurodevelopmental disorders generally and autism, the action of phthalates differs from an unselected population. We also observed that certain phthalates were heightened in specific populations (e.g., mEP2 in African Americans were significantly higher compared to other races). This may be due to the relatively small sample size of the categories, or it may indicate a true difference in phthalate distribution based on those categories.

Multiple testing is of particular concern in our study, due to the multiple tests that needed to be performed; we examined a total of 15 metabolites across two different

facets of language using two separate methods. In addition, some of the groups had a small number of individuals, despite our efforts to choose classes that had greater than 10 percent of the total study population. Because of these two factors, it is possible that we found false negatives in our analysis; that is, the fact that we did not find many statistically significant associations does not mean that phthalates are not without risk. It is possible that any given phthalate (or other endocrine disrupting exposure) could truly support a higher uptake of language skills and decrease the risk of falling into a less optimal language development trajectory. Yet this hypothesis is not consistent with prior literature, which primarily shows that phthalate exposure leads to language development, and so these results should not be taken to indicate that some phthalates support neurodevelopment.

A strength of our approach over most previous publications was the use of biomarkers collected at more than one time period, allowing us to obtain a more accurate representation of phthalate exposure during the entire pregnancy. The areas of the brain that are most likely related to language appear to develop during the second and third trimesters (200,201), and we collected exposure data during those time periods. However, no postnatal samples were collected, which reduces our ability to determine whether phthalate exposure that occurred following birth may have impacted language development. Because some phthalate exposures arise from behavioral patterns that are relatively stable over time, prenatal exposure may, for some phthalates, serve as a proxy of phthalate exposure following birth, but this is imperfect and with unknown degrees of error. In addition, phthalate metabolites tend to have short

half-lives, in the span of hours, which may limit our ability to determine if the levels of metabolites measured truly reflect actual phthalate exposure (202,203).

We compared phthalate measurements in our study to the National Health and Nutrition Examination Survey (NHANES) laboratory data, combining the reported data from the 2011 to 2012 and 2013 to 2014 reports (n= 5171). We used a two-sided Wilcox Rank Sum test to assess whether the phthalate measures significantly differed from each other. As seen in Table 3.13, we found that levels of nearly all phthalates in our study were higher than levels found in NHANES. There may be several explanations for these discrepancies. The population in our study is not nationally representative, but rather concentrated around a handful of states. While we did not examine statewide averages, it may be possible that the states our participants resided in have generally higher phthalate levels than the national average. Furthermore, because our population had higher concentrations of phthalates than a nationally representative sample, there may be a possibility that the effects observed in this study may not be as pronounced in a more generalizable population.

Di-n-butyl phthalate, or DBP, is one of the more well-studied phthalates. A LMW phthalate, it's commonly used as a plasticizer and can be found in shower curtains, raincoats, food wraps, bowls, car interiors, vinyl fabrics, floor tiles, and other products (204). We found that its metabolites tended to increase the risk of falling into a suboptimal language trajectory for expressive language and decrease the risk of falling into a suboptimal language trajectory for receptive language. This is consistent with prior studies, which show both an increase and decrease in risk depending on the specific measure of language used. Bornehag and Hyland both found that a DBP

metabolite resulted in increased risk at language delays, while Olesen found that it largely increased risk, though it decreased risk for girls (74,105,106). Gascon, on the other hand, found that it improved language development (97). However, results from these studies did not meet statistical significance, much like our own.

Di-ethyl phthalate, or DEP, is a LMW phthalate commonly found in plastics, personal care products, industrial materials (e.g. dyes, rocket propellant, sealants, and lubricants), and medical products (83). We found that its metabolite was linked with higher risk of falling into an abnormal language trajectory for both expressive and receptive language. This is consistent with some prior studies that have examined DEP. Olesen, Miodovnik, and Ramos all found that a DEP metabolite increased risk for various measures of language ability (105,205,206). However, some studies have found a protective effect, such as Bornehag, Gascon, and Polanska (74,97,108). However, just like our results, none of these studies achieved statistical significance.

Butyl benzyl phthalate, or BzBP, is a HMW phthalate that is most commonly found in toys, bag, gloves, and plastic tubing in order to improve flexibility and make such products soft and malleable (207). We found that its metabolite increased the risk of belonging to an abnormal language trajectory for expressive language. For receptive language, it increased the risk of falling into the Late Growth trajectory. Our results are fairly consistent with prior literature. Bornehag, Hyland, and Polanska all similarly found increased risk for various measures of language (74,106,108). However, Olesen found increased risk for males, but decreased risk for females, and Gascon found decreased risk (97,105). None of the above studies reached statistical significance, however.

Di(2-ethylhexyl) phthalate, or DEHP, is also one of the more well-studied phthalates. A HMW phthalate, it's commonly found in medical devices such as medical tubing and catheters, along with soft plastic products like toys and infant products (82). We found that its metabolites tended to decrease the risk of falling into a suboptimal language trajectory for both expressive and receptive language. This is inconsistent with most other studies that have examined this phthalate and its metabolites. Bornehag, Hyland, and Polanska all found increased risk for language delays (74,106,108). Olesen and Huang both found increased risk in some instances (Olesen for males only, Huang for Verbal Comprehension on the Wechsler Intelligence Scale for Children -IV), with decreased risk in others (Olesen for females only, Huang for Wechsler Preschool and Primary Scale of Intelligence-Revised Verbal IQ) (103,105). Most of the studies above did not meet statistical significance, save for mEOHP in Huang's study for increased risk for lower verbal comprehension.

Most phthalates in prior studies did not reach statistical significance, much like our study. Of the studies that did find significance, Huang found that miBP, a metabolite of di-isobutyl phthalate (DiBP), and mEOHP, a metabolite of DEHP, were statistically significant, significantly lowering scores on the Wechsler Intelligence Scale for Children-IV Verbal Comprehension Index (103). However, it should be noted that they did not examine prenatal exposure, but instead children who were exposed to products that were tainted by DEHP between 3 and 12 years of age. Olesen found several statistically significant results, all of which increased risk in males for either the MacArthur-Bates Communicative Development Inventories Vocabulary Scores or Complexity Scores (105). These included the metabolites of DEHP, metabolites of DEP, and metabolites of

BzBP. Bornehag et al. found that metabolites of DBP and BzBP were associated with higher risk of language delay, though other phthalates examined returned null results (74). However, they only examined a single time point and used a method of measuring language development that is noticeably unique, examining number of words known rather than utilizing a psychometric test. Finally, Ramos found some statistically significant results in their path analysis, with organophosphates being associated with lower communication and expressive language in both parent and teacher reported models (206).

Loftus et al. used a pregnancy cohort to examine the effects of phthalate exposure on different neurodevelopmental outcomes. Uniquely, they utilized weighted quantile sum (WQS) regression as their statistical method. While this allowed for examination of phthalate mixtures, there are some limitations to WQS regression. Loftus et al. noted that the WQS regression can't account for "relative toxicity of mixture components"; in other words, they were unable to determine individual risk for each of the phthalates used in the mixture. In addition, WQS regression may have lower accuracy with highly correlated exposures; depending on the specific metabolites analyzed, correlation may be high enough that diminished accuracy may be of concern (208).

We explored the possibility of sex modification via the inclusion of an interaction term of phthalate metabolite and child sex, due to the possibility that these EDCs may be affecting each sex differently. A prior review found that there was inconsistent evidence of sex modification between prenatal phthalate exposure and child language development (209). We found mixed results for different phthalate metabolites. While

we found no evidence of sex modification for expressive language for any metabolite, for receptive language, sex was a suggestive effect modifier for the metabolites mCOP, mNP2, mEHHP, and mEHP, and a significant effect modifier for the metabolite mECPP (Tables 3.13 and 3.14). More specifically, when examining the risk ratios, we find that males generally have a higher risk of belonging in the Tracking trajectory compared to females when exposed to higher concentrations of metabolites of DEHP (i.e. mEHHP, mEHP, and mECPP) but have a lower risk of belonging in the Tracking trajectory compared to females for the metabolites of DiNP (i.e. mCOP, mNP2). Our results with DEHP are consistent with the results found in Olesen et al., where males had increased risk, some being statistically significant, when compared to females (105). However, there have been no prior studies that have examined sex differences for DiNP metabolites. These results suggests that certain phthalates and their metabolites may have a differential impact based on sex, although the specific pathophysiology is unclear. It may be possible that the increased risk in males may be due to phthalates disrupting the development of androgen-dependent structures by affecting testosterone synthesis (210). There is also evidence that suggests that certain phthalates act as antiestrogens and anti-androgens, with DEP, DEHP, DiBP, and DBP acting as antiandrogen, and DBP and DEHP acting as anti-estrogens (211). These same phthalates can also act as xenobiotics, activating estrogen and androgen receptors, with DEP, DEHP, DiBP, DiNP, and DBP activating estrogen receptors, and DBP and DEHP activating and rogen receptors (211). This behavior may help to explain why DiNP results the increase in risk we observed in females compared to males; the activation of estrogen receptors may be influencing neurodevelopment, resulting in an increased risk

of language impairment. Further studies should focus on examining these sex differences and potentially explore further routes or biological pathways that may explain these differences.

Our results examining sex differences may intersect with known sex differences in language within children with autism. Some studies have indicated a sex difference, with some indicating that girls have worse language skills than boys (212–214), while others indicate the reverse (215,216). Our results indicated that prenatal exposure to certain phthalates may decrease the risk of females of falling into an abnormal language trajectory compared to males, while exposure to another may increase that risk. This may help to explain the disparate results from prior literature, namely that the differences observed may be caused by differences in androgen or estrogen levels during development.

Strengths of our study include the use of language development trajectories, rather than single time point outcomes, which allowed for a more detailed and accurate examination of how neurodevelopmental outcome classification affects language development. In addition, the use of repeated measurements reduced the possibility of outcome misclassification, as there are multiple points in time where the outcome is assessed, reducing the possibility of incorrectly assigning an individual to a wrong trajectory class. We used a total of four time points, giving us a robust amount of data to use to construct our trajectories classes. Second, the use of cohorts which prospectively follow siblings of children with autism allowed for a greater ability to detect abnormal language development trajectories that may be less common in unselected cohorts.

These cohorts have a heightened prevalence for neurodevelopmental delays compared to the general population.

Because all children in our study have an older sibling with autism, our results may not be applicable to the general population. In addition, while the MSEL is considered to be a suitable psychometric test for measuring expressive and receptive language (185) and has been shown to have good construct validity (186), language is a complex phenotype that develops throughout a child's early development period. Our analysis is limited to the facets of language development that the MSEL measures: namely expressive and receptive language. Future research should explore other facets of language development, such as vocabulary, grammatical development, or semantics. Our analysis was limited to the first 36 months of age, so we are unable to extrapolate language trajectories after this period. Due to the nature of LCGA analysis, our results are sensitive to the number of classes selected and our total sample size. While our sample size is sufficiently large enough to utilize LCGA, selecting for a different number of classes or changing the sample size may alter the results. However, we are confident that the overall direction of effect would remain consistent between various class choices.

In conclusion, we explored how prenatal exposure to phthalates influences language development trajectories and found that most low molecular weight phthalates increased risk and most high molecular weight phthalates decreased risk for belonging to abnormal language trajectories, though most were not statistically significant. In addition, we found evidence that sex may be acting as a modifier for receptive language. While our study may not be generalizable due to the nature of the cohorts

used, our use of multiple exposure and outcome measures, along with the use of LCGA to explore trajectory growth, are considerable advantages to our study. Future studies should continue to explore potential associations, with an emphasis on exploring potential sex modification and mixture effects.

Tables and Figures

Table 3.1: Mean and Standard Deviation of 36 Month Mullen Scores for Included Population by Child Characteristics

	Expressive language raw score (36 months)	Receptive language raw score (36 months)
Cohort MARBLES (N=203)	30.9 (6.31)	30.3 (5.63)
EARLI (N=159)	31.5 (5.86)	31.0 (4.80)
Child Sex		
Female (N=156)	32.2 (4.99)	31.6 (4.19)
Male (N=206)	30.3 (6.76)	30.0 (5.89)
Maternal age		
<30 years (N=70)	29.8 (6.00)	29.7 (4.56)
31 to 35 years (N=142)	31.5 (6.15)	30.7 (5.77)
36 to 40 years (N=104)	31.1 (5.82)	31.1 (4.85)

>40 years (N=49)	32.1 (6.72)	31.1 (5.65)
Maternal Race/Ethnicity		
Non-Hispanic White (N=203)	31.7 (6.02)	31.1 (5.34)
Black/African American (N=24)	30.4 (6.09)	29.9 (4.09)
Hispanic (N=68)	30.0 (5.95)	28.9 (5.39)
Other/Multiracial (N=67)	31.0 (6.56)	31.5 (5.05)
Maternal Educational Attainment		
High School/GED or Less (N=32)	29.0 (6.42)	28.3 (4.74)
Some college (N=128)	30.2 (6.03)	29.7 (5.25)
Bachelor's degree (N=123)	32.2 (5.55)	31.5 (5.11)
Graduate or Professional degree (N=79)	31.8 (6.65)	32.0 (5.27)

Homeownership

Rent (N=131)	30.3 (6.76)	29.8 (5.83)	
Own (N=231)	31.6 (5.70)	31.2 (4.89)	
Birth Season			
Spring (n=71)	31.3 (5.36)	30.4 (4.69)	
Summer (n=111)	30.6 (6.64)	29.9 (5.78)	
Fall (n=94)	31.8 (6.42)	31.6 (5.32)	
Winter (n=86)	30.9 (5.69)	30.8 (4.95)	
Overall (N=362)	31.1 (6.12)	30.7 (5.28)	

Footnotes for Table 3.1: Mean and standard deviations were calculated after removing missing values

	Included (N=378)	Excluded (N=259)	Overall (N=621)
Child Sex			
Female	156 (43.1%)	124 (50.0%)	280 (45.9%)
Male	206 (56.9%)	124 (50.0%)	330 (54.1%)
Missing	0	11	11
Maternal Age			
20 to 30 years	70 (19.3%)	56 (22.3%)	102 (16.6%)
31 to 35 years	142 (39.2%)	102 (40.6%)	244 (39.8%)
36 to 40 years	101 (27.9%)	76 (30.3%)	177 (28.9%)
40 to 49 years	49 (13.5%)	17 (6.8%)	66 (10.8%)
Missing	0	8	8
Maternal Race/Ethnicity			
Non-Hispanic White	203 (56.5%)	100 (49.2%)	303 (53.9%)
Black/African American	24 (6.7%)	15 (7.4%)	39 (6.9%)
Hispanic	66 (18.4%)	46 (22.7%)	112 (19.9%)
Other/Multiracial	66 (18.4%)	42 (20.7%)	108 (19.3%)
Missing	3	56	59

Table 3.2: Comparison of characteristics of included versus excluded individuals

	Included (N=378)	Excluded (N=259)	Overall (N=621)
Maternal Educational Attainment			
High School/GED or less	32 (8.9%)	22 (11.1%)	36 (6.4%)
Some college	128 (35.4%)	73 (36.9%)	201 (36%)
Bachelor's degree	122 (33.8%)	49 (24.7%)	171 (30.6%)
Graduate or Professional degree	79 (21.9%)	54 (27.3%)	133 (23.8%)
Missing	1	61	62
Homeownership			
Rent	128 (36.7%)	90 (46.2%)	218 (40.1%)
Own	221 (63.3%)	105 (53.8%)	326 (59.9%)
Missing	13	64	77
Expressive language raw score (6 months)			
Mean (SD)	6.14 (1.08)	6.66 (1.53)	6.32 (1.28)
Median [IQR]	6.00 [0]	6.00 [1.00]	6.00 [1.00]
Missing	57	98	155
Receptive language raw score (6 months)			
Mean (SD)	7.84 (1.52)	8.14 (1.58)	7.95 (1.55)

	Included (N=378)	Excluded (N=259)	Overall (N=621)
Median [IQR]	8.00 [2.00]	8.00 [2.00]	8.00 [2.00]
Missing	57	98	155
Expressive language raw score (12 months)			
Mean (SD)	11.8 (2.66)	12.4 (2.48)	12.0 (2.62)
Median [IQR]	12.0 [4.00]	13.0 [3.00]	12.0 [4.00]
Missing	22	98	120
Receptive language raw score (12 months)			
Mean (SD)	12.5 (2.03)	12.5 (2.17)	12.5 (2.07)
Median [IQR]	13.0 [3.00]	13.0 [3.00]	13.0 [3.00]
Missing	22	98	120
Expressive language raw score (24 months)			
Mean (SD)	20.4 (4.67)	21.0 (5.20)	20.6 (4.85)
Median [IQR]	21.0 [6.00]	21.0 [6.00]	21.0 [6.00]
Missing	144	154	298
Receptive language raw score (24 months)			

	Included (N=378)	Excluded (N=259)	Overall (N=621)
Mean (SD)	23.2 (5.46)	23.1 (5.65)	23.2 (5.51)
Median [IQR]	25.0 [7.00]	25.0 [7.00]	25.0 [7.00]
Missing	145	154	299
Expressive language raw score (36 months)			
Mean (SD)	31.2 (6.12)	30.2 (7.22)	31.0 (6.36)
Median [IQR]	32.0 [7.00]	32.0 [8.00]	32.0 [7.00]
Missing	3	165	168
Receptive language raw score (36 months)			
Mean (SD)	30.6 (5.29)	29.7 (6.83)	30.4 (5.65)
Median [IQR]	31.0 [5.75]	30.0 [5.75]	31.0 [5.00]
Missing	4	165	169

Footnote for Table 3.2: Percentages were calculated after removing missing individuals from the sample
	Non-imputed (N=362)	Imputed (N=362)
Child Sex	Non-imputed (N=362)Imput (N=36)156 (43.1%)156 (43)206 (56.9%)206 (56)206 (56.9%)206 (56)70 (19.3%)70 (19)142 (39.2%)142 (39)101 (27.9%)101 (27)49 (13.5%)49 (13)203 (56.5%)203 (56)24 (6.7%)24 (6.6)66 (18.4%)68 (18)66 (18.4%)67 (18)33	
Female	156 (43.1%)	156 (43.1%)
Male	206 (56.9%)	206 (56.9%)
Maternal Age		
20 to 30 years	70 (19.3%)	70 (19.3%)
31 to 35 years	142 (39.2%)	142 (39.2%)
36 to 40 years	101 (27.9%)	101 (27.9%)
40 to 49 years	49 (13.5%)	49 (13.5%)
Maternal Race/Ethnicity		
Non-Hispanic White	203 (56.5%)	203 (56.1%)
Black/African American	24 (6.7%)	24 (6.6%)
Hispanic	66 (18.4%)	68 (18.8%)
Other/Multiracial	66 (18.4%)	67 (18.5%)
Missing	3	
Maternal Educational Attainment		
High School/GED or less	32 (8.9%)	32 (8.8%)

Table 3.3: Comparison of Imputed versus Non-Imputed Covariates

	Non-imputed (N=362)	Imputed (N=362)
Some college	128 (35.4%)	128 (35.4%)
Bachelor's degree	122 (33.8%)	123 (34.0%)
Graduate or Professional degree	79 (21.9%)	79 (21.8%)
Missing	1	
Homeownership		
Rent	128 (36.7%)	131 (36.2%)
Own	221 (63.3%)	231 (63.8%)
Missing	13	
Expressive language raw score (6 months)		
Mean (SD)	6.14 (1.08)	6.09 (1.11)
Median [IQR]	6.00 [0]	6.00 [0]
Missing	57	
Receptive language raw score (6 months)		
Mean (SD)	7.84 (1.52)	7.90 (1.51)
Median [IQR]	8.00 [2.00]	8.00 [2.00]
Missing	57	

	Non-imputed (N=362)	Imputed (N=362)
Expressive language raw score (12 months)		
Mean (SD)	11.8 (2.66)	11.7 (2.68)
Median [IQR]	12.0 [4.00]	12.0 [4.00]
Missing	22	
Receptive language raw score (12 months)		
Mean (SD)	12.5 (2.03)	12.4 (2.01)
Median [IQR]	13.0 [3.00]	13.0 [3.00]
Missing	22	
Expressive language raw score (24 months)		
Mean (SD)	20.4 (4.67)	20.3 (4.84)
Median [IQR]	21.0 [6.00]	20.0 [5.75]
Missing	144	
Receptive language raw score (24 months)		
Mean (SD)	23.2 (5.46)	23.1 (5.43)
Median [IQR]	25.0 [7.00]	25.0 [7.00]

	Non-imputed (N=362)	Imputed (N=362)
Missing	145	
Expressive language raw score (36 months)		
Mean (SD)	31.2 (6.12)	31.1 (6.12)
Median [IQR]	32.0 [7.00]	32.0 [7.00]
Missing	3	
Receptive language raw score (36 months)		
Mean (SD)	30.6 (5.29)	30.7 (5.28)
Median [IQR]	31.0 [5.75]	31.0 [5.75]
Missing	4	

Footnote for Table 3.3: Percentages were calculated after removing missing individuals from the

sample

Table 3.4: Relative Risk Ratios and 95% Confidence Intervals for Phthalate Metabolites and Expressive and Receptive Language Trajectories

				Expressive	Language			Receptive	Language	
Molecular	Parent Compound	Metabolite	Trac	king	Low G	Growth	Trac	king	Late G	irowth
Weight		metabolita	One Step	Two Step	One Step	Receptive Language V Growth Tracking Late Grow V Two Step One Step Two Step One Step Two Step 1.24 0.89 0.85 0.91 Two 1.24 0.89 0.85 0.91 Two 1.24 0.89 0.85 0.91 Two 1.26 0.99 0.90 1.03 (0.94, (0.71, (0.69, (0.73, 1.67))))))))))))))))))))))))))))))))))))	Two Step			
	DBP	mBP	1.10 (0.83, 1.46)	1.08 (0.85, 1.37)	1.46 (1.07, 2.00)	1.24 (0.93, 1.65)	0.89 (0.65, 1.22)	0.85 (0.65, 1.11)	0.91 (0.64, 1.28)	0.92 (0.67, 1.26)
	ושט	MHBP	1.08 (0.84, 1.40)	1.13 (0.90, 1.42)	1.32 (0.99, 1.77)	1.26 (0.94, 1.67)	0.99 (0.71, 1.36)	0.90 (0.69, 1.18)	1.03 (0.73, 1.46)	1.05 (0.78, 1.43)
Low Molecular Weight	DEP	mEP2	1.01 (0.92, 1.12)	1.02 (0.93, 1.13)	1.03 (0.92, 1.15)	1.01 (0.90, 1.14)	1.27 (0.92, 1.74)	1.23 (0.93, 1.62)	1.20 (0.87, 1.67)	1.24 (0.93, 1.64)
	Dipp	MHiBP	0.96 (0.71, 1.28)	0.99 (0.78, 1.25)	0.90 (0.59, 1.38)	0.81 (0.57, 1.15)	0.93 (0.66, 1.30)	0.93 (0.68, 1.27)	1.00 (0.70, 1.42)	0.95 (0.66, 1.36)
		miBP	1.01 (0.74, 1.37)	1.00 (0.79, 1.27)	0.98 (0.69, 1.39)	0.85 (0.60, 1.20)	0.98 (0.71, 1.37)	0.96 (0.70, 1.31)	0.99 (0.70, 1.41)	0.90 (0.63, 1.30)
	BzBP	mBzP2	1.17 (0.90, 1.51)	1.14 (0.91, 1.43)	1.21 (0.89, 1.64)	1.20 (0.89, 1.62)	0.95 (0.67, 1.35)	0.93 (0.70, 1.22)	1.16 (0.82, 1.65)	1.14 (0.84, 1.55)
Low Molecular Weight High Molecular Weight	DiDP	mCNP	0.88 (0.79, 0.98)	1.00 (0.96, 1.04)	0.91 (0.80, 1.03)	0.98 (0.91, 1.05)	0.98 (0.94, 1.02)	0.98 (0.95, 1.02)	0.92 (0.82, 1.03)	0.95 (0.88, 1.03)
	DiNP	mCOP	0.94 (0.82, 1.08)	0.95 (0.83, 1.09)	0.92 (0.74, 1.13)	0.91 (0.74, 1.11)	0.91 (0.77, 1.07)	0.94 (0.80, 1.11)	0.85 (0.68, 1.05)	0.85 (0.68, 1.06)

	mNP2	0.95 (0.85, 1.06)	0.97 (0.87, 1.08)	0.99 (0.86, 1.14)	0.97 (0.83, 1.12)	0.93 (0.82, 1.05)	0.95 (0.85, 1.08)	0.86 (0.72, 1.02)	0.86 (0.72, 1.04)
DOP, DBP, other HMW phthalates	mCPP	0.98 (0.91, 1.06)	0.98 (0.92, 1.05)	0.93 (0.82, 1.05)	0.96 (0.88, 1.06)	0.94 (0.87, 1.02)	0.98 (0.91, 1.05)	0.95 (0.86, 1.04)	0.94 (0.84, 1.04)
	mECPP	0.92 (0.82, 1.04)	0.94 (0.86, 1.02)	0.80 (0.62, 1.02)	0.81 (0.65, 1.00)	0.87 (0.78, 0.97)	0.87 (0.78, 0.96)	0.80 (0.66, 0.96)	0.80 (0.67, 0.96)
	mEHHP	0.96 (0.88, 1.04)	0.96 (0.90, 1.02)	0.82 (0.65, 1.03)	0.81 (0.64, 1.01)	0.92 (0.86, 1.00)	0.92 (0.86, 0.99)	0.87 (0.76, 1.00)	0.87 (0.76, 0.99)
DEHP	mEHP	0.97 (0.88, 1.06)	0.98 (0.91, 1.05)	0.81 (0.62, 1.06)	0.80 (0.61, 1.04)	0.94 (0.87, 1.02)	0.93 (0.86, 1.01)	0.87 (0.74, 1.03)	0.87 (0.74, 1.01)
	mEOHP	0.97 (0.89, 1.05)	0.97 (0.92, 1.04)	0.83 (0.66, 1.06)	0.82 (0.65, 1.03)	0.94 (0.87, 1.01)	0.93 (0.88, 1.00)	0.88 (0.77, 1.02)	0.89 (0.78, 1.01)
	ΣDEHP	0.95 (0.87, 1.04)	0.96 (0.89, 1.03)	0.82 (0.64, 1.03)	0.81 (0.65, 1.01)	0.91 (0.84, 0.99)	0.91 (0.84, 0.98)	0.85 (0.73, 0.99)	0.85 (0.74, 0.99)

Footnotes for Table 3.4: High Growth was used as the referent group; All phthalates were winsorized to the 99th percentile; all models were adjusted for site specific Z-score for urinary dilution measure (specific gravity or creatinine), cohort, child sex, maternal age, maternal race/ethnicity, home-ownership, and maternal educational attainment

Number of classes	Log- likelihood	SABIC	BIC	Entropy	ICL	
1	-4302.09	8617.99	8633.85	1	8633.85	-
2	-4008.34	8044.31	8076.04	0.90*	8101.04	
3	-3905.41	7852.24	7899.84	0.82	7973.67	
4	-3857.71	7770.66	7834.11	0.82	7928.66*	
5	-3839.51	7748.08	7827.4*	0.79	7957.89	
6	-3829.19	7741.24	7836.42	0.81	7967.04	
7	-3821.89*	7740.46*	7851.51	0.79	8002.91	

Table 3.5: Measures of Model Fit for Expressive Language Latent Class Growth Analysis, Two-Step Model

Footnote for Table 3.5: Higher log-likelihood, lower SABIC (sample adjusted BIC), lower BIC (Bayesian Inclusion Criterion), higher entropy, and lower ICL (integrated classification likelihood) indicates an overall better fit; * indicates the best model based on the above criteria (single class will always have an entropy of 1 so we select the next highest entropy instead); we prioritized BIC and ICL based on the findings in Diallo et al., 2016; classes had <5% of the population in the 6 class model, so we selected the next best fit model instead

Number of classes	Log- likelihood	SABIC	BIC	Entropy	ICL
1	-4247.94	8509.70	8525.56	1	8525.56
2	-3947.36	7922.34	7954.07	0.929*	7972.757
3	-3893.73	7828.89	7876.48	0.844	7941.112
4	-3883.52	7822.29	7885.75	0.805	7987.721
5	-3880.92	7830.89	7910.21	0.777	8045.602
6	-3670.24*	7423.35*	7518.53*	0.8475	7621.971
7	-3880.92	7858.51	7969.56	0.515	8326.024

Table 3.6: Measures of Model Fit for Receptive Language Latent Class Growth Analysis, Two-Step Model

Footnote for Table 3.6: Higher log-likelihood, lower SABIC (sample adjusted BIC), lower BIC (Bayesian Inclusion Criterion), higher entropy, and lower ICL (integrated classification likelihood) indicates an overall better fit; * indicates the best model based on the above criteria (single class will always have an entropy of 1 so we select the next highest entropy instead); 6 class model failed to converge; we prioritized BIC and ICL based on the findings in Diallo et al., 2016; classes had <5% of the population in the 5 and 7 class models, so we selected the next best fit model instead

		Log-				
Metabolite	Classes	likelihood	SABIC	BIC	Entropy	ICL
mBP	1	-4081.62	8204.03	8251.62	1	8251.62
	2	-3800.66	7685.60	7783.96	0.92	7803.86
	3	-3700.11	7528.01	7677.12	0.82	7747.43
	4	-3674.03	7519.36	7719.23	0.86	7787.20
	5	-3610.86	7436.53	7687.16	0.88	7759.95
mBzP2	1	-4083.89	8208.57	8256.16	1	8256.16
	2	-3802.72	7689.73	7788.08	0.92	7807.69
	3	-3700.86	7529.52	7678.63	0.82	7749.45
	4	-3677.13	7525.56	7725.43	0.77	7841.44
	5	-3620.79	7456.40	7707.03	0.80	7823.00
mCNP	1	-4085.83	8212.44	8260.03	1	8260.03
	2	-3797.79	7679.87	7778.21	0.92	7798.55
	3	-3694.10	7516.00	7665.11	0.83	7731.22
	4	-3643.75	7458.80	7658.67	0.85	7735.95
	5	-3628.33	7471.47	7722.10	0.82	7825.52
mCOP	1	-4084.49	8209.77	8257.36	1	8257.36
	2	-3803.51	7691.32	7789.67	0.92	7809.49
	3	-3702.61	7533.01	7682.12	0.82	7752.31
	4	-3649.82	7470.94	7670.81	0.85	7748.30
	5	-3632.31	7479.44	7730.07	0.79	7850.62
mCPP	1	-4085 59	8211 96	8259 55	1	8259 55
	י 2	-3804 88	7694 04	7792 30	0 02	7812 57
	2	-3703 66	7535 12	7684 23	0.02	7753 75
	0	0100.00	1000.12	1007.20	0.00	1100.10

Table 3.7: Measure of Model Fit for Expressive Language Latent Class Growth Analysis, One-Step Model

	4	-3651.33	7473.97	7673.84	0.84	7755.78
	5	-3631.55	7477.92	7728.55	0.81	7841.41
mECPP	1	-4082.87	8206.52	8254.11	1	8254.11
	2	-3802.31	7688.92	7787.27	0.92	7807.17
	3	-3702.01	7531.81	7680.92	0.82	7750.60
	4	-3684.74	7540.78	7740.65	0.77	7857.42
	5	-3628.84	7472.49	7723.13	0.78	7851.98
mEHHP	1	-4083.14	8207.07	8254.66	1	8254.66
	2	-3802.02	7688.34	7786.68	0.92	7806.29
	3	-3701.97	7531.74	7680.85	0.82	7750.91
	4	-3648.54	7468.38	7668.25	0.84	7749.67
	5	-3626.75	7468.30	7718.93	0.85	7804.45
mEHP	1	-4083.49	8207.77	8255.36	1	8255.36
	2	-3802.06	7688.41	7786.76	0.92	7806.42
	3	-3702.08	7531.96	7681.07	0.82	7751.62
	4	-3647.90	7467.10	7666.97	0.84	7748.16
	5	-3618.38	7451.57	7702.20	0.87	7776.32
mEOHP	1	-4084.01	8208.82	8256.41	1	8256.41
	2	-3802.70	7689.70	7788.05	0.92	7807.56
	3	-3702.59	7532.97	7682.08	0.82	7752.40
	4	-3648.50	7468.30	7668.17	0.85	7743.74
	5	-3617.72	7450.25	7700.88	0.88	7772.26
mEP2	1	-4085.72	8212.22	8259.81	1	8259.81
	2	-3804.08	7692.46	7790.81	0.92	7810.40
	3	-3704.36	7536.52	7685.63	0.82	7756.11
	4	-3648.32	7467.94	7667.81	0.86	7740.47

	5	-3615.25	7445.30	7695.93	0.88	7767.53
MHBP	1	-4079.16	8199.11	8246.70	1	8246.70
	2	-3799.51	7683.32	7781.67	0.92	7801.69
	3	-3698.55	7524.89	7674.00	0.83	7742.94
	4	-3644.31	7459.92	7659.79	0.84	7739.74
	5	-3607.75	7430.30	7680.93	0.87	7758.55
MHiBP	1	-4085.69	8212.16	8259.75	1	8259.75
	2	-3804.96	7694.21	7792.55	0.92	7812.68
	3	-3702.47	7532.73	7681.84	0.83	7751.28
	4	-3645.90	7463.10	7662.97	0.84	7742.12
	5	-3625.01	7464.83	7715.46	0.80	7830.30
miBP	1	-4085.74	8212.28	8259.86	1	8259.86
	2	-3804.74	7693.78	7792.13	0.92	7811.43
	3	-3702.46	7532.72	7681.83	0.82	7752.15
	4	-3646.12	7463.54	7663.41	0.84	7744.42
	5	-3625.25	7465.31	7715.94	0.81	7829.36
mNP2	1	-4085.22	8211.22	8258.81	1	8258.81
	2	-3804.08	7692.44	7790.79	0.92	7810.44
	3	-3702.07	7531.94	7681.05	0.82	7750.81
	4	-3651.48	7474.27	7674.14	0.84	7756.12
	5	-3634.27	7483.36	7734.00	0.77	7870.71
DEHP	1	-4083.24	8207.27	8254.86	1	8254.86
	2	-3802.29	7688.88	7787.23	0.92	7806.96
	3	-3702.17	7532.14	7681.25	0.82	7751.29
	4	-3648.07	7467.45	7667.32	0.85	7742.69
	5	-3617.92	7450.65	7701.28	0.85	7786.97

Footnotes for Table 3.7: Higher log-likelihood, lower SABIC (sample adjusted BIC), lower BIC (Bayesian Inclusion Criterion), higher entropy, and lower ICL (integrated classification likelihood) indicates an overall better fit; All phthalates were winsorized to the 99th percentile; all models were adjusted for site specific Z-score for urinary dilution measure (specific gravity or creatinine), cohort, child sex, maternal age, maternal race/ethnicity, home-ownership, and maternal educational attainment; 4 and 5 class models had <10% of population in at least one group, which resulted in those models being dropped from consideration, even if they had better fit measures

		Log-				
Metabolite	Classes	likelihood	SABIC	BIC	Entropy	ICL
mBP	1	-4016.66	8074.12	8121.70	1	8121.70
	2	-3701.74	7487.78	7586.13	0.94	7600.18
	3	-3635.61	7399.01	7548.12	0.89	7590.25
	4	-3465.42	7102.15	7302.02	0.91	7347.13
	5	-3426.35	7067.50	7318.14	0.85	7407.87
mBzP2	1	-4012.83	8066.44	8114.03	1	8114.03
	2	-3702.46	7489.20	7587.55	0.94	7602.32
	3	-3638.71	7405.22	7554.33	0.86	7609.58
	4	-3463.08	7097.46	7297.33	0.91	7344.58
	5	-3425.24	7065.28	7315.91	0.87	7390.53
mCNP	1	-4017.48	8075.75	8123.34	1	8123.34
-	2	-3705.69	7495.68	7594.03	0.94	7608.81
	3	-3643.77	7415.33	7564.44	0.86	7621.13
	4	-3466.70	7104.70	7304.57	0.90	7353.36
	5	-3427.69	7070.18	7320.81	0.87	7398.07
mCOP	1	-4016 47	8073 73	8121.31	1	8121.31
meer	2	-3703.74	7491.77	7590.12	0.94	7605.42
	-	-3640.59	7408.97	7558.08	0.85	7616.31
	4	-3466.41	7104.12	7303.99	0.90	7354.36
	5	-3434.72	7084.24	7334.87	0.85	7421.26
mCPP	1	-4016.53	8073.86	8121.44	1	8121.44
	2	-3701.96	7488.20	7586.55	0.94	7602.29
	3	-3636.49	7400.78	7549.89	0.87	7603.27

Table 3.8: Measure of Model Fit for Receptive Language Latent Class Growth Analysis, One-Step Model

	4	-3617.65	7406.61	7606.48	0.79	7713.76
	5	-3601.80	7418.42	7669.05	0.84	7764.51
mECPP	1	-4014.64	8070.07	8117.66	1	8117.66
	2	-3702.93	7490.16	7588.51	0.94	7603.32
	3	-3639.56	7406.92	7556.03	0.86	7610.78
	4	-3468.61	7108.52	7308.39	0.90	7356.53
	5	-3426.34	7067.48	7318.12	0.88	7389.98
mEHHP	1	-4015.60	8071.98	8119.56	1	8119.56
	2	-3704.07	7492.44	7590.79	0.94	7605.67
	3	-3640.88	7409.55	7558.66	0.86	7614.38
	4	-3468.91	7109.13	7309.00	0.90	7357.76
	5	-3442.49	7099.79	7350.42	0.85	7439.93
mEHP	1	-4016.42	8073.63	8121.22	1	8121.22
	2	-3704.76	7493.81	7592.16	0.94	7607.1
	3	-3641.67	7411.14	7560.25	0.86	7617.68
	4	-3464.32	7099.94	7299.81	0.90	7348.76
	5	-3601.12	7417.05	7667.68	0.76	7806.25
mEOHP	1	-4016.47	8073.72	8121.31	1	8121.31
	2	-3704.97	7494.24	7592.58	0.94	7607.48
	3	-3641.31	7410.42	7559.53	0.86	7615.51
	4	-3484.59	7140.48	7340.35	0.90	7388.39
	5	-3447.10	7109.00	7359.63	0.83	7458.90
mEP2	1	-4017.53	8075.85	8123.44	1	8123.44
	2	-3705.67	7495.63	7593.98	0.95	7607.75
	3	-3642.22	7412.24	7561.35	0.85	7620.57
	4	-3470.66	7112.62	7312.49	0.90	7363.41

	5	-3437.22	7089.26	7339.89	0.85	7427.09
MHBP	1	-4015.22	8071.22	8118.81	1	8118.81
	2	-3703.01	7490.31	7588.66	0.94	7602.65
	3	-3638.42	7404.65	7553.76	0.89	7597.47
	4	-3466.86	7105.03	7304.90	0.90	7352.79
	5	-3427.86	7070.53	7321.16	0.85	7406.98
MHiBP	1	-4017.28	8075.34	8122.93	1	8122.93
	2	-3705.24	7494.77	7593.12	0.94	7607.53
	3	-3642.25	7412.29	7561.40	0.86	7618.52
	4	-3469.67	7110.64	7310.51	0.90	7360.60
	5	-3606.55	7427.90	7678.54	0.78	7806.04
miBP	1	-4017.49	8075.76	8123.35	1	8123.35
	2	-3704.63	7493.56	7591.91	0.94	7606.03
	3	-3641.38	7410.56	7559.67	0.86	7615.58
	4	-3622.53	7416.36	7616.23	0.85	7690.92
	5	-3432.45	7079.70	7330.33	0.86	7411.89
mNP2	1	-4016.53	8073.84	8121.43	1	8121.43
	2	-3703.54	7491.36	7589.71	0.94	7604.87
	3	-3640.28	7408.36	7557.47	0.85	7616.24
	4	-3466.74	7104.78	7304.65	0.90	7356.12
	5	-3429.68	7074.18	7324.81	0.85	7409.87
DEHP	1	-4015.55	8071.89	8119.48	1	8119.48
	2	-3704.00	7492.29	7590.63	0.94	7605.49
	3	-3640.63	7409.07	7558.18	0.86	7613.79
	4	-3628.18	7427.66	7627.53	0.80	7729.59
	5	-3438.38	7091.56	7342.19	0.87	7420.80

Footnotes for Table 3.8: Higher log-likelihood, lower SABIC (sample adjusted BIC), lower BIC (Bayesian Inclusion Criterion), higher entropy, and lower ICL (integrated classification likelihood) indicates an overall better fit; All phthalates were winsorized to the 99th percentile; all models were adjusted for site specific Z-score for urinary dilution measure (specific gravity or creatinine), cohort, child sex, maternal age, maternal race/ethnicity, home-ownership, and maternal educational attainment; 4 and 5 class models had <10% of population in at least one group, which resulted in those models being dropped from consideration, even if they had better fit measures

	Two-Step			
Metabolite	Categorizations	One-S	tep Catego	rizations
		High		Low
mBP		Growth	Tracking	Growth
	High Growth	100	8	0
	Tracking	8	179	3
	Low Growth	0	6	58
		High		Low
mBzP2		Growth	Tracking	Growth
	High Growth	100	8	0
	Tracking	9	180	1
	Low Growth	0	7	57
		-		
		High		Low
mCNP		Growth	Tracking	Growth
	High Growth	99	9	0
	Tracking	12	174	4
	Low Growth	0	4	60
		Ŭ		00
		Hiah		Low
mCOP		Growth	Tracking	Growth
	High Growth	99	9	0
	Tracking	8	181	4
	Low Growth	0	5	60
		Ū	Ũ	00
		High		Low
mCPP		Growth	Tracking	Growth
	High Growth	100	8	0
	Tracking	9	180	1

Table 3.9: Comparison of Class Assignment between One-Step and Two-Step Models for Expressive Language

	Low Growth	0	5	59
mECPP		High Growth	Tracking	Low Growth
	High Growth	100	8	0
	Tracking	10	179	1
	Low Growth	0	5	59
mEHHP		High Growth	Tracking	Low Growth
	High Growth	99	9	0
	Tracking	9	180	1
	Low Growth	0	5	59
		High		Low
mEHP		Growth	Tracking	Growth
	High Growth	100	8	0
	Tracking	10	179	1
	Low Growth	0	6	58
		High		Low
mEOHP		Growth	Tracking	Growth
	High Growth	100	8	0
	Tracking	9	180	1
	Low Growth	0	5	59
		High		Low
mEP2		Growth	Tracking	Growth
	High Growth	100	8	0
	Tracking	8	181	1
	Low Growth	0	6	58

		High		Low
MHBP		Growth	Tracking	Growth
	High Growth	100	8	0
	Tracking	8	181	1
	Low Growth	0	6	58
		High		Low
MHiBP		Growth	Tracking	Growth
	High Growth	99	8	0
	Tracking	11	181	1
	Low Growth	0	7	57
		High		Low
miBP		Growth	Tracking	Growth
	High Growth	100	8	0
	Tracking	11	178	1
	Low Growth	0	5	59
		High		Low
mNP2		Growth	Tracking	Growth
	High Growth	99	9	0
	Tracking	9	180	1
	Low Growth	0	7	57
		High		Low
DEHP		Growth	Iracking	Growth
	High Growth	99	9	0
	Tracking	9	180	1
	Low Growth	0	5	59

	Two-Step			
Metabolite	Categorizations	One-S	tep Catego	rizations
		High		Late
mBP		Growth	Tracking	Growth
	High Growth	35	18	0
	Tracking	1	223	3
	Late Growth	1	2	79
		High		Late
mBzP2		Growth	Tracking	Growth
	High Growth	44	9	0
	Tracking	2	223	2
	Late Growth	0	2	80
		High		Late
mCNP		Growth	Tracking	Growth
	High Growth	44	9	0
	Tracking	2	223	2
	Late Growth	0	3	79
		High		Late
mCOP		Growth	Tracking	Growth
	High Growth	45	8	0
	Tracking	2	223	2
	Late Growth	0	2	80
		High		Late
mCPP		Growth	Tracking	Growth
	High Growth	44	9	0
	Tracking	1	223	3

Table 3.10: Comparison of Class Assignment between One-Step and Two-Step Models for Receptive Language

	Late Growth	0	2	80
mECPP		High Growth	Tracking	Late Growth
	Hiah Growth	45	8	0
	Tracking	2	224	1
	Late Growth		2	80
	Late Crowin	Ū	2	00
mEHHP		High Growth	Tracking	Late Growth
	High Growth	45	8	0
	Tracking	2	224	1
	Late Growth	0	2	80
		High		Late
mEHP		Growth	Tracking	Growth
	High Growth	45	8	0
	Tracking	2	224	1
	Late Growth	0	2	80
		Lliab		Lata
mEOHP		Growth	Tracking	Growth
	Hiah Growth	45	8	0
	Tracking	2	224	1
	Late Growth		2	80
	Late Crowin	0	2	00
		High		Late
mEP2		Growth	Tracking	Growth
	High Growth	45	8	0
	Tracking	2	224	1
	Late Growth	0	3	79

МНВР		High Growth	Tracking	Late Growth
	High Growth	38	15	0
	Tracking	1	223	3
	Late Growth	1	2	79
		High		Late
MHiBP		Growth	Tracking	Growth
	High Growth	44	9	0
	Tracking	2	222	3
	Late Growth	0	2	80
		High		Late
miBP		Growth	Tracking	Growth
	High Growth	44	9	0
	Tracking	2	222	3
	Late Growth	0	2	80
		High	-	Late
mNP2		Growth	Iracking	Growth
	High Growth	45	8	0
	Tracking	2	223	2
	Late Growth	0	2	80
		High	Tracking	Late
DEHP		Growth	Ггаскіпд	Growth
		45	8	0
	Iracking	2	224	1
	Late Growth	0	2	80

Footnotes for Table 3.10:

Table 3.11: Sex Modification for Expressive Language

Molecular Weight	Parent Compound	Metabolite	LRT P-value	Tracking (female) (N=81)	Tracking (male) (N=109)	Low Growth (female) (N=16)	Low Growth (male) (N=48)
	מפט	mBP	0.39	1.20 (0.89, 1.63)	0.90 (0.61, 1.31)	1.20 (0.78, 1.85)	1.17 (0.78, 1.76)
	DDP	МНВР	0.23	1.30 (0.95, 1.77)	0.89 (0.61, 1.30)	1.22 (0.77, 1.93)	1.16 (0.78, 1.73)
Low Molecular Weight	DEP	mEP2	0.12	1.03 (0.87, 1.21)	1.03 (0.91, 1.16)	0.36 (0.12, 1.11)	1.05 (0.91, 1.20)
	DiBP	MHiBP	0.55	1.03 (0.87, 1.21)	1.03 (0.91, 1.16)	0.36 (0.12, 1.11)	1.05 (0.91, 1.20)
	DIBF	miBP	0.80	1.06 (0.77, 1.46)	0.93 (0.66, 1.31)	0.90 (0.55, 1.47)	0.79 (0.49, 1.29)
	BzBP	mBzP2	0.69	1.16 (0.87, 1.54)	1.12 (0.77, 1.63)	1.09 (0.68, 1.74)	1.26 (0.83, 1.92)
	DiDP	mCNP	0.29	0.98 (0.94, 1.03)	1.19 (0.87, 1.62)	0.86 (0.62, 1.21)	1.18 (0.86, 1.61)
		mCOP	0.53	1.03 (0.86, 1.23)	0.85 (0.69, 1.04)	0.91 (0.62, 1.34)	0.86 (0.68, 1.10)
High Molecular	DINF	mNP2	0.53	1.03 (0.90, 1.18)	0.87, 0.73, 1.03)	1.05 (0.85, 1.29)	0.88 (0.72, 1.07)
Weight	DOP, DBP, other HMW phthalates	mCPP	0.67	0.98 (0.89, 1.08)	0.98 (0.89, 1.08)	0.79 (0.50, 1.24)	0.99 (0.89, 1.09)
		mECPP	0.17	0.93 (0.83, 1.03)	0.95 (0.82, 1.11)	0.78 (0.51, 1.20)	0.83 (0.63, 1.08)
	DEHP	mEHHP	0.11	0.95 (0.88, 1.03)	0.98 (0.89, 1.09)	0.82 (0.56, 1.21)	0.81 (0.61, 1.06)
		mEHP	0.18	0.97 (0.89, 1.07)	0.98 (0.88, 1.09)	0.88 (0.66, 1.19)	0.74 (0.51, 1.07)

	mEOHP	0.22	0.97 (0.90, 1.05)	0.98 (0.89 1.07)	0.84 (0.56, 1.25)	0.81 (0.61, 1.08)
	ΣDEHP	0.16	0.95 (0.87, 1.04)	0.97 (0.87, 1.09)	0.82 (0.56, 1.20)	0.82 (0.63, 1.07)

Footnotes for Table 3.11: Female and High Growth were defaulted as the referent group; All phthalates were winsorized to the 99th percentile; models were adjusted for site specific Z-score for urinary dilution measure (specific gravity or creatinine), cohort, child sex, maternal age, maternal race/ethnicity, home-ownership, and maternal educational attainment

Table 3.12: Sex Modification for Receptive Language

Molecular Weight	Parent Compound	Metabolite	LRT P- value	Tracking (Female) (N=106)	Tracking (Male) (N=121)	Late Growth (Female) (N=24)	Late Growth (Male) (N=58)
	DPD	mBP	0.18	1.07 (0.72, 1.60)	0.63 (0.42, 0.94)	1.26 (0.79, 2.00)	0.64 (0.41, 1.01)
	DBF	MHBP	0.25	1.09 (0.73, 1.63)	0.69 (0.46, 1.04)	1.34 (0.86, 2.10)	0.78 (0.50, 1.21)
Low Molecular Weight	DEP	mEP2	0.16	1.19 (0.84, 1.68)	1.33 (0.84, 2.10)	0.95 (0.59, 1.53)	1.37 (0.86, 2.17)
	ססיס	MHiBP	0.24	1.19 (0.84, 1.68)	1.33 (0.84, 2.10)	0.95 (0.59, 1.53)	1.37 (0.86, 2.17)
	DIBP	miBP	0.32	1.55 (0.81, 2.96)	0.77 (0.53, 1.11)	1.53 (0.76, 3.09)	0.69 (0.43, 1.12)
	BzBP	mBzP2	0.21	1.11 (0.73, 1.69)	0.75 (0.51, 1.11)	1.44 (0.90, 2. 28)	0.89 (0.59, 1.35)
	DiDP	mCNP	0.36	0.99 (0.95, 1.04)	0.96 (0.90, 1.03)	0.85 (0.62, 1.15)	0.97 (0.89, 1.05)
	DiNP	mCOP	0.06	1.14 (0.87, 1.50)	0.77 (0.62, 0.97)	0.79 (0.44, 1.40)	0.78 (0.61, 1.00)
		mNP2	0.05	1.08 (0.90, 1.23)	0.80 (0.67, 0.97)	0.82 (0.55, 1.21)	0.81 (0.66, 0.99)
High Molecular Weight	DOP, DBP, other HMW phthalates	mCPP	0.11	1.06 (0.92, 1.23)	0.93 (0.85, 1.02)	0.74 (0.44, 1.23)	0.94 (0.84, 1.04)
		mECPP	0.01	0.84 (0.74, 0.96)	0.90 (0.77, 1.06)	0.63 (0.36, 1.10)	0.87 (0.71, 1.08)
	DEHP	mEHHP	0.05	0.90 (0.82, 0.99)	0.95 (0.85, 1.06)	0.74 (0.48, 1.12)	0.92 (0.79, 1.07)
		mEHP	0.08	0.91 (0.82, 1.02)	0.95 (0.85, 1.07)	0.58 (0.30, 1.12)	0.92 (0.79, 1.08)

mEOHP	0.17	0.93 (0.85, 1.01)	0.95 (0.86, 1.05)	0.76 (0.49, 1.16)	0.92 (0.80, 1.07)
ΣDEHP	0.04	0.89 (0.80, 0.99)	0.94 (0.83, 1.06)	0.69 (0.42, 1.12)	0.91 (0.77, 1.07)

Footnotes for Table 3.12: Female and High Growth were defaulted as the referent group; All phthalates were winsorized to the 99th percentile; models were adjusted for site specific Z-score for urinary dilution measure (specific gravity or creatinine), cohort, child sex, maternal age, maternal race/ethnicity, home-ownership, and maternal educational attainment

Phthalate metabolite	Study Population (N=362)	Combined NHANES data (N=5171)	Wilcox Rank Sum P-value
mBP	12.9 [14.5]	11.2 [12.8]	<< 0.001
mEP2	28.4 [42.5]	38.4 [83.0]	<< 0.001
miBP	8.46 [11.0]	8.25 [9.05]	0.75
mBzP2	6.68 [10.9]	5.29 [8.02]	<< 0.001
mCNP	3.74 [4.09]	2.64 [3.19]	<< 0.001
mCOP	23.3 [47.7]	19.4 [38.6]	0.01
mNP2	1.41 [2.53]	1.09 [1.87]	<< 0.001
mECPP	20.2 [27.5]	13.7 [15.2]	<< 0.001
mEHHP	12.7 [16.5]	8.33 [9.81]	<< 0.001
mEHP	2.92 [4.86]	1.53 [1.98]	<< 0.001
mEOHP	10.2 [12.8]	5.46 [6.06]	<< 0.001

Table 3.13: Comparison of Urinary Phthalate Concentrations between NHANES and Study Population

Footnotes for Table 3.13: Due to skewness of data, medians and IQRs were calculated after removing any missing individuals; two-sided Wilcox Rank Sum test was performed using unpaired test; all values were adjusted for creatinine (NHANES and EARLI) or specific gravity (MARBLES)



Figure 3.1: Two-Step Latent Class Growth Analysis Plot for Expressive Language

Footnotes for Figure 3.1: Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.2: Two-Step Latent Class Growth Analysis Plot for Receptive Language

Footnotes for Figure 3.2: Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.3: One-Step Latent Class Growth Analysis Plot for Expressive Language for mBP

Footnotes for Figure 3.3: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.4: One-Step Latent Class Growth Analysis Plot for Expressive Language for mBzP2

Footnotes for Figure 3.4: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.5: One-Step Latent Class Growth Analysis Plot for Expressive Language for mCNP

Footnotes for Figure 3.5: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.6: One-Step Latent Class Growth Analysis Plot for Expressive Language for mCOP

Footnotes for Figure 3.6: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.7: One-Step Latent Class Growth Analysis Plot for Expressive Language for mCPP

Footnotes for Figure 3.7: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.8: One-Step Latent Class Growth Analysis Plot for Expressive Language for mECPP

Footnotes for Figure 3.8: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.9: One-Step Latent Class Growth Analysis Plot for Expressive Language for mEHHP

Footnotes for Figure 3.9: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals


Figure 3.10: One-Step Latent Class Growth Analysis Plot for Expressive Language for mEHP

Footnotes for Figure 3.10: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.11: One-Step Latent Class Growth Analysis Plot for Expressive Language for mEOHP

Footnotes for Figure 3.11: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.12: One-Step Latent Class Growth Analysis Plot for Expressive Language for mEP2

Footnotes for Figure 3.12: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.13: One-Step Latent Class Growth Analysis Plot for Expressive Language for MHBP

Footnotes for Figure 3.13: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.14: One-Step Latent Class Growth Analysis Plot for Expressive Language for MHiBP

Footnotes for Figure 3.14: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.15: One-Step Latent Class Growth Analysis Plot for Expressive Language for miBP

Footnotes for Figure 3.15: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.16: One-Step Latent Class Growth Analysis Plot for Expressive Language for mNP2

Footnotes for Figure 3.16: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.17: One-Step Latent Class Growth Analysis Plot for Expressive Language for DEHP

Footnotes for Figure 3.17: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.18: One-Step Latent Class Growth Analysis Plot for Receptive Language for mBP2

Footnotes for Figure 3.18: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.19: One-Step Latent Class Growth Analysis Plot for Receptive Language for mBzP2

Footnotes for Figure 3.19: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.20: One-Step Latent Class Growth Analysis Plot for Receptive Language for mCNP

Footnotes for Figure 3.20: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.21: One-Step Latent Class Growth Analysis Plot for Receptive Language for mCOP

Footnotes for Figure 3.21: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.22: One-Step Latent Class Growth Analysis Plot for Receptive Language for mCPP

Footnotes for Figure 3.22: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.23: One-Step Latent Class Growth Analysis Plot for Receptive Language for mECPP

Footnotes for Figure 3.23: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.24: One-Step Latent Class Growth Analysis Plot for Receptive Language for mEHHP

Footnotes for Figure 3.24: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.25: One-Step Latent Class Growth Analysis Plot for Receptive Language for mEHP

Footnotes for Figure 3.25: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.26: One-Step Latent Class Growth Analysis Plot for Receptive Language for mEOHP

Footnotes for Figure 3.26: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.27: One-Step Latent Class Growth Analysis Plot for Receptive Language for mEP2

Footnotes for Figure 3.27: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.28: One-Step Latent Class Growth Analysis Plot for Receptive Language for MHBP

Footnotes for Figure 3.28: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.29: One-Step Latent Class Growth Analysis Plot for Receptive Language for MHiBP

Footnotes for Figure 3.29: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.30: One-Step Latent Class Growth Analysis Plot for Receptive Language for miBP

Footnotes for Figure 3.30: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.31: One-Step Latent Class Growth Analysis Plot for Receptive Language for mNP2

Footnotes for Figure 3.31: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals



Figure 3.32: One-Step Latent Class Growth Analysis Plot for Receptive Language for DEHP

Footnotes for Figure 3.32: Model was adjusted for child sex, maternal age at birth, maternal race/ethnicity, highest maternal educational attainment, and birth season; Points indicate mean scores at that age with 95 percent confidence intervals

Chapter 4 - The Impact of Prenatal Air Toxic Exposure on Language Development Trajectories in Siblings of Children with Autism

<u>Abstract</u>

Background: Language development is a critical part of neurodevelopment that is correlated with many neurodevelopmental disorders. We aimed to examine how prenatal air toxic exposure affects early childhood language development, utilizing a robust longitudinal analysis methodology.

Methods: Participants were drawn from the Early Autism Risk Longitudinal Investigation (EARLI) (n=251) and the Markers of Autism Risk in Babies – Learning Early Signs (MARBLES) (n=393) cohorts that recruited pregnant mothers who previously had a child with autism (ASD). Expressive and receptive language development was measured using the Mullen Scales of Early Learning (MSEL) at ages 6,12, 24, and 36 months of age. A total of 53 air toxics were assessed by utilizing census tract data to assign exposures via the National Air Toxics Assessment (NATA) datasets from 2011 and 2014. We used latent class growth analysis (LCGA) to determine language trajectories based on MSEL receptive or expressive language raw scores, prenatal air toxic exposure, cohort, child sex, maternal age, maternal race/ethnicity, homeownership, and maternal educational attainment.

Results: We found 3 trajectories for both expressive and receptive language using both a two-step LCGA approach. Most air toxics were not statistically significant. Acetaldehyde showed significant protective effects for expressive language, decreasing the risk of falling into the Low Growth category for each IQR increase in concentration.

Conclusion: Most air toxics were not statistically significant, with only acetaldehyde showing significant protective effects. Observed trends in risk were

inconsistent with prior literature, which may be due to small sample size and differences in analysis methodology.

Introduction

Language development is a key milestone in early childhood development, playing important roles in later stages of life. Delays or impairments in language development can lead to atypical social and emotional development, poor academic performance, and increase risk of certain neurodevelopmental disorders (Beitchman & Brownlie, 2012; Howlin & Udwin, 2002; Irwin et al., 2002; Johnson et al., 1999; Roulstone et al., 2011; Snowling et al., 2006; Whitehouse et al., 2009). Common social factors that may influence language development include family history, child sex, maternal education, and home environment (28,187,188). However, environmental and chemical factors are still an area that requires additional exploration.

Certain systems are critical to early neurodevelopment, particularly the endocrine system. Because the hormones generated by the endocrine system play critical roles during fetal development, disruption of this system can lead to language development. Environmental pollutants that disrupt this system, known as endocrine disrupting chemicals (EDCs) are therefore of vital interest, having a higher *a priori* suspicion of causing language impairment. As such, understanding environmental causes that may act as EDCs will not only provide a better understanding behind the etiology of language disorders, but also provide stakeholders with vital knowledge to make impactful and lasting policy changes to reduce overall risk.

Air toxics, as defined by the United States Environmental Protection Agency (EPA), are toxic or hazardous air pollutants that cause or may cause serious health effects such as cancer, reproductive effects, and/or adverse environmental and ecological effects (116,117). Because of this broad definition, there are several classes

of chemicals that are considered to be air toxics; the Clean Air Act identifies a total of 187 air toxics the EPA is required to control, though there are many more (117). Polychlorinated biphenyls (PCBs), phthalates, polycyclic aromatic hydrocarbons (PAHs), polybrominated diphenyl ethers (PBDEs), certain heavy metals, nitrogen oxides (NO₂ and NO_x), particulate matter (specifically PM_{2.5}), and certain volatile organic compounds (VOCs) are a few examples of air toxics with known endocrine disrupting effects (61,123–142,217–219). The mechanisms by which these air toxics cause these effects are varied, with some mimicking sex hormones (123,143–145), while others disrupt thyroid hormone pathways (123,124,146–149).

Because many air toxics are considered EDCs, their impact on neurodevelopment, and thus language development, must be considered. Several reviews have found that various EDCs that are considered air toxics impact neurodevelopment and language development specifically. Davis et al. found in their review that PCBs and certain pesticides were negatively associated with language delays, and Suades-Gonzalez et al. found PAHs to be negatively associated with language development (220,221). However, some studies have found that exposure to certain air toxics does not lead to language delays. Guxens et al. did a comprehensive study over six European cohorts, finding no association between NO₂, NO_x, particulate matter, and language development (222). Stingone et al. found that perchloroethylene did not influence standardized language test scores, and Huang et al. and Gascon et al. both found that certain phthalates did not negatively affect language (97,103,223). However, it should be noted that these studies did not specify the primary exposure

source for these EDCs. Rather, these studies examined all exposure, rather than focusing on airborne exposure to these EDCs.

There have been several studies that have primarily focused on airborne exposure to air toxics that are EDCs, rather than all exposure, and their possible connections with neurodevelopment. Exposure to airborne PAHs were found to be associated with a decrease in infant mental development across several studies using various measures of neurodevelopment, including the Bayley's Scales of Infant Development and Raven Coloured Progressive Matrices (21,61,62). Nitrogen dioxide was also found to be negatively associated with infant mental development in Freire et al. and resulted in a decrease in all subscales of the McCarthy Scales of Children's Abilities in Guxens et al. (224,225). Guxens additionally found that benzene was also negatively associated with infant mental development (225). There have also been multiple studies that have looked at hazardous air pollutants and their association with autism spectrum disorder, with some airborne EDCs being associated with ASD, such as certain heavy metals, methylene chloride, styrene, certain conjugates of xylene, and other aromatic solvents (118,119,159,160). A review by Volk et al. found comprehensive evidence that prenatal air pollution exposure is linked to various neurodevelopmental disorders and delays (226). However, there is a lack of literature utilizing trajectory analysis to assess this connection, as well as a lack of literature that specifically examines the impact of air toxics on language development.

We conducted a study based on data collected from two ASD sibling cohorts. We examined the association of prenatal air toxic exposure to language development

trajectories to explore potential associations between those trajectories with prenatal air toxic exposure.

<u>Methods</u>

Population

We included individuals in the Early Autism Risk Longitudinal Investigation (EARLI) and the Markers of Autism Risk in Babies – Learning Early Signs (MARBLES) cohorts (173,174). Briefly, both studies recruited pregnant women who already had a child with a diagnosis of ASD and families whose fathers who had a biological child with ASD, with EARLI recruiting from northeast Maryland, southeast Pennsylvania, and northern California, while MARBLES recruited primarily from northern California. Both studies followed the mother and the expected child longitudinally, collecting demographic, neurodevelopmental, and exposure data at set time intervals. Children in EARLI were born between 2009 and 2013, while children in MARBLES were born between 2006 and 2023. Demographic information was obtained via in -person interviews and questionnaires.

Language Development and Neurodevelopment Measurement

The Mullen Scales of Early Learning (MSEL, 20) was used in order to assess language development, which was administered by trained study staff at 6, 12, 24, and 36 months of age at in-person study visits. Briefly, the MSEL is a standardized psychometric test that is used to measure cognitive development in children ages 3 to 60 months, providing scores on five different domains: gross motor, fine motor, expressive language, receptive language, and visual reception (175). Expressive

language involves the use of words and gestures to convey accurate and appropriate messages, thoughts, and ideas to others. Receptive language involves the ability to understand and process the meaning of language directed towards an individual. We used both expressive and receptive language raw scores to generate our language development trajectories, using the 6, 12, 24 and 36 month assessments.

Air Toxic Measurements

We assessed air toxic exposure by using an emissions-based model from the US EPA: the National Air Toxics Assessment (NATA). The NATA model is run every 3 years, and uses inputs from the National Emissions Inventory to generate the average ambient concentration of multiple air toxics. The national Emission Inventory provides estimate of air emissions of criteria pollutants and air toxics using emission sources, including point sources (e.g. industrial facilities, electric power plants), nonpoint sources (e.g. residential heating, commercial combustion), onroad sources (i.e. emission from onroad vehicles), nonroad sources (e.g. construction equipment, locomotives), and fire sources (227). A hybrid model that combines a regional air-quality model (CMAQ) and dispersion model (AERMOD) is then used to estimate annual average ambient air toxic concentrations across US census tracts (228,229). The NATA model has been shown to be reasonably accurate with its estimations, with one validation study confirming that NATA could predict national medians of most toxics well, though there are some instances where the model may overestimate or underestimate certain types of toxics from specific sources (230–232).

We used both the 2011 and 2014 NATA datasets. A total of 180 air toxics were assessed in 2011 and 181 air toxics in 2014. We selected 82 air toxics with known or

suspected endocrine disrupting effects based on prior literature (Table 4.6). We included air toxics that were measured in both the 2011 and 2014 datasets, leaving 53 total air toxics. Individuals who were born after 2012 (i.e. a birth year of 2013 to 2016) were assigned exposure corresponding to the 2014 NATA dataset, while those born in 2006 to 2012 were assigned exposure corresponding to the 2014 NATA dataset. The NATA datasets are publicly available on the US EPA website

(https://www.epa.gov/national-air-toxics-assessment), and we used Total Concentration for all air toxics from both the 2011 and 2014 datasets, measured in ug/m³ (233,234). Air toxic exposure was assigned via census tract. Home addresses were obtained from study participants at enrollment, which were then used to assign 2010 census tracts to individuals. These census tracts were then used to connect to state-specific NATA datasets for each air toxic exposure. Out of the original 621 individuals, 74 were not assigned a census tract, who were removed from the final analysis, as air toxic exposure could not be assigned.

To assist in the comparison of results and to better examine patterns that may arise due to correlations between air toxics, we grouped air toxics using agglomerative hierarchical clustering based on multiscale bootstrap resampling by using the pvclust function in the *pvclust* package in R. Briefly, agglomerative hierarchical clustering is a nesting algorithm that combines two variables, called clusters, that are the most similar into one larger cluster. This process is repeated until all variables are in one cluster, creating a dendrogram. We used Ward's Method for clustering and used Euclidian distance, defined as $\sqrt{\sum_i (x_i - y_i)^2}$, where x and y are two vectors (i.e. air toxic

measurements). Groupings were determined based on approximately unbiased test pvalue, with an alpha of 0.95 or higher (235).

We performed an AIC analysis on several key air toxics to determine if a transformation was required. These air toxics were selected due to known toxicities (e.g. lead and mercury) or that were observed to have high magnitudes of risk in a preliminary examination, where they were transformed first by winsorizing to the 99th percentile and then dividing by their interquartile range. We calculated AIC based on the following transformations, along with their untransformed values: natural log, log base 2, log base 10, squared, root, winsorizing to the 99th percentile and dividing by interquartile range, and categorized based on quantiles (Table 4.8). Based on the AIC observed, we elected to transform all air toxics by winsorizing to the 99th percentile and dividing by their interquartile range.

Statistical Analysis

A total of 621 individuals were in the initial study populations. We excluded persons with 1 or no recorded language scores (n=89) or who were missing all air toxic measurements (n=74), yielding a sample size of 471 individuals (Table 4.1). The most frequently missing variables were expressive and receptive language scores at 24 months: 38.9% and 39.1% missing, respectively (Table 4.7). Other missing covariates include expressive and receptive Mullen scores at other time points (14.0% for 6 months, 5.9% for 12 months, and 6.2% and 6.4% for 36 months), maternal educational attainment (0.6%), maternal race (1.3%), and homeownership (4.3%). To include persons missing covariates or Mullen scores, we used multiple imputation by chained equations using the R package *mice* to impute missing values for variables listed above,

including Mullen scores (178). Imputed distribution of these variables can be seen in Table 4.7.

We used a latent class growth analysis (LCGA) approach to analyze the relationship between individual air toxics and language development trajectories, separately modeling expressive vs. receptive language. LCGA is a method to identify unmeasured (or latent) class membership using observed variables, with the goal of creating classes (or in our case, trajectories) so that individuals within a trajectory class are more similar than individuals between trajectory classes (179,180).

We assigned language trajectories to individuals via an LCGA with the expressive or receptive language raw scores as the inputs, and in a separate model regress these on air toxic exposures. We ran several LCGA models with different numbers of classes, ranging from 2 to 7 classes. While the following measures of fit were available to us (Bayesian information criteria (BIC), sample size adjusted BIC (SABIC), integrated complete likelihood (ICL), and entropy), we prioritized low BIC when selecting the best model, following recommendations by Diallo et al. (181). Additionally, we favored models with fewer classes to aid in interpretability and avoided models with low number of individuals within classes (<10% of the population). After individuals were assigned, a multinomial logistic regression was performed using the assigned classes as the outcome and air toxic exposure as our main predictor. We adjusted all models for cohort (EARLI vs. MARBLES) and also accounted for the following a priori potential predictors of language: child sex, maternal age, maternal race/ethnicity, maternal education and home-ownership (as proxies for socioeconomic status) (27–30).

We assessed the coefficients and confidence intervals for each air toxic as a predictor of language trajectory class. Following each process, each person was assigned a probability of belonging to each trajectory class. For tables and figures, we assigned each person to their highest probability class. We selected class names based on the overall shape of the class in the plot, guided by designations used in prior literature. Because of the high number of statistical tests run, we also adjusted for multiple comparisons by using the Benjamini and Hochberg false discovery rate (236). The results of this adjustment can be found on Table 4.3.

In order to assess the potential impact of air toxic mixtures, we conducted a Bayesian kernel machine regression (BKMR). Briefly, BKMR utilizes a kernel function to represent the exposure-outcome response, which allows the model to make use of a large number of potential exposures (237,238). Because BKMR requires a binary outcome, if more than 2 trajectory classes are found to be ideal, then the "non-typically developing" trajectories will be combined to form a binary outcome for analysis. We then examined whether any interaction terms between exposures were statistically significant, along with the overall impact of the mixture model. Due to the high correlation between certain air toxics, we elected to use hierarchical variable selection, grouping air toxics that had a Spearman correlation of 0.6 or greater together prior to analysis (Figure 4.4).

All analyses were conducted using Rstudio version 2022.02.1 Build 461 with R Version 4.1.3 "One Push-Up". The LCGA was conducted using the *lcmm* package (182).

<u>Results</u>

Males had slightly lower MSEL raw scores than females at 36 months of age (Table 4.1). Children whose mothers were 30 or younger or who did not receive at least a college degree had lower MSEL raw scores than children who had mothers older than 30 or who received at least a college degree. Children of African American or Hispanic mothers also had lower MSEL raw scores than children from non-Hispanic white or multiracial mothers. Children whose parents rented had lower MSEL scores than children of homeowners. There was no appreciable difference in MSEL scores between birth seasons.

Hierarchical clustering denoted 7 total groupings of air toxics (Figure 4.3). These groupings are likely due to air toxics within the group sharing common sources, or that may act on the body using similar or identical pathophysiologies. It is also probable that some of the groups are due to those members belonging to the same class of chemicals (e.g. volatile organic compounds).

For expressive language, 3 trajectory classes were optimal for capturing patterns in language development. These trajectory classes were also named High Growth, Tracking, and Low Growth (Figure 4.1). We used High Growth as the referent class for our multinomial regression. In general, we observed that associations of many air toxics with expressive language trajectories were not statistically significant (Table 4.2). However, lead, naphthalene, and acetaldehyde were statistically significant, decreasing risk of falling into the Low Growth trajectory class for every IQR increase in concentration.

For receptive language, 3 trajectory classes were optimal for capturing patterns in language development, which we named High Growth, Tracking, and Low Growth
(Figure 4.2). We used High Growth as the referent class for our multinomial regression. We observed that none of the air toxics examined were not statistically significant (Table 4.2).

The posterior probabilities of inclusion for air toxics groups (Group PIP) and each individual air toxic (Conditional PIP) can be found in Tables 4.4 and 4.5, which indicate the relative importance of the air toxic, both within their defined groups and individually, to the mixture model. Our BKMR analysis indicates that there is a slight increase in overall risk of falling into a non-typically developing trajectory for both expressive and receptive language as overall air toxic concentration increases, though there is a drop in risk at higher quantiles (Figures 4.5 and 4.8). However, this increase in risk was not statistically significant. In addition, for both expressive and receptive language, interactions between air toxics do not appear to be statistically significant (Figures 4.7 and 4.10), although there is an indication that a general direction may be present for some interactions (Figures 4.6 and 4.9).

Discussion

The goal of this study was to characterize language development trajectories in two cohorts of children with enhanced risk for autism and evaluate if prenatal air toxic exposure was associated with delayed or abnormal language development. Results indicated that most air toxics were not associated with impaired language trajectories. Prior to adjusting for multiple comparisons, lead, naphthalene, and acetaldehyde were statistically significant for expressive language, leading to a decrease in risk of falling into the Low Growth trajectory class for each IQR increase. Following adjustment for multiple testing, only acetaldehyde remained statistically significant.

Acetaldehyde is a volatile organic compound (VOC) that's commonly used as a solvent in industrial applications. Our results indicated that acetaldehyde had a protective effect on language development; that is, higher levels of prenatal exposure led to a decreased risk of belonging to an abnormal language trajectory. Prior literature has found mixed results regarding exposure to airborne acetaldehyde and its effect on neurodevelopment. While von Ehrenstein et al. found that exposure increased the odds for ASD, Kalkbrenner et al. did not find this result (24,160). The results from Kalkbrenner et al. showed decreased odds for ASD, as well as slightly higher scores on the Social Responsiveness Scales, though results were not statistically significant (24). Further, Madaniyazi et al. also found that acetaldehyde did not significantly impact various measures of neurodevelopment as reported by the Ages and Stages Questionnaire, which included a measure of language (239).

Lead was found to be protective in our results; that is, higher exposure to lead was associated with a decreased risk of belonging to an abnormal trajectory class for both expressive and receptive language. While these results were not statistically significant when adjusted for multiple testing, it nonetheless is inconsistent with prior studies that show that lead is neurotoxic. Several prior studies have also shown that higher levels of lead exposure were associated with worse language development. Hou et al. found that blood lead levels were associated with worse neurodevelopmental outcomes in children, including language development as measured by the Gesell Development Schedules (240). Lin et al. found that prenatal exposure to lead was associated with lower language quotients as measured by the Comprehensive Developmental Inventory for Infants and Toddlers within 2 year olds (241). Campbell et

al. found that higher bone lead levels in male teenagers from 11 to 14 years old decreased language processing performance, lending evidence that lead exposure continues to negatively impact language past childhood into adolescence (242).

One of the groups created via hierarchical clustering included the toxics toluene, xylene, ethylbenzene, and hexane. These toxics fall under a class of chemicals known as volatile organic compounds, or VOCs. All toxics within this group were shown to have protective effects for both expressive and receptive language, although the effects were not statistically significant; that is, an increase in prenatal exposure led to a decreased risk of falling into an abnormal language trajectory for all four of the toxics. This result is contradictory to prior literature that has shown xylene, toluene, and ethylbenzene to be neurotoxic in nature. Madaniyazi et al. found that increases in the concentration of certain forms of xylene were associated with lower scores in the Ages and Stages Questionnaire, while Grandjean and Landrigan found in a review that toluene exposure can lead to deficits in neurodevelopment (239,243). Additionally, Von Ehrenstein et al. found that ethylbenzene, toluene, and xylenes were associated with a greater risk for autism spectrum disorder (160). Kalkbrenner et al. also found that ethylbenzene, hexane, toluene, and xylenes all increased the risk of autism, though only ethylbenzene and xylenes were statistically significant. However, our results show that all four toxics had protective effects, although our results were not statistically significant. This may be due to random chance due to multiple testing, or it may be due to the relatively small sample size, which led to comparatively smaller populations that fall into each of the trajectory classes.

Multiple testing is of particular concern in this study, due to the large number of air toxics examined; in total, we examined 53 air toxics over two different measures of language across three classes. In order to help address the problem of multiple comparisons, we adjusted p-values using the Benjamini and Hochberg false discovery rate (236). After adjusting for multiple comparisons, only acetaldehyde remained statistically significant for expressive language. In addition, some of the groups had a small number of individuals, despite our efforts to choose classes that had greater than 10 percent of the study population. Because of these two factors, it is possible that we found false negatives in our analysis; that is, the fact that we did not find many statistically significant associations does not mean that air toxics are not without risk.

Most prior studies that examined airborne air toxics and neurodevelopment found that some toxics were linked to measures of neurodevelopment. However, in our study, after adjusting for multiple comparisons, we only found that acetaldehyde was statistically significant. Furthermore, we found that acetaldehyde decreased the risk of falling into an abnormal language trajectory, which runs counter to our hypothesis that exposure to air toxics would increase risk of neurodevelopmental delays or disorder. While our smaller sample size may explain some of this discrepancies, there may be further reasons for our disparate results. Because we used NATA to collect exposure data, we may not be capturing all sources of airborne exposure to these air toxics, since NATA focuses mostly on outdoor sources of air pollution. As such, our exposure may be underrepresenting the true exposure for some air toxics. It is also possible that our findings were influenced by the uniqueness of our sample, where children were at enhanced risk for autism.

We examined the potential effects of air toxic mixtures on language development trajectories by utilizing BKMR analysis. Our results found that there was an increase in risk in a mixture model as concentrations increased from the 50th quantile up until the 60th quantile, though risk then decreased past the 65th percentile. This change in risk was also not statistically significant. In addition, when we examined interactions between any single air toxic and all other air toxics, we found that there were no statistically significant interactions, though some trends could be seen with some air toxics. It should be noted that our sample size is relatively small, which limits our ability to assess interactions, even within the BKMR framework.

Strengths of our study include the use of language development trajectories as our outcome, which allows for a more detailed and accurate examination of how air toxic affects language development when compared to a single time point outcome. In addition, because we made use of repeated measures for our outcome in order to develop these trajectories, there is a decreased risk of exposure misclassification, as these measures would decrease the probability of incorrect assignment of trajectories. Second, the use of cohorts with heightened familial risk of autism allows for a greater ability to detect abnormal language development trajectories that may be less common in the general population, as these cohorts are at a heightened prevalence for neurodevelopmental delays.

Our results are not applicable to the general population, as both cohorts used were at heightened risk of autism. In addition, while the MSEL is considered to be a suitable psychometric measure of expressive and receptive language (185) and has been shown to have good validity (186), language is a complex phenotype, with multiple

facets to consider. Our analysis is limited to expressive and receptive language only, and only through the methodology of the MSEL. Future research should explore other facets of language, such as vocabulary, grammatical development, or semantics. We are also unable to extrapolate language trajectories after 36 months of age. Future studies should aim to examine language development over a longer period of time, particularly because language can develop rapidly within the first few years of life. Due to the nature of LCGA analysis, our results are sensitive to the number of classes selected. While we followed recommendations outlined in Diallo et al., we may have selected for a non-ideal number of classes, which may have resulted in different results (181). However, we are confident that the overall direction of effect would remain consistent between various class choices. Finally, our study focused on airborne exposure of air toxics, but it is not the only route of exposure for many of the toxics examined. Our study can't account for those additional routes of exposure, nor did it account for the actual body burden the mother experienced. In addition, because NATA is not published every year, we used datasets closest to each child's birth year, which may not reflect the actual concentration of air toxics absorbed during the entire prenatal period.

Tables and Figures

Table 4.1: Mean and Standard Deviation of 36 Month Mullen Scores for Included Population by Child Characteristics

	Expressive language raw score (36 months)	Receptive language raw score (36 months)
Child Sex		
Female (N=200)	32.2 (5.25)	31.6 (4.63)
Male (N=271)	30.2 (6.84)	29.7 (5.90)
Maternal age		
20 to 30 years (N=92)	29.5 (6.81)	29.3 (5.19)
31 to 35 years (N=186)	31.4 (6.03)	30.6 (5.62)
36 to 40 years (N=140)	31.2 (6.12)	31.0 (5.47)
40 to 49 years (N=53)	32.0 (6.44)	31.0 (5.30)
Maternal Race/Ethnicity		

Non-Hispanic White (N=262)	31.7 (5.92)	31.0 (5.30)
Black/African American (N=31)	29.8 (6.28)	29.3 (3.54)
Hispanic (N=90)	29.9 (6.60)	28.7 (6.24)
Other/Multiracial (N=88)	30.7 (6.88)	31.3 (5.36)
Maternal Educational Attainment		
High School/GED or Less (N=43)	28.8 (6.86)	28.0 (4.96)
Some college (N=170)	30.1 (6.28)	29.4 (5.53)
Bachelor's degree (N=149)	32.4 (5.65)	31.4 (5.19)
Graduate or Professional degree (N=109)	31.6 (6.48)	31.9 (5.34)
Homeownership		
Rent (N=178)	30.3 (7.12)	29.6 (6.31)
Own (N=293)	31.5 (5.69)	31.1 (4.83)

Birth Season

Spring (n=103)	31.1 (5.84)	30.3 (5.03)
Summer (n=141)	30.7 (6.57)	30.0 (5.65)
Fall (n=120)	31.5 (6.47)	31.0 (5.82)
Winter (n=107)	31.0 (6.19)	30.8 (5.27)
Overall (N=471)	31.0 (6.29)	30.5 (5.48)

Footnotes for Table 4.1: Mean and standard deviations calculated after removing missing values

Air Toxic Expressive Language **Receptive Language** Tracking Low Growth Tracking Low Growth Group 1 Cresol/Cresylic Acid 1.00 (0.78, 1.28) 1.00 (0.70, 1.42) 1.00 (0.67, 1.49) 1.00 (0.64, 1.56) 1.00 (0.70, 1.42) 1.00 (0.67, 1.49) 1.00 (0.78, 1.28) 1.00 (0.64, 1.56) Phenol Group 2 DEHP 0.97 (0.76, 1.25) 0.96 (0.67, 1.38) 0.99 (0.67, 1.48) 0.97 (0.62, 1.52) Carbon Disulfide 0.99 (0.77, 1.27) 0.98 (0.68, 1.41) 1.00 (0.67, 1.49) 0.99 (0.63, 1.55) Group 3 **Xylenes** 0.76 (0.53, 1.10) 0.97 (0.65, 1.43) 0.69 (0.44, 1.08) 0.90 (0.71, 1.16) Toluene 0.74 (0.52, 1.07) 0.88 (0.69, 1.12) 0.99 (0.67, 1.47) 0.76 (0.49, 1.19) Ethylbenzene 0.85 (0.67, 1.09) 0.77 (0.54, 1.11) 1.01 (0.68, 1.50) 0.79 (0.50, 1.23) Hexane 0.90 (0.71, 1.16) 0.76 (0.53, 1.10) 0.97 (0.65, 1.43) 0.69 (0.44, 1.08) Group 4 1.01 (0.68, 1.50) Dibenzofuran 1.25 (0.97, 1.60) 0.92 (0.65, 1.30) 1.13 (0.72, 1.76) **Polychlorinated Biphenyls** 1.01 (0.79, 1.29) 0.99 (0.70, 1.42) 0.98 (0.66, 1.47) 1.01 (0.65, 1.58) Group 5 Hexachlorobenzene 1.00 (0.78, 1.28) 1.01 (0.71, 1.44) 1.00 (0.67, 1.49) 1.00 (0.64, 1.56) 1.20 (0.94, 1.53) 1.24 (0.86, 1.77) 0.91 (0.61, 1.36) 0.97 (0.62, 1.52) Trichlorobenzene Group 6 2,4,6-Trichlorophenol 0.92 (0.72, 1.18) 0.91 (0.64, 1.29) 0.79 (0.53, 1.19) 0.99 (0.63, 1.56) 0.97 (0.76, 1.25) 1.03 (0.73, 1.47) 0.88 (0.59, 1.31) 1.00 (0.64, 1.55) Trifluralin Arsenic 1.10 (0.86, 1.40) 1.05 (0.74, 1.50) 0.90 (0.60, 1.35) 0.96 (0.61, 1.50) Carbaryl 0.96 (0.75, 1.23) 0.92 (0.64, 1.31) 1.03 (0.66, 1.61) 1.08 (0.72, 1.60) 2,4-Dichlorophenoxyacetic Acid 0.90 (0.71, 1.15) 1.00 (0.67, 1.49) 0.97 (0.68, 1.38) 0.87 (0.56, 1.36) Salts And Esters Group 7 Manganese 1.00 (0.78, 1.28) 1.15 (0.80, 1.64) 0.91 (0.61, 1.36) 1.02 (0.65, 1.59) Nickel 0.98 (0.77, 1.26) 0.98 (0.69, 1.40) 1.03 (0.69, 1.53) 0.93 (0.59, 1.45) **Dimethyl Phthalate** 1.09 (0.85, 1.40) 1.08 (0.76, 1.54) 1.01 (0.68, 1.52) 1.13 (0.73, 1.77)

Table 4.2: Relative Risk Ratios for each IQR increase of Air Toxics

Acrylonitrile	1 07 (0 84 1 37)	1 01 (0 71 1 44)	1 44 (0 95 2 20)	1 40 (0 88 2 23)
Vipul Chlorido	(0.0-, 1.07)	$0.77 (0.54 \ 1.11)$	0.05 (0.64, 1.42)	0.82 (0.53 1.20)
Methyl Ledide	0.99(0.70, 1.27)	0.77 (0.54, 1.11)	0.95 (0.04, 1.42)	0.03 (0.03, 1.00)
	1.04 (0.01, 1.32)	0.94 (0.00, 1.34)	1.03 (0.09, 1.04)	0.99 (0.64, 1.55)
	1.01 (0.79, 1.30)	0.95 (0.67, 1.36)	1.03 (0.70, 1.54)	1.03 (0.66, 1.61)
Ethylene Dibromide	1.09 (0.85, 1.39)	1.09 (0.77, 1.56)	1.05 (0.70, 1.57)	1.11 (0.71, 1.74)
Mercury	1.16 (0.91, 1.49)	1.07 (0.75, 1.53)	1.10 (0.73, 1.64)	1.08 (0.69, 1.70)
Methyl Chloride	1.01 (0.79, 1.29)	1.01 (0.71, 1.45)	1.01 (0.68, 1.50)	1.01 (0.65, 1.58)
Nitrophenol	1.01 (0.79, 1.29)	0.96 (0.67, 1.37)	1.31 (0.88, 1.96)	1.40 (0.89, 2.19)
No Group				
Acetaldehyde	0.91 (0.71, 1.16)	0.53 (0.36, 0.77)	0.81 (0.55, 1.20)	0.71 (0.46, 1.10)
Acrolein	1.08 (0.84, 1.37)	0.70 (0.48, 1.02)	1.06 (0.71, 1.58)	0.93 (0.59, 1.45)
Acrylamide	1.00 (0.78, 1.27)	0.96 (0.67, 1.37)	0.99 (0.66, 1.48)	0.97 (0.62, 1.52)
Chlorobenzene	1.01 (0.79, 1.29)	0.99 (0.70, 1.41)	0.99 (0.66, 1.47)	0.99 (0.63, 1.55)
Dibutyl phthalate	1.11 (0.87, 1.42)	0.98 (0.68, 1.41)	1.41 (0.91, 2.17)	1.37 (0.85, 2.21)
Dichlorobenzene	1.01 (0.79, 1.29)	0.75 (0.52, 1.09)	1.24 (0.82, 1.87)	1.01 (0.64, 1.61)
Dichlorvos	0.99 (0.78, 1.27)	0.98 (0.68, 1.39)	1.02 (0.68, 1.53)	1.01 (0.64, 1.58)
Dimethyl Formamide	0.92 (0.72, 1.18)	0.89 (0.62, 1.27)	1.22 (0.81, 1.85)	1.04 (0.66, 1.64)
Ethylene Oxide	0.97 (0.76, 1.24)	0.99 (0.70, 1.41)	1.01 (0.67, 1.50)	0.99 (0.63, 1.55)
Hydrazine	1.00 (0.78, 1.28)	1.01 (0.71, 1.44)	0.98 (0.66, 1.46)	0.98 (0.63, 1.53)
Methyl Bromide	1.01 (0.79, 1.29)	0.96 (0.67, 1.37)	0.95 (0.64, 1.42)	0.99 (0.63, 1.54)
Methyl Tertbutyl Ether	1.10 (0.86, 1.41)	1.04 (0.73, 1.49)	0.97 (0.65, 1.44)	0.92 (0.59, 1.43)
Naphthalene	0.97 (0.76, 1.23)	0.60 (0.42, 0.88)	0.99 (0.67, 1.46)	0.69 (0.44, 1.08)
Nitrobenzene	1.00 (0.78, 1.28)	1.00 (0.70, 1.42)	1.00 (0.67, 1.50)	1.00 (0.64, 1.56)
Propylene Dichloride	1.21 (0.95, 1.55)	0.98 (0.69, 1.39)	0.84 (0.56, 1.26)	0.96 (0.61, 1.50)
Styrene	0.93 (0.73, 1.19)	0.76 (0.53, 1.09)	0.97 (0.65, 1.45)	0.79 (0.51, 1.24)
Trichloroethylene	0.98 (0.77, 1.25)	0.78 (0.54, 1.12)	1.29 (0.85, 1.96)	0.91 (0.57, 1.45)
Triethylamine	0.95 (0.74, 1.21)	0.77 (0.53, 1.11)	0.87 (0.59, 1.28)	0.69 (0.45, 1.08)
Vinyl Acetate	1.01 (0.79, 1.29)	0.79 (0.55, 1.13)	0.92 (0.62, 1.38)	0.93 (0.59, 1.45)
Vinylidene Chloride	1.02 (0.79, 1.30)	1.02 (0.72, 1.46)	1.10 (0.72, 1.67)	1.12 (0.70, 1.78)
Cobalt	0.92 (0.72, 1.18)	0.97 (0.68, 1.39)	0.99 (0.67, 1.48)	0.91 (0.58, 1.43)
Cyanide	1.01 (0.79, 1.29)	1.04 (0.73, 1.49)	1.02 (0.68, 1.52)	1.01 (0.65, 1.57)

Lead	0.95 (0.74, 1.22)	0.64 (0.44, 0.92)	0.84 (0.56, 1.25)	0.77 (0.50, 1.21)
Selenium	0.86 (0.67, 1.10)	0.76 (0.53, 1.09)	1.02 (0.68, 1.51)	0.81 (0.51, 1.26)
Tetrachloroethylene	0.95 (0.75, 1.21)	0.84 (0.58, 1.22)	1.05 (0.70, 1.57)	0.97 (0.62, 1.53)

Footnotes for Table 4.2: All models were adjusted for child sex, maternal age, maternal race/ethnicity, maternal education and homeownership; each relative risk is per IQR increase for each air toxic; High Growth was used as the referent group for both models

Air Toxic		Expressive	Language		Receptive Language			
	Trac	king	Low G	rowth	Tracl	king	Low G	rowth
	Non- corrected	Corrected	Non- corrected	Corrected	Non- corrected	Corrected	Non- corrected	Corrected
Group 1								
Cresol/Cresylic Acid	0.998	0.998	0.995	0.995	0.998	0.998	0.998	0.998
Phenol	0.998	0.998	0.995	0.995	0.998	0.998	0.998	0.998
Group 2								
DEHP	0.836	0.998	0.832	0.995	0.973	0.998	0.904	0.998
Carbon Disulfide	0.924	0.998	0.928	0.995	0.994	0.998	0.962	0.998
Group 3								
Xylenes	0.423	0.998	0.144	0.678	0.864	0.998	0.106	0.998
Toluene	0.294	0.998	0.108	0.678	0.954	0.998	0.227	0.998
Ethylbenzene	0.562	0.998	0.181	0.678	0.567	0.998	0.239	0.998
Hexane	0.201	0.998	0.163	0.678	0.961	0.998	0.29	0.998
Group 4								
Dibenzofuran	0.079	0.998	0.627	0.995	0.968	0.998	0.594	0.998
Polychlorinated Biphenyls	0.936	0.998	0.974	0.995	0.935	0.998	0.97	0.998
Group 5								
Hexachlorobenzene	0.982	0.998	0.97	0.995	0.997	0.998	0.994	0.998
Trichlorobenzene	0.151	0.998	0.245	0.811	0.641	0.998	0.885	0.998
Group 6								
2,4,6-Trichlorophenol	0.512	0.998	0.586	0.995	0.259	0.998	0.98	0.998
Trifluralin	0.836	0.998	0.851	0.995	0.54	0.998	0.995	0.998
Arsenic	0.468	0.998	0.78	0.995	0.615	0.998	0.854	0.998
Carbaryl	0.758	0.998	0.635	0.995	0.721	0.998	0.895	0.998
2,4-Dichlorophenoxyacetic Acid Salts And Esters	0.409	0.998	0.852	0.995	0.996	0.998	0.553	0.998
Group 7								

Table 4.3: False Discovery Rate Adjusted P-Values

Manganese	0.988	0.998	0.451	0.995	0.662	0.998	0.928	0.998
Nickel	0.892	0.998	0.927	0.995	0.89	0.998	0.738	0.998
Dimethyl Phthalate	0.48	0.998	0.662	0.995	0.944	0.998	0.584	0.998
Acrylonitrile	0.571	0.998	0.961	0.995	0.089	0.998	0.154	0.998
Vinyl Chloride	0.95	0.998	0.158	0.678	0.819	0.998	0.414	0.998
Methyl lodide	0.784	0.998	0.744	0.995	0.87	0.998	0.97	0.998
Aniline	0.907	0.998	0.797	0.995	0.867	0.998	0.882	0.998
Ethylene Dibromide	0.497	0.998	0.622	0.995	0.806	0.998	0.635	0.998
Mercury	0.228	0.998	0.71	0.995	0.649	0.998	0.722	0.998
Methyl Chloride	0.938	0.998	0.94	0.995	0.966	0.998	0.955	0.998
Nitrophenol	0.928	0.998	0.831	0.995	0.183	0.998	0.142	0.998
No Group								
Acetaldehyde	0.452	0.998	0.001	0.044	0.295	0.998	0.126	0.998
Acrolein	0.55	0.998	0.061	0.678	0.792	0.998	0.743	0.998
Acrylamide	0.97	0.998	0.815	0.995	0.953	0.998	0.897	0.998
Chlorobenzene	0.941	0.998	0.953	0.995	0.944	0.998	0.969	0.998
Dibutyl Phthalate	0.405	0.998	0.917	0.995	0.125	0.998	0.196	0.998
Dichlorobenzene	0.918	0.998	0.132	0.678	0.315	0.998	0.951	0.998
Dichlorvos	0.949	0.998	0.89	0.995	0.912	0.998	0.971	0.998
Dimethyl Formamide	0.522	0.998	0.51	0.995	0.34	0.998	0.872	0.998
Ethylene Oxide	0.79	0.998	0.975	0.995	0.978	0.998	0.964	0.998
Hydrazine	0.984	0.998	0.962	0.995	0.928	0.998	0.94	0.998
Methyl Bromide	0.949	0.998	0.82	0.995	0.796	0.998	0.963	0.998
Methyl Tertbutyl Ether	0.442	0.998	0.819	0.995	0.874	0.998	0.711	0.998
Naphthalene	0.776	0.998	0.009	0.236	0.95	0.998	0.102	0.998
Nitrobenzene	0.996	0.998	0.993	0.995	0.996	0.998	0.994	0.998
Propylene Dichloride	0.127	0.998	0.898	0.995	0.396	0.998	0.843	0.998
Styrene	0.588	0.998	0.135	0.678	0.887	0.998	0.306	0.998
Trichloroethylene	0.86	0.998	0.173	0.678	0.223	0.998	0.687	0.998
Triethylamine	0.684	0.998	0.161	0.678	0.482	0.998	0.104	0.998
Vinyl Acetate	0.924	0.998	0.192	0.678	0.691	0.998	0.733	0.998

Vinylidene Chloride	0.895	0.998	0.895	0.995	0.665	0.998	0.639	0.998
Cobalt	0.531	0.998	0.887	0.995	0.977	0.998	0.692	0.998
Cyanide	0.944	0.998	0.828	0.995	0.933	0.998	0.968	0.998
Lead	0.686	0.998	0.015	0.262	0.379	0.998	0.261	0.998
Selenium	0.243	0.998	0.135	0.678	0.941	0.998	0.348	0.998
Tetrachloroethylene	0.686	0.998	0.372	0.995	0.816	0.998	0.895	0.998

Footnotes for Table 4.3: An FDR of 0.1 was used

Conditional Group Air Toxic PIP PIP Group 1 Acetaldehyde 0.47 0.32 Acrylamide 0.47 0 Hydrazine 0.47 0.28 Methyl Tertbutyl Ether 0.47 0.40 Group 2 Acrolein 0.59 0.04 **Dibutyl Phthalate** 0.59 0.003 Dichlorobenzene 0.59 0.18 **Dimethyl Formamide** 0.59 0.10 Hexachlorobenzene 0.59 0.42 Naphthalene 0.59 0.24 Trichloroethylene 0.59 0.003 Vinyl Acetate 0.59 0.01 Tetrachloroethylene 0.59 0 Group 3 Acrylonitrile 0.66 0.61 Vinyl Chloride 0.66 0.39 Group 4 DEHP 0.42 0.20 Ethylbenzene 0.42 0.04 **Xylenes** 0.42 0.09 Hexane 0.42 0.27 Styrene 0.42 0 Toluene 0.42 0 Triethylamine 0.42 0.01 Cobalt 0.42 0.21

Table 4.4: Posterior Inclusion Probabilities for Expressive Language

Selenium	0.42	0.19
Group 5		
Chlorobenzene	0.48	0
Dimethyl Phthalate	0.48	0
Nitrophenol	0.48	0.01
Arsenic	0.48	0.66
Manganese	0.48	0.26
Nickel	0.48	0.07
Group 6		
Cresol/Cresylic Acid	0.74	0.42
Phenol	0.74	0.58
Group 7		
Methyl Bromide	0.89	0.55
Trifluralin	0.89	0.45
Group 8		
Ethylene Dibromide	0.79	0.53
Propylene Dichloride	0.79	0.47
No Group		
Aniline	0.80	1
2,4,6-Trichlorophenol	0.68	1
Carbaryl	1	1
Carbon Disulfide	1	1
2,4-Dichlorophenoxyacetic	0.22	1
Acid Sails And Esters	0.49	1
Diberizorurari	0.49	1
Dictilor Vos		1
	0.00	1
Methyl Ledide	1	1
	1	1
INITIODENZEN3	1	1
Polychiorinated Biphenyls	0.92	1

Trichlorobenzene	1	1
Vinylidene Chloride	0.79	1
Cyanide	0.93	1
Lead	0.58	1
Mercury	1	1

Footnotes for Table 4.4: All models were adjusted for child sex, maternal age, maternal race/ethnicity, maternal education and homeownership; Tracking and Low Growth classes were combined into a "non-optimal" class for comparison

Conditional Group Air Toxic PIP PIP Group 1 Acetaldehyde 0.88 0.08 Acrylamide 0.82 0.76 Hydrazine 0.88 0.15 Methyl Tertbutyl Ether 0.88 0.01 Group 2 Acrolein 0.86 0 **Dibutyl Phthalate** 0.86 0.12 Dichlorobenzene 0.86 0.02 **Dimethyl Formamide** 0.86 0.18 Hexachlorobenzene 0.08 0.86 Naphthalene 0.86 0.05 Trichloroethylene 0.86 0.20 Vinyl Acetate 0.86 0.14 Tetrachloroethylene 0.21 0.86 Group 3 Acrylonitrile 0.69 1 Vinyl Chloride 0.31 1 Group 4 DEHP 0.72 0.12 Ethylbenzene 0.72 0.13 **Xylenes** 0.72 0 Hexane 0.72 0.22 Styrene 0.72 0.02 Toluene 0.72 0.25 Triethylamine 0 0.72 Cobalt 0.72 0.17

Table 4.5: Posterior Inclusion Probabilities for Receptive Language

Selenium	0.72	0.09
Group 5		
Chlorobenzene	1	0.23
Dimethyl Phthalate	1	0.03
Nitrophenol	1	0.05
Arsenic	1	0.28
Manganese	1	0.04
Nickel	1	0.37
Group 6		
Cresol/Cresylic Acid	1	0.52
Phenol	1	0.48
Group 7		
Methyl Bromide	1	0.62
Trifluralin	1	0.38
Group 8		
Ethylene Dibromide	1	0.25
Propylene Dichloride	1	0.75
No Group		
Aniline	1	1
2,4,6-Trichlorophenol	1	1
Carbaryl	1	1
Carbon Disulfide	0.84	1
2,4-Dichlorophenoxyacetic		
Acid Salts And Esters	1	1
Dibenzofuran	1	1
Dichlorvos	0.85	1
Ethylene Oxide	0.96	1
Methyl Chloride	1	1
Methyl Iodide	0.81	1
Nitrobenzene	1	1
Polychlorinated Biphenyls	1	1

Trichlorobenzene	0.51	1
Vinylidene Chloride	0.98	1
Cyanide	1	1
Lead	1	1
Mercury	1	1

Footnotes for Table 4.5: All models were adjusted for child sex, maternal age, maternal race/ethnicity, maternal education and homeownership; Tracking and Low Growth classes were combined into a "non-optimal" class for comparison

Acetaldehyde (244)	o-Cresol (245)	Xylenes (isomers and mixture) (246,247)	Methyl tert butyl ether (248)
Acrolein (249,250)	m-Cresol (245)	Ethylene oxide (251)	4,4'-Methylenedianiline (252)
Acrylamide (253)	p-Cresol (245)	Ethylene thiourea (254)	Naphthalene (255)
Acrylonitrile (256)	2,4-D, salts and esters (257)	Heptachlor (258,259)	Nitrobenzene (260)
4-Aminobiphenyl (261)	DDE (262)	Hexachlorobenzene (263)	4-Nitrophenol (264,265)
Aniline (266)	Dibenzofurans (267,268)	Hexachlorocyclopentadiene (269)	Ethylene dibromide (Dibromoethane) (270)
2,4,6-Trichlorophenol (271)	1,2-Dibromo-3-chloropropane (272)	Hexachloroethane (273)	N-Nitrosomorpholine (274)
Bis(2-ethylhexyl)phthalate (DEHP) (275)	Dibutylphthalate (276,277)	Hexane (278,279)	Parathion (280)
Carbaryl (281)	1,4-Dichlorobenzene(p) (282)	Hydrazine (283,284)	Pentachloronitrobenzene (Quintobenzene) (285)
Carbon disulfide (286)	Dichlorvos (287)	Lindane (all isomers) (288)	Pentachlorophenol (289)
Chlordane (288,290)	Dimethyl formamide (291)	Methoxychlor (292,293)	Phenol (294,295)
Chlorobenzene (263)	Dimethyl phthalate (77,296)	Methyl bromide (Bromomethane) (297)	Polychlorinated biphenyls (Aroclors) (123,143,267)
Chlorobenzilate (298,299)	2,4-Dinitrophenol (300,301)	Methyl chloride (Chloromethane) (302)	Propoxur (Baygon) (303)
Cresols/Cresylic acid (isomers and mixture) (245)	Ethyl benzene (246,247)	Methyl iodide (lodomethane) (304)	Propylene dichloride (1,2- Dichloropropane) (305)
o-Xylenes (246,247)	Styrene (306)	Arsenic Compounds (inorganic including arsine) (307)	2,4-Toluene diamine (308)
m-Xylenes (246,247)	2,3,7,8-Tetrachlorodibenzo-p- dioxin (309)	Chromium Compounds (310)	Toxaphene (chlorinated camphene) (311)
p-Xylenes (246,247)	Toluene (312)	Cobalt Compounds (313)	1,2,4-Trichlorobenzene (314,315)
Cyanide Compounds (316)	Trichloroethylene (317)	Mercury Compounds (318)	Trifluralin (319)
Lead Compounds (320)	2,4,5-Trichlorophenol (321)	Nickel Compounds (322)	Vinyl acetate (323)
Manganese Compounds (324)	Triethylamine (325)	Selenium Compounds (326)	Vinyl chloride (327)
Tetrachloroethylene	Vinylidene chloride (1,1-	Tetrachloroethylene	
(Perchloroethylene) (328)	Dichloroethylene)	(Perchloroethylene) (328)	

Table 4.6: Air Toxics with Evidence of Endocrine Disrupting Activity

Footnotes for Table 4.6: All toxics have shown potential evidence of endocrine disruption in either animal or epidemiological studies

	Included (N=471)	Excluded (N=150)	Overall (N=621)
Child Sex			
Female	200 (42.5%)	80 (57.6%)	280 (45.9%)
Male	271 (57.5%)	59 (42.4%)	330 (54.1%)
Missing	0	11	11
Maternal Age			
20 to 30 years	92 (19.5%)	34 (23.9%)	102 (16.6%)
31 to 35 years	58 (12.3%)	8 (5.6%)	244 (39.8%)
36 to 40 years	186 (39.5%)	58 (40.9%)	177 (28.9%)
40 to 49 years	135 (28.7%)	42 (29.6%)	66 (10.8%)
Missing	0	8	8
Maternal Race/Ethnicity			
Non-Hispanic White	260 (55.9%)	43 (44.3%)	303 (53.9%)
Black/African American	29 (6.3%)	10 (10.3%)	39 (6.9%)
Hispanic	89 (19.1%)	23 (23.7%)	112 (19.9%)
Other/Multiracial	87 (18.7%)	21 (21.7%)	108 (19.3%)
Missing	6	53	59

Table 4.7: Comparison of characteristics of included versus excluded individuals

	Included (N=471)	Excluded (N=150)	Overall (N=621)
Maternal Educational Attainment			
High School/GED or less	148 (31.6%)	23 (25.3%)	36 (6.4%)
Some college	109 (23.3%)	24 (26.4%)	201 (36%)
Bachelor's degree	43 (9.2%)	11 (12.1%)	171 (30.6%)
Graduate or Professional degree	168 (35.9%)	33 (36.2%)	133 (23.8%)
Missing	3	59	62
Homeownership			
Rent	175 (38.5%)	43 (48.3%)	218 (40.1%)
Own	280 (61.5%)	46 (51.7%)	326 (59.9%)
Missing	16	61	77
Expressive language raw score (6 months)			
Mean (SD)	6.26 (1.20)	6.74 (1.65)	6.32 (1.28)
Median [IQR]	6.00 [1.00]	6.00 [1.00]	6.00 [1.00]
Missing	66	89	155
Receptive language raw score (6 months)			
Mean (SD)	7.84 (1.51)	8.62 (1.59)	7.95 (1.55)

	Included (N=471)	Excluded (N=150)	Overall (N=621)
Median [IQR]	8.00 [2.00]	9.00 [2.00]	8.00 [2.00]
Missing	66	89	155
Expressive language raw score (12 months)			
Mean (SD)	11.8 (2.61)	13.1 (2.44)	12.0 (2.62)
Median [IQR]	12.0 [4.00]	13.0 [3.00]	12.0 [4.00]
Missing	28	92	120
Receptive language raw score (12 months)			
Mean (SD)	12.4 (2.11)	12.7 (1.80)	12.5 (2.07)
Median [IQR]	13.0 [3.00]	13.0 [2.00]	13.0 [3.00]
Missing	28	92	120
Expressive language raw score (24 months)			
Mean (SD)	20.6 (4.76)	20.7 (5.58)	20.6 (4.85)
Median [IQR]	21.0 [6.00]	21.0 [6.50]	21.0 [6.00]
Missing	183	115	298
Receptive language raw score			

(24 months)

	Included (N=471)	Excluded (N=150)	Overall (N=621)
Mean (SD)	23.2 (5.47)	22.7 (5.88)	23.2 (5.51)
Median [IQR]	25.0 [6.50]	25.0 [8.50]	25.0 [7.00]
Missing	184	115	299
Expressive language raw score (36 months)			
Mean (SD)	31.1 (6.29)	27.0 (8.22)	31.0 (6.36)
Median [IQR]	32.0 [6.75]	26.0 [11.5]	32.0 [7.00]
Missing	29	139	168
Receptive language raw score (36 months)			
Mean (SD)	30.5 (5.54)	27.1 (8.65)	30.4 (5.65)
Median [IQR]	31.0 [5.00]	28.0 [6.00]	31.0 [5.00]
Missing	30	139	169

Footnote for Table 4.7: Percentages were calculated after removing missing individuals from the sample

	Non-imputed	Imputed
	(N=471)	(N=471)
Child Sex		
Female	200 (42.5%)	200 (42.5%)
Male	271 (57.5%)	271 (57.5%)
Missing	0	
Maternal Age		
20 to 30 years	92 (19.5%)	92 (19.5%)
31 to 35 years	58 (12.3%)	186 (39.5%)
36 to 40 years	186 (39.5%)	140 (29.7%)
40 to 49 years	135 (28.7%)	53 (11.3%)
Missing	0	
Maternal Race/Ethnicity		
Non-Hispanic White	260 (55.9%)	262 (55.6%)
Black/African American	29 (6.3%)	31 (6.6%)
Hispanic	89 (19.1%)	90 (19.1%)
Other/Multiracial	87 (18.7%)	88 (18.7%)
Missing	6	

 Table 4.8: Comparison of Imputed versus Non-Imputed Covariates

	Non-imputed (N=471)	Imputed (N=471)
Maternal Educational Attainment		
High School/GED or less	148 (31.6%)	43 (9.1%)
Some college	109 (23.3%)	170 (36.1%)
Bachelor's degree	43 (9.2%)	149 (31.6%)
Graduate or Professional degree	168 (35.9%)	109 (23.1%)
Missing	3	
Homeownership		
Rent	175 (38.5%)	178 (37.8%)
Own	280 (61.5%)	293 (62.2%)
Missing	16	
Expressive language raw score (6 months)		
Mean (SD)	6.26 (1.20)	6.26 (1.21)
Median [IQR]	6.00 [1.00]	6.00 [1.00]
Missing	66	
Receptive language raw score (6 months)		
Mean (SD)	7.84 (1.51)	7.80 (1.54)

	Non-imputed (N=471)	Imputed (N=471)
Median [IQR]	8.00 [2.00]	8.00 [2.00]
Missing	66	
Expressive language raw score (12 months)		
Mean (SD)	11.8 (2.61)	11.9 (2.58)
Median [IQR]	12.0 [4.00]	12.0 [4.00]
Missing	28	
Receptive language raw score (12 months)		
Mean (SD)	12.4 (2.11)	12.4 (2.10)
Median [IQR]	13.0 [3.00]	13.0 [3.00]
Missing	28	
Expressive language raw score (24 months)		
Mean (SD)	20.6 (4.76)	20.3 (4.79)
Median [IQR]	21.0 [6.00]	20.0 [7.00]
Missing	183	
Receptive language raw score (24 months)		

	Non-imputed (N=471)	Imputed (N=471)
Mean (SD)	23.2 (5.47)	22.9 (5.46)
Median [IQR]	25.0 [6.50]	25.0 [7.00]
Missing	184	
Expressive language raw score (36 months)		
Mean (SD)	31.1 (6.29)	31.0 (6.29)
Median [IQR]	32.0 [6.75]	32.0 [7.00]
Missing	29	
Receptive language raw score (36 months)		
Mean (SD)	30.5 (5.54)	30.5 (5.48)
Median [IQR]	31.0 [5.00]	31.0 [5.00]
Missing	30	

Footnote for Table 4.8: Percentages were calculated after removing missing individuals from the sample

Transformation	Lead	Mercury	Methyl Chloride	Acrylonitrile	Dibutyl Phthalate	Acetaldehyde	Dichlorobenzene	Naphthalene	Xylenes	Hexane
Unaltered	2436.87	2438.52	2439.32	2434.7	2437.89	2440.35	2440.12	2440.23	2439.64	2439.17
Natural log	2433.44	2437.60	2439.13	2436.78	2438.17	2440.15	2439.34	2440.09	2438.13	2437.74
Log base 2	2433.44	2437.60	2439.13	2436.78	2438.17	2440.15	2439.34	2440.09	2438.13	2437.74
Log base 10	2433.44	2437.60	2439.13	2436.78	2438.17	2440.15	2439.34	2440.09	2438.13	2437.74
Squared	2436.56	2434.82	2440.13	2435.85	2439.06	2440.62	2440.55	2437.66	2441.28	2441.10
Root	2434.95	2437.80	2439.22	2435.04	2437.43	2440.26	2439.69	2440.47	2439.05	2438.57
Winsorized and divided by IQR	2435.49	2436.75	2439.11	2434.26	2437.31	2440.30	2439.40	2440.44	2439.83	2439.27
Quantile	2436.65	2433.46	2440.26	2433.00	2438.76	2442.27	2440.92	2441.98	2442.84	2440.70

Table 4.9: Comparison of Transformation of Air Toxics

Footnotes for Table 4.9: AIC values are presented and rounded to the nearest hundredth; lower values indicate better fit



Figure 4.1: Language Development Trajectories from Latent Class Growth Analysis for Expressive Language

Footnotes for Figure 4.1: Points indicate mean scores at that age with 95 percent confidence intervals



Figure 4.2: Language Development Trajectories from Latent Class Growth Analysis for Receptive Language

Footnotes for Figure 4.2: Points indicate mean scores at that age with 95 percent confidence intervals



Figure 4.3: Cluster Dendrogram for Air Toxics

Footnotes for Figure 4.3: Red boxes indicate clusters; Values at nodes indicate p-value percentages for clustering (higher percentage indicates tighter clustering); clusters were drawn on neighboring nodes with p-values greater than or equal to 95



Figure 4.4: Spearman Correlation between Air Toxics Heat Map

Footnotes for Figure 4.4: Spearman correlations were used between each pairwise combination; redder squares indicate greater positive correlation (i.e. greater concentration of one toxic leads to greater concentration in the other)



Figure 4.5: Bayesian Kernel Machine Regression Overall Risk – Expressive Language

Footnotes for Figure 4.5: Overall effect of chemical mixture of all air toxics on effect of belonging to non-optimal language trajectory; 50th percentile is used as baseline comparison; adjusted for child sex, maternal age, maternal race/ethnicity, maternal education and home-ownership




Footnote for Figure 4.6: Effect of single air toxic and probability of belonging to a non-optimal language trajectory associated with the exposure at 75th versus 25th percentile, while holding other air toxics at their 25th, 50th, and 75th percentiles; adjusted for child sex, maternal age, maternal race/ethnicity, maternal education and home-ownership





Footnote for Figure 4.7: Each point shows difference between effect size of air toxic when all other air toxics are held at 75th percentile and when all other air toxics are held at 25th percentile, including 95 percent confidence intervals; adjusted for child sex, maternal age, maternal race/ethnicity, maternal education and home-ownership



Figure 4.8: Bayesian Kernel Machine Regression Overall Risk – Receptive Language

Footnotes for Figure 4.8: Overall effect of chemical mixture of all air toxics on effect of belonging to non-optimal language trajectory; 50th percentile is used as baseline comparison; adjusted for child sex, maternal age, maternal race/ethnicity, maternal education and home-ownership



-4

ACRYLONITRILE -ACYRLAMIDE -ACROLEIN -ACETALDEHYDE -

Figure 4.9: Bayesian Kernel Machine Regression Single Exposure Effects – Receptive Language

Footnote for Figure 4.9: Effect of single air toxic and probability of belonging to a non-optimal language trajectory associated with the exposure at 75th versus 25th percentile, while holding other air toxics at their 25th, 50th, and 75th percentiles; adjusted for child sex, maternal age, maternal race/ethnicity, maternal education and home-ownership

0 est

Figure 4.10: Bayesian Kernel Machine Regression Interactive Effects - Receptive Language



Footnote for Figure 4.10: Each point shows difference between effect size of air toxic when all other air toxics are held at 75th percentile and when all other air toxics are held at 25th percentile, including 95 percent confidence intervals; adjusted for child sex, maternal age, maternal race/ethnicity, maternal education and home-ownership

Chapter 5 – Discussion, Conclusions, and Future Directions

Mixtures Analysis

Prior research has used a variety of methods to examine chemical mixtures. Loftus et al. used a weighted quantile sum regression (WQS) to examine the effects of a phthalate mixture on neurodevelopment, and Ramos et al. used a combination of principal component analysis (PCA) with structural equation modeling (SEM) to assess the effects of phthalates, organophosphate esters, and organophosphorous pesticides on language ability (107,206). WQS is also used in air toxics literature in order to assess air toxic mixtures (239,329) However, WQS has several limitations, namely that by using quantiles, information about the exposure is lost. In addition, while WQS may be used to assess the effects of a mixture, as well as determine which individual exposure contribute the most, it is unable to generate any measure of risk (208). Finally, the WQS assumes that all exposures in the weighted index have associations in the same direction with the outcome, an assumption that is highly unlikely to occur in an environmental mixture (330).

While we performed a mixtures analysis in our third analysis with air toxic exposures, we did not perform a mixtures analysis for our second analysis with phthalate exposures due to constraints on time. If we were to repeat this analysis, then a mixtures analysis should be performed, as exposure to phthalates rarely occurs in isolation. Because phthalates can degrade into multiple metabolites, it may also be prudent to use a mixture analysis in order to completely capture the effect of the parent compound. However, the metabolites may have differing mechanisms of actions that may lead to different physiological responses, so if this approach and rationale were to be used, then one must be certain that the metabolite in question all act via similar

pathophysiologies. In addition, while our analysis did not show that an air toxic mixture significantly influences language development trajectories, this may not be because a mixture does not have an effect. Our sample size was relatively small, which may have affected our ability to statistically resolve our BKMR, which can be sensitive to sample size. In addition, while none of the interactions were statistically significant, there was evidence of trends for certain air toxics, which indicates that a mixture of certain air toxics may impact language development trajectories.

Future research should focus on correcting the shortcomings of our mixture analysis. A mixtures analysis should be performed on all phthalate metabolites, ideally with a larger sample size in order to achieve adequate statistical power and resolution to observe any small effect sizes that may arise. Different methodology may be prudent to use as well; we utilized BKMR for its advantages over other methodologies. First, because it conducts variable selection and effect estimates simultaneously, it is more capable of capturing uncertainty in the exposure-response function compared to other methods. Second, it can examine both main effect and interaction components of the model while also accounting for uncertainty within the model (237,331). However, other methods may prove useful to answer other questions or to help support the results from a BKRM analysis. For example, weighted quantile sum regression may be able to provide a clearer picture on the effect of an overall mixture, and might be able to corroborate results taken from a BKMR analysis.

Use of High/Enhanced Risk Cohorts

The population used for this research consisted of families where the pregnant mother or father had a biological child that had already been diagnosed with autism.

The mother and the younger sibling were then followed through time. As a result, this population would be considered an enhanced risk autism cohort due to the heritable nature of autism. This also allowed us to observe a greater variability of potentially neurodivergent phenotypes, giving us greater power and ability to detect and characterize language development trajectories. However, this also limits the generalizability of our results. While enhanced risk cohorts have been used in the past to examine neurodevelopmental disorders and delays (195,196,332–334), we run the risk of not being able to appropriately apply our results to the general population, limiting the overall utility of our results. Future research should look at the general population in order to improve overall generalizability.

Use of Trajectory Analysis (Latent Class Growth Analysis)

The statistical method used for this research was latent class growth analysis (LCGA), a type of growth curve modeling that tries to predict latent classes using observed variables. This methodology may provide better insight into how prenatal exposure to endocrine disruptors impacts neurodevelopment. While not performed in this research, a comparison of this method to other cross-sectional methods would be prudent to observe any differences in risk, as well as assess precision (e.g. comparison of risk ratios obtained via LCGA to odds ratios from a more traditional logistic regression using a subscore from a specific time point). Ideally, we would see greater precision with the use of LCGA compared to a logistic regression.

Beyond comparison to cross-sectional methods, there are also other longitudinal methods that may be employed to assess risk using repeat measures. Such methods include mixed effect models, group based trajectory methods, and growth mixture

modeling. Prior studies have made use of such methods to assess the effects of environmental toxics on neurodevelopmental (335–337). Each method has their own strengths and weaknesses, but it is important to note that the trajectories determined in each study may be unique to that study's population (169). While trajectory analysis may prove useful in better assessing and describing health outcomes over time, they are not without their limitations, particularly when it comes to interpretability and generalizability of results.

Sex Differences in the Effect of Phthalates

We examined whether sex acted as a modifier of prenatal phthalate exposure and language, finding that it did not act as a modifier for expressive language, but did act as a modifier for some metabolites for receptive language. Our results showed that for the metabolites of di-isononyl phthalate (DiNP), females may have a higher risk of falling into the Tracking trajectory compared to males. For certain metabolites of DEHP, females may be at a lower risk for falling into the Late Growth trajectory compared to males, although both sexes see a decreased risk of falling into the trajectory in general.

This research is one of the few that has examined potential sex differences in the effects of phthalate exposure on language development. Olesen et al. performed a stratified analysis on boys and girls, but did not examine specifically whether sex was a modifier (105). However, the results indicated that there may be a sex difference present for some phthalate metabolites, as difference between effect measures were observed between girls and boys. For example, the metabolites for DEHP were shown to result in increased risk for boys, but decreased risk for girls, a result that is reflected in our own study. Dewey et al. also performed a stratified analysis, with their results

indicating that DEHP decreased language composite scores on the Bayley Scales of Infant Development – Third Edition for males, but increased it in females (338). While they did not compare the stratified results directly, it can be inferred that a sex difference is present.

Our study examined modification more directly by including a sex-phthalate interaction term and examining whether that term was statistically significant. We did not stratify by sex like Dewey and Olesen; rather, we kept all individuals in the analysis. Our results with DEHP largely agreed with both studies, namely that DEHP or metabolite for DEHP affected males more strongly than females, with males generally having heightened risk of falling into a suboptimal language trajectory compared to females. Thus, this study supports prior literature, and the use of an interaction term rather than stratification provides stronger evidence that a sex difference is present. Future studies should strongly consider performing a similar analysis in order to better separate the differential effects phthalates have on males versus females.

Discussion on Environmental Exposure

We examined prenatal phthalate exposure in two different studies using two different exposure pathways and methodologies. In one study, we assigned prenatal phthalate exposure through metabolite biomarker concentrations determined in urine samples provided by the pregnant mother, adjusting for creatinine or specific gravity depending on the specific cohort the child belonged to. In another study, certain phthalate exposures were included in the model as air toxics, where total concentration was assigned to each individual based on their birth year and census tract. Two

phthalates were common between both studies: dibutyl phthalate (DBP) and di(2ethylhexyl) phthalate (DEHP).

While results were consistent for DEHP across both studies, showing that DEHP decreased risk of falling into an abnormal language trajectory for both expressive and receptive language, the results for DBP differed between the two studies. When examined as an air toxic, DBP appeared to increase risk for falling into an abnormal language trajectory for both expressive and receptive language. However, when examined via biomarkers, DBP appeared to decrease the risk for falling into an abnormal language trajectory for receptive language.

This finding highlights the potential differences in association that may be found for any environmental exposure depending on the route of exposure taken as well as the measurement method used. Biomarkers are often considered to be the "gold standard" for measuring overall environmental exposure, though the nature of the toxicant and the biomarkers being used must be taken into consideration. In this instance, the use of urinary biomarkers to measure phthalate exposure is commonly regarded to be among the more accurate methods of measuring phthalate exposure, though the timing of urine collection is an important factor in the overall accuracy of this method (339). Since our study using urinary biomarkers consisted of multiple measurements, and because the parent cohorts had specified instructions for the collection of urine samples, the overall accuracy of our phthalate exposure is likely to be high. As a result, the study that used the urinary biomarkers is the stronger study when examining the impact of prenatal phthalate exposure on language development.

However, it still must be considered that these urinary biomarkers are measuring the metabolites of the phthalates, rather than the parent compound. The metabolites have their own toxicity when compared to their parent compound, and as such, we must consider the possibility that our results may be due to the effects of the metabolites, rather than the parent compound. This is counter to the study utilizing air toxics, which is measuring the effects of the parent compound. The different result with dibutyl phthalate may be a result of this.

In our third study, we examined the effects of air toxics that act as EDCs on language development by using the National Air Toxics Assessment. It logically follows that the route of exposure for our toxics was purely airborne, and that we did not consider other possible exposure routes for those phthalates. While this may appear to be a weakness of our study, and it can be considered one, we were interested in examining the effects of prenatal airborne exposure to these toxics, rather than all possible exposure.

While there have been many studies that have examined the same air toxics in the past, many of them do not specify airborne exposure only. For example, there are many studies that have examined the effects of pesticides on neurodevelopment. However, pesticide exposure can occur via multiple mediums, such as airborne exposure (i.e. inhalation), ingestion (e.g. via contaminated food or water), or dermal exposure (e.g. handling of pesticide containers or sprayers). As such, a study that examines a pesticide such as dichlorvos, trifluralin, or parathion may be capturing all possible routes of exposure, rather than just airborne exposure.

Our study is one of the few studies that specifically examined airborne exposure of air toxics and its potential relationship with language development. Of the air toxics that have been examined, airborne PAHs were found to be associated with a decrease in infant mental development across several studies using various measures of neurodevelopment, including the Bayley's Scales of Infant Development and Raven Coloured Progressive Matrices (21,61,62). Guxens found that benzene was negatively associated with infant mental development (225), and there have been multiple studies that have linked some airborne EDCs with an increased risk of ASD, including certain heavy metals, methylene chloride, styrene, certain conjugates of xylene, and other aromatic solvents (118,119,159,160). However, while some of the measures above measure some degree of language, none of the studies specifically looked at language. Our study therefore provides vital information on the impact of airborne exposure to EDCs and their potential impact on language development.

When examining and measuring environmental exposures, it is prudent to consider the route we are assuming the exposure is taking and whether this route is the prevalent or biologically viable pathway. As mentioned above, air toxics are not found solely in the air; lead, pesticides, and some VOCs are found in other mediums, such as water, food, or personal care products. These additional routes may result in an underestimation of total exposure to the toxic in question. However, airborne exposure may be the primary route for some toxics, even if there are other routes of exposure. As a result, airborne exposure may be the most important route to examine. Furthermore, it may not always be viable to collect exposure data to the level of detail and granularity that would give us the true level of exposure for an individual. In these instances, it may

be more prudent to use exposure data that is more readily available, even at the cost of accuracy. The findings from such a study are not invalid, but may be used to justify further examination and study, which may lead to more accurate exposure data collection. This, in turn, will lead to stronger studies that can help guide and inform new policies to reduce the exposure and impact of these environmental contaminants.

Future Examination of Language Development as an Outcome

Our primary outcome of interest was language development. Because our population included two cohorts that had an enhanced familial risk for autism, we were able to detect more nuanced differences in neurodevelopment than a more generalized cohort. Future studies should aim to examine language in a more generalizable cohort in order to determine if the patterns observed in this research can be found in a cohort that is not enriched for autism or other neurodevelopmental disorders.

Furthermore, the use of language impairment as an outcome may help to better identify individuals who may go on to develop other neurodevelopmental disorders. Some studies have noted that individuals that present with language impairments go on to develop other neurodevelopmental disorders such as autism (340,341). Identification of individuals with language development impairments may be crucial to improve the gender gap in neurodevelopmental diagnoses, such as in autism, where females are often under-diagnosed (342,343). This is due to language milestones being relatively gender agnostic which may result in a more objective assessment on neurodevelopment. Pediatricians, parents, and caretakers may then use these assessments in order to refer a child for further psychometric or diagnostic testing in order to determine if a child may possess a neurodevelopmental disorder, or that the

child may need to be reassessed at a later time to determine if sub-clinical symptoms worsen such that a diagnosis can be made.

Conclusions

We performed three studies in order to assess the effects of endocrine disruptors on language development. The first analysis examined if child neurodevelopmental diagnosis was related to overall language development trajectory. We found that diagnostic status was related to language trajectories, with children who were classified as ASD or non-TD being more likely to fall into trajectories that would indicate deficits in language development.

The second analysis examined if prenatal exposure to various phthalates affected language development. Most of the phthalates examined failed to statistically resolve. However, trends were observed among the phthalates based on their weights, with low molecular weight phthalates generally showing an increased risk of falling into an abnormal language development trajectory and high molecular weight phthalates generally showing a decreased risk instead. We further found that sex was not a modifier for expressive language, but was a modifier for receptive language for the metabolites of DiNP (mCOP and mNP2) and the metabolites of DEHP (mEHP, mEHHP, mEOHP, and mECPP), showing a stronger protective effect for females versus males for falling into the Late Growth trajectory.

The third analysis examined if prenatal exposure to specific air toxics affected language development. After adjusting for multiple testing via false discovery rate, only acetaldehyde was statistically significant, showing an overall decrease in risk for

belonging to an abnormal language trajectory. Further, we found that a model that contained all air toxics did not appear to have an impact on overall language trajectory, and that there did not appear to be any statistically significant interactions between the multiple air toxics, although some trends were observed with some air toxics, with mixtures with those air toxics showing a protective effect.

In summary, our findings indicate that most EDCs had a null effect on language development trajectories. Of the EDCs that had a significant effect, most were protective in nature, decreasing the risk of falling into a non-optimal language trajectory with increasing concentration. Furthermore, we found that sex may be acting as a modifier for phthalates with respect to receptive language, and that air toxic mixtures do not significantly impact the risk of falling into a non-optimal trajectory. Future research should focus on examining a larger and more representative population, ideally with multiple outcome measurements in order to continue to better capture the progression of language and neurodevelopment.

References

- 1. Howlin P, Udwin O. Outcomes in Neurodevelopmental and Genetic Disorders. Cambridge University Press; 2002 306 p.
- 2. Johnson CJ, Beitchman JH, Young A, et al. Fourteen-Year Follow-Up of Children With and Without Speech/Language ImpairmentsSpeech/Language Stability and Outcomes. 1999;17.
- 3. Snowling MJ, Bishop DVM, Stothard SE, et al. Psychosocial outcomes at 15 years of children with a preschool history of speech-language impairment. *J. Child Psychol. Psychiatry*. 2006;47(8):759–765.
- 4. Whitehouse AJO, Watt HJ, Line EA, et al. Adult psychosocial outcomes of children with specific language impairment, pragmatic language impairment and autism. *Int. J. Lang. Commun. Disord.* 2009;44(4):511–528.
- 5. Beitchman J, Brownlie E. Language Development and its Impact on Children's Psychosocial and Emotional Development. 2012;
- 6. Irwin JR, Carter AS, Briggs-gowan MJ. The Social-Emotional Development of "Late-Talking" Toddlers. *J. Am. Acad. Child Adolesc. Psychiatry*. 2002;41(11):1324–1332.
- Roulstone S, Law J, Rush R, et al. Investigating the role of language in children's early educational outcomes. 2011;(https://eresearch.qmu.ac.uk/handle/20.500.12289/2484). (Accessed December 14, 2020)
- American Psychiatric Association. Diagnostic and Statistical Manual of Mental Disorders. Fifth Edition. American Psychiatric Association; 2013 (Accessed September 1, 2020).(http://psychiatryonline.org/doi/book/10.1176/appi.books.9780890425596). (Accessed September 1, 2020)
- 9. Mueller KL, Tomblin JB. Examining the comorbidity of language disorders and ADHD. *Top. Lang. Disord.* 2012;32(3):228–246.
- 10. McGrath LM, Hutaff-Lee C, Scott A, et al. Children with Comorbid Speech Sound Disorder and Specific Language Impairment are at Increased Risk for Attention-Deficit/Hyperactivity Disorder. J. Abnorm. Child Psychol. 2008;36(2):151–163.
- 11. Conti-Ramsden G, Simkin Z, Botting N. The prevalence of autistic spectrum disorders in adolescents with a history of specific language impairment (SLI). *J. Child Psychol. Psychiatry*. 2006;47(6):621–628.
- 12. Booth JR, Wood L, Lu D, et al. The role of the basal ganglia and cerebellum in language processing. *Brain Res.* 2007;1133:136–144.
- Watkins K. Chapter 13 Developmental disorders of speech and language: From genes to brain structure and function. In: Braddick O, Atkinson J, Innocenti GM, eds. *Progress in Brain Research*. Elsevier; 2011 (Accessed January 14, 2021):225–

238.(http://www.sciencedirect.com/science/article/pii/B9780444538840000270). (Accessed January 14, 2021)

- 14. Davis C, Kleinman JT, Newhart M, et al. Speech and language functions that require a functioning Broca's area. *Brain Lang.* 2008;105(1):50–58.
- 15. Fedorenko E, Duncan J, Kanwisher N. Language-Selective and Domain-General Regions Lie Side by Side within Broca's Area. *Curr. Biol.* 2012;22(21):2059–2062.
- 16. Hagoort P. Nodes and networks in the neural architecture for language: Broca's region and beyond. *Curr. Opin. Neurobiol.* 2014;28:136–141.
- Stewart PW, Lonky Edward, Reihman Jacqueline, et al. The Relationship between Prenatal PCB Exposure and Intelligence (IQ) in 9-Year-Old Children. *Environ. Health Perspect.* 2008;116(10):1416–1422.
- 18. Shah S, Jeong KS, Park H, et al. Environmental pollutants affecting children's growth and development: Collective results from the MOCEH study, a multi-centric prospective birth cohort in Korea. *Environ. Int.* 2020;137:105547.
- 19. Harris MH, Gold Diane R., Rifas-Shiman Sheryl L., et al. Prenatal and Childhood Traffic-Related Pollution Exposure and Childhood Cognition in the Project Viva Cohort (Massachusetts, USA). *Environ. Health Perspect.* 2015;123(10):1072–1078.
- 20. Porta D, Narduzzi S, Badaloni C, et al. Air Pollution and Cognitive Development at Age 7 in a Prospective Italian Birth Cohort. *Epidemiology*. 2016;27(2):228–236.
- 21. Perera FP, Li Z, Whyatt R, et al. Prenatal Airborne Polycyclic Aromatic Hydrocarbon Exposure and Child IQ at Age 5 Years. *Pediatrics*. 2009;124(2):e195–e202.
- 22. Tang D, Li Tin-yu, Liu Jason J., et al. Effects of Prenatal Exposure to Coal-Burning Pollutants on Children's Development in China. *Environ. Health Perspect.* 2008;116(5):674–679.
- 23. Ehrenstein OS von, Ling C, Cui X, et al. Prenatal and infant exposure to ambient pesticides and autism spectrum disorder in children: population based case-control study. *BMJ* [electronic article]. 2019;364. (https://www.bmj.com/content/364/bmj.l962). (Accessed June 28, 2020)
- Kalkbrenner AE, Windham GC, Zheng C, et al. Air Toxics in Relation to Autism Diagnosis, Phenotype, and Severity in a U.S. Family-Based Study. *Environ. Health Perspect.* 2018;126(3):037004.
- 25. Oh J, Bennett DH, Calafat AM, et al. Prenatal exposure to per- and polyfluoroalkyl substances in association with autism spectrum disorder in the MARBLES study. *Environ. Int.* 2020;147:106328.
- 26. Berkman ND, Wallace I, Watson L, et al. Screening for Speech and Language Delays and Disorders in Children Age 5 Years or Younger: A Systematic Review for the U.S. Preventive Services Task Force. Rockville (MD): Agency for Healthcare Research and Quality (US); 2015 (Accessed January 14, 2021).(http://www.ncbi.nlm.nih.gov/books/NBK305674/). (Accessed January 14, 2021)

- 27. Prathanee B, Thinkhamrop B, Dechongkit S. Factors Associated With Specific Language Impairment and Later Language Development During Early Life: A Literature Review. *Clin. Pediatr.* (*Phila.*). 2007;46(1):22–29.
- 28. Rudolph JM. Case History Risk Factors for Specific Language Impairment: A Systematic Review and Meta-Analysis. *Am. J. Speech Lang. Pathol.* 2017;26(3):991–1010.
- 29. Stanton-Chapman TL, Chapman DA, Bainbridge NL, et al. Identification of early risk factors for language impairment. *Res. Dev. Disabil.* 2002;23(6):390–405.
- 30. Tomblin JB, Smith E, Zhang X. Epidemiology of specific language impairment: Prenatal and perinatal risk factors. *J. Commun. Disord.* 1997;30(4):325–344.
- 31. Carr RC, Bratsch-Hines M, Varghese C, et al. Latent class growth trajectories of letter name knowledge during pre-kindergarten and kindergarten. J. Appl. Dev. Psychol. 2020;69:101141.
- 32. Ukoumunne OC, Wake M, Carlin J, et al. Profiles of language development in pre-school children: a longitudinal latent class analysis of data from the Early Language in Victoria Study. *Child Care Health Dev.* 2012;38(3):341–349.
- 33. McKean C, Wraith D, Eadie P, et al. Subgroups in language trajectories from 4 to 11 years: the nature and predictors of stable, improving and decreasing language trajectory groups. *J. Child Psychol. Psychiatry*. 2017;58(10):1081–1091.
- 34. Law J, Tomblin BJ, Zhang X. Characterizing the Growth Trajectories of Language-Impaired Children Between 7 and 11 Years of Age. J. Speech Lang. Hear. Res. 2008;51(3):739–749.
- 35. Conti-Ramsden G, St Clair MC, Pickles A, et al. Developmental Trajectories of Verbal and Nonverbal Skills in Individuals With a History of Specific Language Impairment: From Childhood to Adolescence. J. Speech Lang. Hear. Res. 2012;55(6):1716–1735.
- 36. Tambyraja SR, Schmitt Mary Beth, Farquharson Kelly, et al. Stability of Language and Literacy Profiles of Children With Language Impairment in the Public Schools. *J. Speech Lang. Hear. Res.* 2015;58(4):1167–1181.
- Pickles A, Anderson DK, Lord C. Heterogeneity and plasticity in the development of language: a 17-year follow-up of children referred early for possible autism. J. Child Psychol. Psychiatry. 2014;55(12):1354–1362.
- 38. Landa RJ, Gross AL, Stuart EA, et al. Latent class analysis of early developmental trajectory in baby siblings of children with autism. *J. Child Psychol. Psychiatry*. 2012;53(9):986–996.
- 39. Brignell A, Williams K, Jachno K, et al. Patterns and Predictors of Language Development from 4 to 7 Years in Verbal Children With and Without Autism Spectrum Disorder. *J. Autism Dev. Disord.* 2018;48(10):3282–3295.
- 40. Landa RJ, Gross AL, Stuart EA, et al. Developmental Trajectories in Children With and Without Autism Spectrum Disorders: The First 3 Years. *Child Dev.* 2013;84(2):429–442.

- 41. Bernal J. Thyroid hormone receptors in brain development and function. *Nat. Clin. Pract. Endocrinol. Metab.* 2007;3(3):249–259.
- 42. Horn S, Heuer H. Thyroid hormone action during brain development: More questions than answers. *Mol. Cell. Endocrinol.* 2010;315(1):19–26.
- 43. Escobar GM de, Obregón MJ, Rey FE del. Maternal thyroid hormones early in pregnancy and fetal brain development. *Best Pract. Res. Clin. Endocrinol. Metab.* 2004;18(2):225–248.
- 44. Ghassabian A, El Marroun H, Peeters RP, et al. Downstream Effects of Maternal Hypothyroxinemia in Early Pregnancy: Nonverbal IQ and Brain Morphology in School-Age Children. J. Clin. Endocrinol. Metab. 2014;99(7):2383–2390.
- 45. Päkkilä F, Männistö T, Pouta A, et al. The Impact of Gestational Thyroid Hormone Concentrations on ADHD Symptoms of the Child. *J. Clin. Endocrinol. Metab.* 2014;99(1):E1–E8.
- 46. Román GC, Ghassabian A, Bongers-Schokking JJ, et al. Association of gestational maternal hypothyroxinemia and increased autism risk. *Ann. Neurol.* 2013;74(5):733–742.
- 47. Skeaff SA. Iodine Deficiency in Pregnancy: The Effect on Neurodevelopment in the Child. *Nutrients*. 2011;3(2):265–273.
- 48. Ding D, Xu L, Fang H, et al. The EDKB: an established knowledge base for endocrine disrupting chemicals. *BMC Bioinformatics*. 2010;11(6):S5.
- 49. Tabb MM, Blumberg B. New Modes of Action for Endocrine-Disrupting Chemicals. *Mol. Endocrinol.* 2006;20(3):475–482.
- Ghassabian A, Trasande L. Disruption in Thyroid Signaling Pathway: A Mechanism for the Effect of Endocrine-Disrupting Chemicals on Child Neurodevelopment. *Front. Endocrinol.* [electronic article]. 2018;9.
 (https://www.frontiersin.org/journals/endocrinology/articles/10.3389/fendo.2018.00204/full).
 (Accessed April 7, 2024)
- 51. Schug TT, Blawas AM, Gray K, et al. Elucidating the Links Between Endocrine Disruptors and Neurodevelopment. *Endocrinology*. 2015;156(6):1941–1951.
- 52. Trasande L, Zoeller RT, Hass U, et al. Burden of disease and costs of exposure to endocrine disrupting chemicals in the European Union: an updated analysis. *Andrology*. 2016;4(4):565–572.
- 53. Kahn LG, Philippat C, Nakayama SF, et al. Endocrine-disrupting chemicals: implications for human health. *Lancet Diabetes Endocrinol.* 2020;8(8):703–718.
- 54. Sagiv SK, Thurston SW, Bellinger DC, et al. Prenatal Organochlorine Exposure and Behaviors Associated With Attention Deficit Hyperactivity Disorder in School-Aged Children. *Am. J. Epidemiol.* 2010;171(5):593–601.

- 55. Park H-Y, Hertz-Picciotto I, Sovcikova E, et al. Neurodevelopmental toxicity of prenatal polychlorinated biphenyls (PCBs) by chemical structure and activity: a birth cohort study. *Environ. Health*. 2010;9(1):51.
- 56. Costa LG, de Laat R, Tagliaferri S, et al. A mechanistic view of polybrominated diphenyl ether (PBDE) developmental neurotoxicity. *Toxicol. Lett.* 2014;230(2):282–294.
- 57. Costa LG, Giordano G. Is decabromodiphenyl ether (BDE-209) a developmental neurotoxicant? *NeuroToxicology*. 2011;32(1):9–24.
- 58. Schreiber T, Gassmann Kathrin, Götz Christine, et al. Polybrominated Diphenyl Ethers Induce Developmental Neurotoxicity in a Human in Vitro Model: Evidence for Endocrine Disruption. *Environ. Health Perspect.* 2010;118(4):572–578.
- 59. Eskenazi Brenda, Chevrier Jonathan, Rauch Stephen A., et al. In Utero and Childhood Polybrominated Diphenyl Ether (PBDE) Exposures and Neurodevelopment in the CHAMACOS Study. *Environ. Health Perspect.* 2013;121(2):257–262.
- 60. Vuong AM, Yolton K, Dietrich KN, et al. Exposure to polybrominated diphenyl ethers (PBDEs) and child behavior: Current findings and future directions. *Horm. Behav.* 2018;101:94–104.
- 61. Edwards SC, Jedrychowski W, Butscher M, et al. Prenatal Exposure to Airborne Polycyclic Aromatic Hydrocarbons and Children's Intelligence at 5 Years of Age in a Prospective Cohort Study in Poland. *Environ. Health Perspect.* 2010;118(9):1326–1331.
- 62. Perera FP, Rauh V, Whyatt RM, et al. Effect of Prenatal Exposure to Airborne Polycyclic Aromatic Hydrocarbons on Neurodevelopment in the First 3 Years of Life among Inner-City Children. Environ. Health Perspect. 2006;114(8):1287–1292.
- 63. Tang Y, Donnelly KC, Tiffany-Castiglioni E, et al. Neurotoxicity of Polycyclic Aromatic Hydrocarbons and Simple Chemical Mixtures. *J. Toxicol. Environ. Health A*. 2003;66(10):919–940.
- 64. Bouchard MF, Bellinger DC, Wright RO, et al. Attention-Deficit/Hyperactivity Disorder and Urinary Metabolites of Organophosphate Pesticides. *Pediatrics*. 2010;125(6):e1270–e1277.
- 65. Bouchard MF, Chevrier Jonathan, Harley Kim G., et al. Prenatal Exposure to Organophosphate Pesticides and IQ in 7-Year-Old Children. *Environ. Health Perspect.* 2011;119(8):1189–1195.
- 66. Engel SM, Wetmur James, Chen Jia, et al. Prenatal Exposure to Organophosphates, Paraoxonase 1, and Cognitive Development in Childhood. *Environ. Health Perspect.* 2011;119(8):1182–1188.
- 67. Engel SM, Bradman Asa, Wolff Mary S., et al. Prenatal Organophosphorus Pesticide Exposure and Child Neurodevelopment at 24 Months: An Analysis of Four Birth Cohorts. *Environ. Health Perspect.* 2016;124(6):822–830.
- Harari R, Julvez Jordi, Murata Katsuyuki, et al. Neurobehavioral Deficits and Increased Blood Pressure in School-Age Children Prenatally Exposed to Pesticides. *Environ. Health Perspect.* 2010;118(6):890–896.

- 69. London L, Beseler C, Bouchard MF, et al. Neurobehavioral and neurodevelopmental effects of pesticide exposures. *NeuroToxicology*. 2012;33(4):887–896.
- Rauh V, Arunajadai Srikesh, Horton Megan, et al. Seven-Year Neurodevelopmental Scores and Prenatal Exposure to Chlorpyrifos, a Common Agricultural Pesticide. *Environ. Health Perspect.* 2011;119(8):1196–1201.
- Xu X, Nembhard WN, Kan H, et al. Urinary trichlorophenol levels and increased risk of attention deficit hyperactivity disorder among US school-aged children. *Occup. Environ. Med.* 2011;68(8):557–561.
- 72. Rovet JF, Ehrlich RM, Sorbara DL. Neurodevelopment in Infants and Preschool Children with Congenital Hypothyroidism: Etiological and Treatment Factors Affecting Outcome 1. J. Pediatr. Psychol. 1992;17(2):187–213.
- 73. Quast A, Hesse V, Hain J, et al. Baby babbling at five months linked to sex hormone levels in early infancy. *Infant Behav. Dev.* 2016;44:1–10.
- 74. Bornehag C-G, Lindh C, Reichenberg A, et al. Association of Prenatal Phthalate Exposure With Language Development in Early Childhood. *JAMA Pediatr.* 2018;172(12):1169–1176.
- 75. Kao C-C, Que DE, Bongo SJ, et al. Residue Levels of Organochlorine Pesticides in Breast Milk and Its Associations with Cord Blood Thyroid Hormones and the Offspring's Neurodevelopment. *Int. J. Environ. Res. Public. Health.* 2019;16(8):1438.
- 76. Zhang J, Guo J, Wu C, et al. Exposure to carbamate and neurodevelopment in children: Evidence from the SMBCS cohort in China. *Environ. Res.* 2019;177:108590.
- 77. Chen X, Xu S, Tan T, et al. Toxicity and Estrogenic Endocrine Disrupting Activity of Phthalates and Their Mixtures. *Int. J. Environ. Res. Public. Health*. 2014;11(3):3156–3168.
- 78. Engel SM, Villanger GD, Nethery RC, et al. Prenatal Phthalates, Maternal Thyroid Function, and Risk of Attention-Deficit Hyperactivity Disorder in the Norwegian Mother and Child Cohort. *Environ. Health Perspect.* 2018;126(5):057004.
- 79. Lee BM, Koo HJ. Hershberger Assay for Antiandrogenic Effects of Phthalates. *J. Toxicol. Environ. Health A*. 2007;70(15–16):1365–1370.
- 80. Li X-J, Jiang L, Chen L, et al. Neurotoxicity of dibutyl phthalate in brain development following perinatal exposure: A study in rats. *Environ. Toxicol. Pharmacol.* 2013;36(2):392–402.
- 81. Radke EG, Braun JM, Nachman RM, et al. Phthalate exposure and neurodevelopment: A systematic review and meta-analysis of human epidemiological evidence. *Environ. Int.* 2020;137:105408.
- Rowdhwal SSS, Chen J. Toxic Effects of Di-2-ethylhexyl Phthalate: An Overview. *BioMed Res. Int.* 2018;2018:e1750368. (https://www.hindawi.com/journals/bmri/2018/1750368/). (Accessed January 12, 2021)

- 83. Weaver JA, Beverly BEJ, Keshava N, et al. Hazards of diethyl phthalate (DEP) exposure: A systematic review of animal toxicology studies. *Environ. Int.* 2020;145:105848.
- 84. Radke EG, Galizia A, Thayer KA, et al. Phthalate exposure and metabolic effects: a systematic review of the human epidemiological evidence. *Environ. Int.* 2019;132:104768.
- 85. Swedenborg E, Rüegg J, Mäkelä S, et al. Endocrine disruptive chemicals: mechanisms of action and involvement in metabolic disorders. *J. Mol. Endocrinol.* 2009;43(1):1–10.
- 86. Radke EG, Glenn BS, Braun JM, et al. Phthalate exposure and female reproductive and developmental outcomes: a systematic review of the human epidemiological evidence. *Environ. Int.* 2019;130:104580.
- 87. Wójtowicz AK, Szychowski KA, Wnuk A, et al. Dibutyl Phthalate (DBP)-Induced Apoptosis and Neurotoxicity are Mediated via the Aryl Hydrocarbon Receptor (AhR) but not by Estrogen Receptor Alpha (ERα), Estrogen Receptor Beta (ERβ), or Peroxisome Proliferator-Activated Receptor Gamma (PPARy) in Mouse Cortical Neurons. *Neurotox. Res.* 2017;31(1):77–89.
- Xu H, Shao X, Zhang Z, et al. Effects of Di-n-butyl Phthalate and Diethyl Phthalate on Acetylcholinesterase Activity and Neurotoxicity Related Gene Expression in Embryonic Zebrafish. *Bull. Environ. Contam. Toxicol.* 2013;91(6):635–639.
- Hannas BR, Lambright CS, Furr J, et al. Dose-Response Assessment of Fetal Testosterone Production and Gene Expression Levels in Rat Testes Following InUtero Exposure to Diethylhexyl Phthalate, Diisobutyl Phthalate, Diisoheptyl Phthalate, and Diisononyl Phthalate. *Toxicol. Sci.* 2011;123(1):206–216.
- Ye H, Ha M, Yang M, et al. Di2-ethylhexyl phthalate disrupts thyroid hormone homeostasis through activating the Ras/Akt/TRHr pathway and inducing hepatic enzymes. *Sci. Rep.* 2017;7(1):40153.
- 91. Jeddi MZ, Janani L, Memari AH, et al. The role of phthalate esters in autism development: A systematic review. *Environ. Res.* 2016;151:493–504.
- 92. Ku H-Y, Tsai T-L, Wang P-L, et al. Prenatal and childhood phthalate exposure and attention deficit hyperactivity disorder traits in child temperament: A 12-year follow-up birth cohort study. *Sci. Total Environ.* 2020;699:134053.
- 93. Factor-Litvak P, Insel B, Calafat AM, et al. Persistent Associations between Maternal Prenatal Exposure to Phthalates on Child IQ at Age 7 Years. *PLOS ONE*. 2014;9(12):e114003.
- 24. Zhang Q, Chen X-Z, Huang X, et al. The association between prenatal exposure to phthalates and cognition and neurobehavior of children-evidence from birth cohorts. *NeuroToxicology*. 2019;73:199–212.
- 95. Balalian AA, Whyatt RM, Liu X, et al. Prenatal and childhood exposure to phthalates and motor skills at age 11 years. *Environ. Res.* 2019;171:416–427.

- 96. Daniel S, Balalian AA, Insel BJ, et al. Prenatal and early childhood exposure to phthalates and childhood behavior at age 7 years. *Environ. Int.* 2020;143:105894.
- 97. Gascon M, Valvi D, Forns J, et al. Prenatal exposure to phthalates and neuropsychological development during childhood. *Int. J. Hyg. Environ. Health*. 2015;218(6):550–558.
- 98. Li N, Papandonatos GD, Calafat AM, et al. Gestational and childhood exposure to phthalates and child behavior. *Environ. Int.* 2020;144:106036.
- 99. Cowell WJ, Wright RJ. Sex-Specific Effects of Combined Exposure to Chemical and Non-chemical Stressors on Neuroendocrine Development: a Review of Recent Findings and Putative Mechanisms. *Curr. Environ. Health Rep.* 2017;4(4):415–425.
- 100. Holahan MR, Smith CA. Phthalates and neurotoxic effects on hippocampal network plasticity. *NeuroToxicology*. 2015;48:21–34.
- 101. Ponsonby A-L, Symeonides C, Vuillermin P, et al. Epigenetic regulation of neurodevelopmental genes in response to in utero exposure to phthalate plastic chemicals: How can we delineate causal effects? *NeuroToxicology*. 2016;55:92–101.
- 102. Xu S, Zhang H, Pao P-C, et al. Exposure to phthalates impaired neurodevelopment through estrogenic effects and induced DNA damage in neurons. *Aquat. Toxicol.* 2020;222:105469.
- 103. Huang P-C, Tsai C-H, Chen C-C, et al. Intellectual evaluation of children exposed to phthalatetainted products after the 2011 Taiwan phthalate episode. *Environ. Res.* 2017;156:158–166.
- Jones B, Han T-L, Delplancke T, et al. Association between maternal exposure to phthalates and lower language ability in offspring derived from hair metabolome analysis. *Sci. Rep.* 2018;8(1):6745.
- 105. Olesen TS, Bleses D, Andersen HR, et al. Prenatal phthalate exposure and language development in toddlers from the Odense Child Cohort. *Neurotoxicol. Teratol.* 2018;65:34–41.
- 106. Hyland C, Mora Ana M., Kogut Katherine, et al. Prenatal Exposure to Phthalates and Neurodevelopment in the CHAMACOS Cohort. *Environ. Health Perspect.* 2019;127(10):107010.
- 107. Loftus CT, Bush NR, Day DB, et al. Exposure to prenatal phthalate mixtures and neurodevelopment in the Conditions Affecting Neurocognitive Development and Learning in Early childhood (CANDLE) study. *Environ. Int.* 2021;150:106409.
- 108. Polanska K, Ligocka D, Sobala W, et al. Phthalate exposure and child development: The Polish Mother and Child Cohort Study. *Early Hum. Dev.* 2014;90(9):477–485.
- 109. Sree CG, Buddolla V, Lakshmi BA, et al. Phthalate toxicity mechanisms: An update. *Comp. Biochem. Physiol. Part C Toxicol. Pharmacol.* 2023;263:109498.
- 110. Morgenstern R, Whyatt RM, Insel BJ, et al. Phthalates and thyroid function in preschool age children: Sex specific associations. *Environ. Int.* 2017;106:11–18.

- 111. Wang J-Q, Hu Y-B, Gao H, et al. Sex-specific difference in placental inflammatory transcriptional biomarkers of maternal phthalate exposure: a prospective cohort study. *J. Expo. Sci. Environ. Epidemiol.* 2020;30(5):835–844.
- 112. Whyatt RM, Liu Xinhua, Rauh Virginia A., et al. Maternal Prenatal Urinary Phthalate Metabolite Concentrations and Child Mental, Psychomotor, and Behavioral Development at 3 Years of Age. *Environ. Health Perspect.* 2012;120(2):290–295.
- 113. Téllez-Rojo MM, Cantoral A, Cantonwine DE, et al. Prenatal urinary phthalate metabolites levels and neurodevelopment in children at two and three years of age. *Sci. Total Environ.* 2013;461–462:386–390.
- 114. Etchell A, Adhikari A, Weinberg LS, et al. A systematic literature review of sex differences in childhood language and brain development. *Neuropsychologia*. 2018;114:19–31.
- 115. May T, Adesina I, McGillivray J, et al. Sex differences in neurodevelopmental disorders. *Curr. Opin. Neurol.* 2019;32(4):622–626.
- Schwela D (Dieter). Chapter 27 Environmental toxicology: air. In: Wexler P, ed. Information Resources in Toxicology (Fifth Edition). Academic Press; 2020 (Accessed January 26, 2021):279– 320.(http://www.sciencedirect.com/science/article/pii/B978012813724600027X). (Accessed January 26, 2021)
- 117. US EPA O. About Urban Air Toxics. *US EPA*. 2014;(https://www.epa.gov/urban-air-toxics/about-urban-air-toxics). (Accessed January 28, 2021)
- 118. Talbott EO, Marshall LP, Rager JR, et al. Air toxics and the risk of autism spectrum disorder: the results of a population based case–control study in southwestern Pennsylvania. *Environ. Health*. 2015;14(1):80.
- 119. Windham GC, Zhang L, Gunier R, et al. Autism Spectrum Disorders in Relation to Distribution of Hazardous Air Pollutants in the San Francisco Bay Area. *Environ. Health Perspect.* 2006;114(9):1438–1444.
- Zhang H, Yolton Kimberly, Webster Glenys M., et al. Prenatal PBDE and PCB Exposures and Reading, Cognition, and Externalizing Behavior in Children. *Environ. Health Perspect.* 2017;125(4):746–752.
- 121. Delgado-Saborit JM, Guercio V, Gowers AM, et al. A critical review of the epidemiological evidence of effects of air pollution on dementia, cognitive function and cognitive decline in adult population. *Sci. Total Environ.* 2021;757:143734.
- 122. Zhou Y, Li C, Huijbregts MAJ, et al. Carcinogenic Air Toxics Exposure and Their Cancer-Related Health Impacts in the United States. *PLOS ONE*. 2015;10(10):e0140013.
- 123. Buha Djordjevic A, Antonijevic E, Curcic M, et al. Endocrine-disrupting mechanisms of polychlorinated biphenyls. *Curr. Opin. Toxicol.* 2020;19:42–49.

- 124. Zheng J, He C-T, Chen S-J, et al. Disruption of thyroid hormone (TH) levels and TH-regulated gene expression by polybrominated diphenyl ethers (PBDEs), polychlorinated biphenyls (PCBs), and hydroxylated PCBs in e-waste recycling workers. *Environ. Int.* 2017;102:138–144.
- 125. Brar NK, Waggoner C, Reyes JA, et al. Evidence for thyroid endocrine disruption in wild fish in San Francisco Bay, California, USA. Relationships to contaminant exposures. *Aquat. Toxicol.* 2010;96(3):203–215.
- 126. Yu L, Lam JCW, Guo Y, et al. Parental Transfer of Polybrominated Diphenyl Ethers (PBDEs) and Thyroid Endocrine Disruption in Zebrafish. *Environ. Sci. Technol.* 2011;45(24):10652–10659.
- 127. Ding G, Yu J, Chen L, et al. Polybrominated diphenyl ethers (PBDEs) and thyroid hormones in cord blood. *Environ. Pollut.* 2017;229:489–495.
- 128. Dickerson SM, Cunningham SL, Patisaul HB, et al. Endocrine Disruption of Brain Sexual Differentiation by Developmental PCB Exposure. *Endocrinology*. 2011;152(2):581–594.
- 129. Seegal RF, Brosch KO, Okoniewski RJ. Coplanar PCB Congeners Increase Uterine Weight and Frontal Cortical Dopamine in the Developing Rat: Implications for Developmental Neurotoxicity. *Toxicol. Sci.* 2005;86(1):125–131.
- 130. Alarcan J, Waizenegger J, Solano M de LM, et al. Hepatotoxicity of the pesticides imazalil, thiacloprid and clothianidin Individual and mixture effects in a 28-day study in female Wistar rats. *Food Chem. Toxicol.* 2020;140:111306.
- 131. Pereira LC, Cabral Miranda LF, Franco-Bernardes MF, et al. Mitochondrial damage and apoptosis: Key features in BDE-153-induced hepatotoxicity. *Chem. Biol. Interact.* 2018;291:192–201.
- 132. Sun Y, Wang Y, Liang B, et al. Hepatotoxicity of decabromodiphenyl ethane (DBDPE) and decabromodiphenyl ether (BDE-209) in 28-day exposed Sprague-Dawley rats. *Sci. Total Environ.* 2020;705:135783.
- 133. Cecchi A, Rovedatti MG, Sabino G, et al. Environmental exposure to organophosphate pesticides: Assessment of endocrine disruption and hepatotoxicity in pregnant women. *Ecotoxicol. Environ. Saf.* 2012;80:280–287.
- Hayes TB, Case Paola, Chui Sarah, et al. Pesticide Mixtures, Endocrine Disruption, and Amphibian Declines: Are We Underestimating the Impact? *Environ. Health Perspect.* 2006;114(Suppl 1):40– 50.
- 135. Senthilkumaran B. Pesticide- and sex steroid analogue-induced endocrine disruption differentially targets hypothalamo-hypophyseal-gonadal system during gametogenesis in teleosts A review. *Gen. Comp. Endocrinol.* 2015;219:136–142.
- 136. Silvia SC, Magnarelli G, Rovedatti MG. Evaluation of endocrine disruption and gestational disorders in women residing in areas with intensive pesticide application: An exploratory study. *Environ. Toxicol. Pharmacol.* 2020;73:103280.

- Singh PB, Canario AVM. Reproductive endocrine disruption in the freshwater catfish, Heteropneustes fossilis, in response to the pesticide γ-hexachlorocyclohexane. *Ecotoxicol. Environ. Saf.* 2004;58(1):77–83.
- 138. Diggs DL, Harris KL, Rekhadevi PV, et al. Tumor microsomal metabolism of the food toxicant, benzo(a)pyrene, in ApcMinmouse model of colon cancer. *Tumor Biol.* 2012;33(4):1255–1260.
- 139. Kim K-H, Jahan SA, Kabir E, et al. A review of airborne polycyclic aromatic hydrocarbons (PAHs) and their human health effects. *Environ. Int.* 2013;60:71–80.
- 140. Kristensen P, Eilertsen E, Einarsdóttir E, et al. Fertility in mice after prenatal exposure to benzo[a]pyrene and inorganic lead. *Environ. Health Perspect.* 1995;103(6):588–590.
- 141. Shen H, Tao S, Liu J, et al. Global lung cancer risk from PAH exposure highly depends on emission sources and individual susceptibility. *Sci. Rep.* 2014;4(1):6561.
- 142. Zhang Y, Dong S, Wang H, et al. Biological impact of environmental polycyclic aromatic hydrocarbons (ePAHs) as endocrine disruptors. *Environ. Pollut.* 2016;213:809–824.
- 143. deCastro BR, Korrick SA, Spengler JD, et al. Estrogenic Activity of Polychlorinated Biphenyls Present in Human Tissue and the Environment. *Environ. Sci. Technol.* 2006;40(8):2819–2825.
- Sheikh IA. Endocrine-disrupting potential of polybrominated diphenyl ethers (PBDEs) on androgen receptor signaling: a structural insight. *Struct. Chem.* [electronic article].
 2020;(https://doi.org/10.1007/s11224-020-01664-z). (Accessed January 28, 2021)
- 145. Cantón RF, Scholten DEA, Marsh G, et al. Inhibition of human placental aromatase activity by hydroxylated polybrominated diphenyl ethers (OH-PBDEs). *Toxicol. Appl. Pharmacol.* 2008;227(1):68–75.
- 146. Zoeller TR. Environmental chemicals targeting thyroid. Hormones. 2010;9(1):28-40.
- 147. Dishaw LV, J Macaulay L, Roberts SC, et al. Exposures, mechanisms, and impacts of endocrineactive flame retardants. *Curr. Opin. Pharmacol.* 2014;19:125–133.
- 148. Godswill AC, Godspel AC. Physiological Effects of Plastic Wastes on the Endocrine System (Bisphenol A, Phthalates, Bisphenol S, PBDEs, TBBPA). 2019;20.
- 149. Szabo DT, Richardson VM, Ross DG, et al. Effects of Perinatal PBDE Exposure on Hepatic Phase I, Phase II, Phase III, and Deiodinase 1 Gene Expression Involved in Thyroid Hormone Metabolism in Male Rat Pups. *Toxicol. Sci.* 2009;107(1):27–39.
- Wolf K, Popp A, Schneider A, et al. Association Between Long-term Exposure to Air Pollution and Biomarkers Related to Insulin Resistance, Subclinical Inflammation, and Adipokines. *Diabetes*. 2016;65(11):3314–3326.
- 151. Krämer U, Herder C, Sugiri D, et al. Traffic-Related Air Pollution and Incident Type 2 Diabetes: Results from the SALIA Cohort Study. *Environ. Health Perspect.* 2010;118(9):1273–1279.

- 152. Carré J, Gatimel N, Moreau J, et al. Does air pollution play a role in infertility?: a systematic review. *Environ. Health*. 2017;16(1):82.
- 153. Janssen BG, Saenen ND, Roels HA, et al. Fetal Thyroid Function, Birth Weight, and in Utero Exposure to Fine Particle Air Pollution: A Birth Cohort Study. *Environ. Health Perspect.* 2017;125(4):699–705.
- Large C, Wei Y. Geographic variations in female breast cancer incidence in relation to ambient air emissions of polycyclic aromatic hydrocarbons. *Environ. Sci. Pollut. Res.* 2017;24(21):17874– 17880.
- 155. Pedersen M, Giorgis-Allemand L, Bernard C, et al. Ambient air pollution and low birthweight: a European cohort study (ESCAPE). *Lancet Respir. Med.* 2013;1(9):695–704.
- 156. Shmuel S, White AJ, Sandler DP. Residential exposure to vehicular traffic-related air pollution during childhood and breast cancer risk. *Environ. Res.* 2017;159:257–263.
- 157. Darbre PD. Overview of air pollution and endocrine disorders. *Int. J. Gen. Med.* 2018;11:191–207.
- 158. Kalkbrenner AE, Daniels JL, Chen J-C, et al. Perinatal Exposure to Hazardous Air Pollutants and Autism Spectrum Disorders at Age 8. *Epidemiol. Camb. Mass.* 2010;21(5):631–641.
- 159. Roberts AL, Lyall Kristen, Hart Jaime E., et al. Perinatal Air Pollutant Exposures and Autism Spectrum Disorder in the Children of Nurses' Health Study II Participants. *Environ. Health Perspect.* 2013;121(8):978–984.
- 160. von Ehrenstein OS, Aralis H, Cockburn M, et al. In Utero Exposure to Toxic Air Pollutants and Risk of Childhood Autism. *Epidemiol. Camb. Mass.* 2014;25(6):851–858.
- 161. Peterson BS, Rauh VA, Bansal R, et al. Effects of Prenatal Exposure to Air Pollutants (Polycyclic Aromatic Hydrocarbons) on the Development of Brain White Matter, Cognition, and Behavior in Later Childhood. *JAMA Psychiatry*. 2015;72(6):531–540.
- 162. Liang H, Vuong AM, Xie C, et al. Childhood polybrominated diphenyl ether (PBDE) serum concentration and reading ability at ages 5 and 8 years: The HOME Study. *Environ. Int.* 2019;122:330–339.
- 163. Caspersen IH, Haugen M, Schjølberg S, et al. Maternal dietary exposure to dioxins and polychlorinated biphenyls (PCBs) is associated with language delay in 3 year old Norwegian children. *Environ. Int.* 2016;91:180–187.
- 164. Cao X, Li J, Cheng L, et al. The associations between prenatal exposure to polycyclic aromatic hydrocarbon metabolites, umbilical cord blood mitochondrial DNA copy number, and children's neurobehavioral development. *Environ. Pollut.* 2020;265:114594.
- 165. Chiu K-C, Sisca F, Ying J-H, et al. Prenatal chlorpyrifos exposure in association with PPARγ H3K4me3 and DNA methylation levels and child development. *Environ. Pollut.* 2021;274:116511.

- 166. Sturza J, Silver MK, Xu L, et al. Prenatal exposure to multiple pesticides is associated with auditory brainstem response at 9months in a cohort study of Chinese infants. *Environ. Int.* 2016;92–93:478–485.
- 167. Braun JM, Gennings Chris, Hauser Russ, et al. What Can Epidemiological Studies Tell Us about the Impact of Chemical Mixtures on Human Health? *Environ. Health Perspect.* 2016;124(1):A6–A9.
- 168. Taylor KW, Joubert Bonnie R., Braun Joe M., et al. Statistical Approaches for Assessing Health Effects of Environmental Chemical Mixtures in Epidemiology: Lessons from an Innovative Workshop. *Environ. Health Perspect.* 2016;124(12):A227–A229.
- 169. Nguena Nguefack HL, Pagé MG, Katz J, et al. Trajectory Modelling Techniques Useful to Epidemiological Research: A Comparative Narrative Review of Approaches. *Clin. Epidemiol.* 2020;12:1205–1222.
- 170. Gonzalez-Casanova I, Stein AD, Barraza-Villarreal A, et al. Prenatal exposure to environmental pollutants and child development trajectories through 7 years. *Int. J. Hyg. Environ. Health*. 2018;221(4):616–622.
- 171. Boucher J. Language development in autism. Int. Congr. Ser. 2003;1254:247–253.
- 172. Gernsbacher MA, Morson EM, Grace EJ. Chapter 70 Language Development in Autism. In: Hickok G, Small SL, eds. *Neurobiology of Language*. San Diego: Academic Press; 2016 (Accessed April 26, 2023):879– 886.(https://www.sciencedirect.com/science/article/pii/B9780124077942000705). (Accessed April 26, 2023)
- 173. Hertz-Picciotto I, Schmidt RJ, Walker CK, et al. A Prospective Study of Environmental Exposures and Early Biomarkers in Autism Spectrum Disorder: Design, Protocols, and Preliminary Data from the MARBLES Study. *Environ. Health Perspect.* 2018;126(11):117004.
- 174. Newschaffer CJ, Croen LA, Fallin MD, et al. Infant siblings and the investigation of autism risk factors. *J. Neurodev. Disord.* 2012;4(1):7.
- 175. Mullen EM. Mullen scales of early learning. AGS Circle Pines, MN; 1995.
- 176. Lord C, Risi S, Lambrecht L, et al. The Autism Diagnostic Observation Schedule Generic: A Standard Measure of Social and Communication Deficits Associated with the Spectrum of Autism. J. Autism Dev. Disord. 2000;30(3):205–223.
- 177. Ozonoff S, Young GS, Belding A, et al. The Broader Autism Phenotype in Infancy: When Does It Emerge? J. Am. Acad. Child Adolesc. Psychiatry. 2014;53(4):398-407.e2.
- 178. Buuren S van, Groothuis-Oudshoorn K. mice: Multivariate Imputation by Chained Equations in R. *J. Stat. Softw.* 2011;45(1):1–67.
- 179. Jung T, Wickrama K a. S. An Introduction to Latent Class Growth Analysis and Growth Mixture Modeling. *Soc. Personal. Psychol. Compass.* 2008;2(1):302–317.

- 180. Berlin KS, Parra GR, Williams NA. An Introduction to Latent Variable Mixture Modeling (Part 2): Longitudinal Latent Class Growth Analysis and Growth Mixture Models. J. Pediatr. Psychol. 2014;39(2):188–203.
- 181. Diallo TMO, Morin AJS, Lu H. The impact of total and partial inclusion or exclusion of active and inactive time invariant covariates in growth mixture models. *Psychol. Methods*. 2016;22(1):166.
- Proust-Lima C, Philipps V, Liquet B. Estimation of extended mixed models using latent classes and latent processes: the R package lcmm. *J. Stat. Softw.* [electronic article]. 2017;78(2). (http://arxiv.org/abs/1503.00890). (Accessed January 10, 2021)
- 183. Tek S, Mesite L, Fein D, et al. Longitudinal Analyses of Expressive Language Development Reveal Two Distinct Language Profiles Among Young Children with Autism Spectrum Disorders. J. Autism Dev. Disord. 2014;44(1):75–89.
- 184. Visser JC, Rommelse NNJ, Lappenschaar M, et al. Variation in the Early Trajectories of Autism Symptoms Is Related to the Development of Language, Cognition, and Behavior Problems. J. Am. Acad. Child Adolesc. Psychiatry. 2017;56(8):659–668.
- 185. Nordahl-Hansen A, Kaale A, Ulvund SE. Language assessment in children with autism spectrum disorder: Concurrent validity between report-based assessments and direct tests. *Res. Autism Spectr. Disord.* 2014;8(9):1100–1106.
- 186. Swineford LB, Guthrie W, Thurm A. Convergent and divergent validity of the Mullen Scales of Early Learning in young children with and without autism spectrum disorder. *Psychol. Assess.* 2015;27(4):1364–1378.
- 187. Mondal N, Bhat BV, Plakkal N, et al. Prevalence and Risk Factors of Speech and Language Delay in Children Less Than Three Years of Age. J. Compr. Pediatr. [electronic article]. 2016;7(2). (https://brieflands.com/articles/jcp-19837#abstract). (Accessed March 25, 2024)
- Zubrick SR, Taylor CL, Rice ML, et al. Late Language Emergence at 24 Months: An Epidemiological Study of Prevalence, Predictors, and Covariates. J. Speech Lang. Hear. Res. 2007;50(6):1562– 1592.
- 189. Chien M, Chen M, Tsai M, et al. Prenatal exposure to environmental heavy metals and language development in toddlers. *Environ. Epidemiol.* 2019;3:68.
- 190. Dzwilewski KLC, Schantz SL. Prenatal chemical exposures and child language development. *J. Commun. Disord.* 2015;57:41–65.
- 191. Herbstman JB, Sjödin A, Kurzon M, et al. Prenatal Exposure to PBDEs and Neurodevelopment. *Environ. Health Perspect.* 2010;118(5):712–719.
- Vejrup K, Schjølberg S, Knutsen HK, et al. Prenatal methylmercury exposure and language delay at three years of age in the Norwegian Mother and Child Cohort Study. *Environ. Int.* 2016;92–93:63– 69.

- 193. National Research Council (US) Committee on the Health Risks of Phthalates. Phthalate Exposure Assessment in Humans. In: *Phthalates and Cumulative Risk Assessment: The Tasks Ahead*. National Academies Press (US); 2008 (Accessed March 25, 2024)(https://www.ncbi.nlm.nih.gov/books/NBK215044/). (Accessed March 25, 2024)
- 194. Schettler T. Human exposure to phthalates via consumer products. *Int. J. Androl.* 2006;29(1):134– 139.
- 195. Patti MA, Newschaffer C, Eliot M, et al. Gestational Exposure to Phthalates and Social Responsiveness Scores in Children Using Quantile Regression: The EARLI and HOME Studies. *Int. J. Environ. Res. Public. Health*. 2021;18(3):1254.
- 196. Shin H-M, Schmidt RJ, Tancredi D, et al. Prenatal exposure to phthalates and autism spectrum disorder in the MARBLES study. *Environ. Health*. 2018;17(1):85.
- Chiu Y-H, Bellavia A, James-Todd T, et al. Evaluating effects of prenatal exposure to phthalate mixtures on birth weight: A comparison of three statistical approaches. *Environ. Int.* 2018;113:231–239.
- 198. Kamata A, Kara Y, Patarapichayatham C, et al. Evaluation of Analysis Approaches for Latent Class Analysis with Auxiliary Linear Growth Model. *Front. Psychol.* [electronic article]. 2018;9. (https://www.frontiersin.org/articles/10.3389/fpsyg.2018.00130/full). (Accessed June 18, 2021)
- 199. Clark S, Muthén B. Latent Class Analysis Results to Variables not Included in the Analysis. 2009;(http://www.statmodel.com/download/relatinglca.pdf)
- 200. Ghio M, Cara C, Tettamanti M. The prenatal brain readiness for speech processing: A review on foetal development of auditory and primordial language networks. *Neurosci. Biobehav. Rev.* 2021;128:709–719.
- 201. Nedelcuta RM, Calin G, Baleanu VD, et al. Broca's Area and Language Disorders in the Child. *Rev. Chim.* 2019;70(8):3014–3016.
- 202. Domínguez-Romero E, Scheringer M. A review of phthalate pharmacokinetics in human and rat: what factors drive phthalate distribution and partitioning? *Drug Metab. Rev.* 2019;51(3):314–329.
- 203. Zhang Y-J, Guo J-L, Xue J, et al. Phthalate metabolites: Characterization, toxicities, global distribution, and exposure assessment. *Environ. Pollut.* 2021;291:118106.
- 204. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological profile for Di-n-butyl Phthalate. 2001;(https://www.atsdr.cdc.gov/ToxProfiles/tp135.pdf). (Accessed March 17, 2021)
- 205. Miodovnik A, Engel SM, Zhu C, et al. Endocrine disruptors and childhood social impairment. *NeuroToxicology*. 2011;32(2):261–267.
- 206. Ramos AM, Herring AH, Villanger GD, et al. The association of prenatal phthalates, organophosphorous pesticides, and organophosphate esters with early child language ability in Norway. *Environ. Res.* 2023;225:115508.

- 207. Wang Y, Zhu H, Kannan K. A Review of Biomonitoring of Phthalate Exposures. *Toxics*. 2019;7(2):21.
- 208. Carrico C, Gennings C, Wheeler DC, et al. Characterization of Weighted Quantile Sum Regression for Highly Correlated Data in a Risk Analysis Setting. *J. Agric. Biol. Environ. Stat.* 2015;20(1):100–120.
- 209. Jankowska A, Nazareth L, Kaleta D, et al. Review of the Existing Evidence for Sex-Specific Relationships between Prenatal Phthalate Exposure and Children's Neurodevelopment. *Int. J. Environ. Res. Public. Health*. 2021;18(24):13013.
- 210. Liao Y-C, Xu Y-J, Chen J-K, et al. Sex differences in children's cognitive functions and phthalates exposure: a meta-analysis. *Pediatr. Res.* 2023;94(5):1609–1618.
- 211. Hlisníková H, Petrovičová I, Kolena B, et al. Effects and Mechanisms of Phthalates' Action on Reproductive Processes and Reproductive Health: A Literature Review. *Int. J. Environ. Res. Public. Health*. 2020;17(18):6811.
- 212. Xiong H, Liu X, Yang F, et al. Developmental Language Differences in Children with Autism Spectrum Disorders and Possible Sex Difference. J. Autism Dev. Disord. 2024;54(3):841–851.
- 213. Carter AS, Black DO, Tewani S, et al. Sex Differences in Toddlers with Autism Spectrum Disorders. J. Autism Dev. Disord. 2007;37(1):86–97.
- 214. Zhang Y, Qin B, Wang L, et al. Sex differences of language abilities of preschool children with autism spectrum disorder and their anatomical correlation with Broca and Wernicke areas. *Front. Pediatr.* 2022;10:762621.
- 215. Saure E, Castrén M, Mikkola K, et al. Intellectual disabilities moderate sex/gender differences in autism spectrum disorder: a systematic review and meta-analysis. *J. Intellect. Disabil. Res.* 2023;67(1):1–34.
- 216. Dillon EF, Kanne S, Landa RJ, et al. Sex Differences in Autism: Examining Intrinsic and Extrinsic Factors in Children and Adolescents Enrolled in a National ASD Cohort. J. Autism Dev. Disord. 2023;53(4):1305–1318.
- Saleem A, Awan T, Akhtar MF. A comprehensive review on endocrine toxicity of gaseous components and particulate matter in smog. *Front. Endocrinol.* [electronic article]. 2024;15. (https://www.frontiersin.org/journals/endocrinology/articles/10.3389/fendo.2024.1294205/full). (Accessed April 13, 2024)
- Wei C, Cao L, Zhou Y, et al. Multiple statistical models reveal specific volatile organic compounds affect sex hormones in American adult male: NHANES 2013–2016. Front. Endocrinol. [electronic article]. 2023;13.
 (https://www.frontiersin.org/journals/endocrinology/articles/10.3389/fendo.2022.1076664/full). (Accessed April 9, 2024)

- 219. Wei C, Pan Y, Zhang W, et al. Comprehensive analysis between volatile organic compound (VOC) exposure and female sex hormones: a cross-sectional study from NHANES 2013–2016. *Environ. Sci. Pollut. Res.* 2023;30(42):95828–95839.
- 220. Davis AN, Carlo G, Gulseven Z, et al. Exposure to environmental toxicants and young children's cognitive and social development. *Rev. Environ. Health*. 2019;34(1):35–56.
- 221. Suades-González E, Gascon M, Guxens M, et al. Air Pollution and Neuropsychological Development: A Review of the Latest Evidence. *Endocrinology*. 2015;156(10):3473–3482.
- 222. Guxens M, Garcia-Esteban R, Giorgis-Allemand L, et al. Air Pollution During Pregnancy and Childhood Cognitive and Psychomotor Development: Six European Birth Cohorts. *Epidemiology*. 2014;25(5):636.
- 223. Stingone JA, McVeigh KH, Claudio L. Association between prenatal exposure to ambient diesel particulate matter and perchloroethylene with children's 3rd grade standardized test scores. *Environ. Res.* 2016;148:144–153.
- 224. Freire C, Ramos R, Puertas R, et al. Association of traffic-related air pollution with cognitive development in children. *J. Epidemiol. Community Health*. 2010;64(3):223–228.
- 225. Guxens M, Aguilera I, Ballester F, et al. Prenatal Exposure to Residential Air Pollution and Infant Mental Development: Modulation by Antioxidants and Detoxification Factors. *Environ. Health Perspect.* 2012;120(1):144–149.
- 226. Volk HE, Perera F, Braun JM, et al. Prenatal air pollution exposure and neurodevelopment: A review and blueprint for a harmonized approach within ECHO. *Environ. Res.* 2021;196:110320.
- 227. US EPA O. National Emissions Inventory (NEI). 2015;(https://www.epa.gov/air-emissions-inventories/national-emissions-inventory-nei). (Accessed April 9, 2024)
- 228. Scheffe RD, Strum M, Phillips SB, et al. Hybrid Modeling Approach to Estimate Exposures of Hazardous Air Pollutants (HAPs) for the National Air Toxics Assessment (NATA). *Environ. Sci. Technol.* 2016;50(22):12356–12364.
- USEPA. Technical Support Document EPA's 2014 National Air Toxics Assessment.
 2018;(https://www.epa.gov/sites/default/files/2018-09/documents/2014_nata_technical_support_document.pdf). (Accessed March 30, 2024)
- George BJ, Schultz BD, Palma T, et al. An evaluation of EPA's National-Scale Air Toxics Assessment (NATA): Comparison with benzene measurements in Detroit, Michigan. *Atmos. Environ.* 2011;45(19):3301–3308.
- 231. Logue JM, Small MJ, Robinson AL. Evaluating the national air toxics assessment (NATA): Comparison of predicted and measured air toxics concentrations, risks, and sources in Pittsburgh, Pennsylvania. *Atmos. Environ.* 2011;45(2):476–484.

- 232. Xue Z, Jia C. A Model-to-Monitor Evaluation of 2011 National-Scale Air Toxics Assessment (NATA). *Toxics* [electronic article]. 2019;7(1). (https://www.ncbi.nlm.nih.gov/pmc/articles/PMC6468659/). (Accessed March 17, 2021)
- 233. US EPA O. 2011 NATA: Assessment Results. 2015;(https://www.epa.gov/national-air-toxicsassessment/2011-nata-assessment-results). (Accessed April 9, 2024)
- 234. US EPA O. 2014 NATA: Assessment Methods. US EPA. 2018;(https://www.epa.gov/national-air-toxics-assessment/2014-nata-assessment-methods). (Accessed March 8, 2021)
- 235. Shimodaira H. An approximately unbiased test of phylogenetic tree selection. *Syst. Biol.* 2002;51(3):492–508.
- 236. Benjamini Y, Hochberg Y. Controlling the False Discovery Rate: A Practical and Powerful Approach to Multiple Testing. J. R. Stat. Soc. Ser. B Methodol. 1995;57(1):289–300.
- 237. Bobb JF, Valeri L, Claus Henn B, et al. Bayesian kernel machine regression for estimating the health effects of multi-pollutant mixtures. *Biostat. Oxf. Engl.* 2015;16(3):493–508.
- Bobb JF, Claus Henn B, Valeri L, et al. Statistical software for analyzing the health effects of multiple concurrent exposures via Bayesian kernel machine regression. *Environ. Health*. 2018;17(1):67.
- 239. Madaniyazi L, Jung C-R, Fook Sheng Ng C, et al. Early life exposure to indoor air pollutants and the risk of neurodevelopmental delays: The Japan Environment and Children's Study. *Environ. Int.* 2022;158:107004.
- 240. Hou S, Yuan L, Jin P, et al. A clinical study of the effects of lead poisoning on the intelligence and neurobehavioral abilities of children. *Theor. Biol. Med. Model.* 2013;10(1):13.
- 241. Lin C-C, Chen Y-C, Su F-C, et al. *In utero* exposure to environmental lead and manganese and neurodevelopment at 2 years of age. *Environ. Res.* 2013;123:52–57.
- 242. Campbell TF, Needleman HL, Riess JA, et al. Bone Lead Levels and Language Processing Performance. *Dev. Neuropsychol.* 2000;18(2):171–186.
- 243. Grandjean P, Landrigan P. Developmental neurotoxicity of industrial chemicals. *The Lancet*. 2006;368(9553):2167–2178.
- 244. Kawano T, Matsuse H, Fukahori S, et al. Acetaldehyde at a Low Concentration Synergistically Exacerbates Allergic Airway Inflammation as an Endocrine-Disrupting Chemical and as a Volatile Organic Compound. *Respiration*. 2012;84(2):135–141.
- 245. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Cresols. 2008;283.
- 246. Reutman SR, LeMasters GK, Knecht EA, et al. Evidence of reproductive endocrine effects in women with occupational fuel and solvent exposures. *Environ. Health Perspect.* 2002;110(8):805–811.

- 247. Bolden AL, Schultz K, Pelch KE, et al. Exploring the endocrine activity of air pollutants associated with unconventional oil and gas extraction. *Environ. Health*. 2018;17(1):26.
- 248. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Methyl tert-Butyl Ether. 1996;(https://www.atsdr.cdc.gov/ToxProfiles/tp91.pdf). (Accessed March 18, 2021)
- 249. Yang Y, Zhang Z, Zhang H, et al. Effects of maternal acrolein exposure during pregnancy on testicular testosterone production in fetal rats. *Mol. Med. Rep.* 2017;16(1):491–498.
- 250. Jeelani R, Aldhaheri SR, Mikhael S, et al. Acrolein: a common toxin impacting reproductive potential Fertility and Sterility. 2017;(https://www.fertstert.org/article/S0015-0282(17)31465-6/fulltext). (Accessed March 17, 2021)
- 251. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Ethylene Oxide. 2020;246.
- Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Methylenedianiline. 1998;(https://www.atsdr.cdc.gov/ToxProfiles/tp122.pdf). (Accessed March 18, 2021)
- 253. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological profile for Acrylamide. 2012;
- 254. Mutic AD, Baker BJ, McCauley LA. Deleterious Effects From Occupational Exposure to Ethylene Thiourea in Pregnant Women. *Workplace Health Saf.* 2017;65(12):595–602.
- 255. Teles M, Pacheco M, Santos MA. Anguilla anguilla L. liver ethoxyresorufin O-deethylation, glutathione S-tranferase, erythrocytic nuclear abnormalities, and endocrine responses to naphthalene and β-naphthoflavone. *Ecotoxicol. Environ. Saf.* 2003;55(1):98–107.
- 256. Yumin Z, Mingyue M, Haishan L, et al. Effects of Acrylonitrile on Reproductive and Endocrine System of Male Rats. *Chin. J. Ind. Med.* 1999;01.
- 257. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for 2,4-Dichlorophenoxyacetic acid. 2020;255.
- 258. Davis LK, Visitacion N, Riley LG, et al. Effects of o,p'-DDE, heptachlor, and 17β-estradiol on vitellogenin gene expression and the growth hormone/insulin-like growth factor-I axis in the tilapia, Oreochromis mossambicus. *Comp. Biochem. Physiol. Part C Toxicol. Pharmacol.* 2009;149(4):507–514.
- 259. Snyder MJ, Mulder EP. Environmental endocrine disruption in decapod crustacean larvae: hormone titers, cytochrome P450, and stress protein responses to heptachlor exposure. *Aquat. Toxicol.* 2001;55(3):177–190.
- 260. Oladele JO, Oyeleke OM, Oladele OT, et al. Nitrobenzene-induced hormonal disruption, alteration of steroidogenic pathway, and oxidative damage in rat: protective effects of Vernonia amygdalina. *Clin. Phytoscience*. 2020;6(1):15.
- 261. Kumar V, Majumdar C, Roy P. Effects of endocrine disrupting chemicals from leather industry effluents on male reproductive system. *J. Steroid Biochem. Mol. Biol.* 2008;111(3):208–216.
- 262. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for DDT, DDE, DDD. 2019;486.
- 263. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Hexachlorobenzene. 2015;432.
- 264. Li C, Taneda S, Suzuki AK, et al. Estrogenic and anti-androgenic activities of 4-nitrophenol in diesel exhaust particles. *Toxicol. Appl. Pharmacol.* 2006;217(1):1–6.
- 265. Zhang Y, Piao Y, Li Y, et al. 4-Nitrophenol induces Leydig cells hyperplasia, which may contribute to the differential modulation of the androgen receptor and estrogen receptor- α and - β expression in male rat testes. *Toxicol. Lett.* 2013;223(2):228–235.
- 266. Huda Bhuiyan MN, Kang H, Kim JH, et al. Endocrine disruption by several aniline derivatives and related mechanisms in a human adrenal H295R cell line and adult male zebrafish. *Ecotoxicol. Environ. Saf.* 2019;180:326–332.
- 267. Hsu P-C, Lai T-J, Guo N-W, et al. Serum Hormones in Boys Prenatally Exposed to Polychlorinated Biphenyls and Dibenzofurans. *J. Toxicol. Environ. Health A*. 2005;68(17–18):1447–1456.
- 268. Wang Shu-Li, Chang Yu-Chen, Chao How-Ran, et al. Body Burdens of Polychlorinated Dibenzo-pdioxins, Dibenzofurans, and Biphenyls and Their Relations to Estrogen Metabolism in Pregnant Women. *Environ. Health Perspect.* 2006;114(5):740–745.
- 269. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological profile for Hexachlorocyclopentadiene.
 1999;(http://www.crcnetbase.com/doi/book/10.1201/9781420061888). (Accessed March 18, 2021)
- 270. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for 1,2-Dibromoethane. 2018;175.
- 271. Yu C, Wang C, Lu Z, et al. The endocrine-disrupting potential of four chlorophenols by in vitro and in silico assay. *Chemosphere*. 2019;218:941–947.
- 272. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for 1,2-Dibromo-3-Chloropropane. 2018;155.
- Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological profile for Hexachloroethane. 1997;(https://www.atsdr.cdc.gov/ToxProfiles/tp97.pdf). (Accessed March 18, 2021)
- 274. Moore MA, Weber E, Mayer D, et al. Adrenal cortical foci and nodules in Sprague-Dawley rats treated with N-nitrosomorpholine. *Virchows Arch. B.* 1989;57(1):91.

- 275. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological profile for Di(2ethylhexyl)phthalate (DEHP). 2019;(https://www.atsdr.cdc.gov/ToxProfiles/tp9-c2.pdf). (Accessed March 17, 2021)
- 276. Johnson KJ, McDowell EN, Viereck MP, et al. Species-Specific Dibutyl Phthalate Fetal Testis Endocrine Disruption Correlates with Inhibition of SREBP2-Dependent Gene Expression Pathways. *Toxicol. Sci.* 2011;120(2):460–474.
- 277. Ohtani H, Miura I, Ichikawa Y. Effects of dibutyl phthalate as an environmental endocrine disruptor on gonadal sex differentiation of genetic males of the frog Rana rugosa. *Environ. Health Perspect.* 2000;108(12):1189–1193.
- 278. Ruiz-García L, Figueroa-Vega N, Malacara JM, et al. Possible role of n-hexane as an endocrine disruptor in occupationally exposed women at reproductive age. *Toxicol. Lett.* 2020;330:73–79.
- 279. Li H, Zhang C, Ni F, et al. Gestational N-hexane inhalation alters the expression of genes related to ovarian hormone production and DNA methylation states in adult female F1 rat offspring. *Toxicol. Lett.* 2015;239(3):141–151.
- 280. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Parathion. 2017;298.
- 281. Sun H, Shen O-X, Xu X-L, et al. Carbaryl, 1-naphthol and 2-naphthol inhibit the beta-1 thyroid hormone receptor-mediated transcription in vitro. *Toxicology*. 2008;249(2):238–242.
- 282. Makita Y. Effects of Perinatal, Combined Exposure to 1,4-Dichlorobenzene and 1,1-Dichloro-2,2bis(p-chlorophenyl)ethylene on Rat Female Reproductive System. *Basic Clin. Pharmacol. Toxicol.* 2008;102(4):360–364.
- 283. ALEYASSINE H, LEE SH. Inhibition by Hydrazine, Phenelzine and Pargyline of Insulin Release from Rat Pancreas. *Endocrinology*. 1971;89(1):125–129.
- 284. HIDAKA H, NAGASAKA A, DEGROOT LJ. Inhibition of Thyroid Iodide Peroxidase by Hydrazines and NSD-1055 (4-Bromo-3-hydroxybenzyloxyamine). *Endocrinology*. 1971;88(5):1264–1266.
- 285. Ashby J, Odum J, Burns A, et al. The reported in vitro anti-estrogen pentachloronitrobenzene enhances the estrogenic activity of estradiol in vivo in the rat. *Environ. Toxicol. Pharmacol.* 2005;20(1):199–208.
- 286. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Carbon Disulfide. 1996;(https://www.atsdr.cdc.gov/ToxProfiles/tp82.pdf). (Accessed March 17, 2021)
- 287. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Dichlorvos. 1997;(https://www.atsdr.cdc.gov/ToxProfiles/tp88.pdf). (Accessed March 17, 2021)
- 288. Huang D-J, Wang S-Y, Chen H-C. Effects of the endocrine disrupter chemicals chlordane and lindane on the male green neon shrimp (Neocaridina denticulata). *Chemosphere*. 2004;57(11):1621–1627.

- Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological profile for Pentachlorophenol. 2001;(https://www.atsdr.cdc.gov/ToxProfiles/tp51.pdf). (Accessed March 18, 2021)
- 290. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Chlordane. 2018;252.
- 291. Turner C, Sawle A, Fenske M, et al. Implications of the solvent vehicles dimethylformamide and dimethylsulfoxide for establishing transcriptomic endpoints in the zebrafish embryo toxicity test. *Environ. Toxicol. Chem.* 2012;31(3):593–604.
- 292. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological profile for Methoxychlor. 2002;(https://www.atsdr.cdc.gov/ToxProfiles/tp47.pdf). (Accessed March 18, 2021)
- 293. Stouder C, Paoloni-Giacobino A. Specific transgenerational imprinting effects of the endocrine disruptor methoxychlor on male gametes. *REPRODUCTION*. 2011;141(2):207–216.
- 294. Tijani JO, Fatoba OO, Babajide OO, et al. Pharmaceuticals, endocrine disruptors, personal care products, nanomaterials and perfluorinated pollutants: a review. *Environ. Chem. Lett.* 2016;14(1):27–49.
- 295. Kabir ER, Rahman MS, Rahman I. A review on endocrine disruptors and their possible impacts on human health. *Environ. Toxicol. Pharmacol.* 2015;40(1):241–258.
- 296. Qureshi MS, Yusoff AR bin M, Wirzal MDH, et al. Methods for the Determination of Endocrine Disrupting Phthalate Esters. *Crit. Rev. Anal. Chem.* 2016;46(2):146–159.
- 297. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Bromomethane. 2020;184.
- 298. Kojima Hiroyuki, Katsura Eiji, Takeuchi Shinji, et al. Screening for estrogen and androgen receptor activities in 200 pesticides by in vitro reporter gene assays using Chinese hamster ovary cells. *Environ. Health Perspect.* 2004;112(5):524–531.
- 299. Kojima H, Takeuchi S, Nagai T. Endocrine-disrupting Potential of Pesticides via Nuclear Receptors and Aryl Hydrocarbon Receptor. *J. Health Sci.* 2010;56(4):374–386.
- 300. Lee W-K, Lee K-W, Kwak E-J, et al. Effects of environmental endocrine disruptors on the sex differentiation in Korean rockfish, Sebastes schlegeli. *Water Sci. Technol.* 2003;47(9):65–70.
- 301. CASTOR CW, BEIERWALTES W, DORSTEWITZ E, et al. EFFECT OF 2,4-DINITROPHENOL ON THYROID FUNCTION IN MAN*. J. Clin. Endocrinol. Metab. 1956;16(8):1026–1031.
- Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological profile for Chloromethane. 1998;(https://www.atsdr.cdc.gov/ToxProfiles/tp106.pdf). (Accessed March 18, 2021)
- 303. Kenfack A, Akono Nantia E, Ngoula F, et al. Post Exposure Effects of Propoxur (Agricultural Pesticide) on Male Fertility in Wistar Rat. *Iran. J. Toxicol.* 2018;12(3):21–27.

- 304. Mileson BE, Sweeney LM, Gargas ML, et al. Iodomethane human health risk characterization. *Inhal. Toxicol.* 2009;21(6):583–605.
- 305. Harper C, Roney N, Buser MC, et al. Toxicological profile for 1,2-Dichloropropane : draft for public comment. 2019;(https://stacks.cdc.gov/view/cdc/84088)
- 306. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Styrene. 2010;283.
- 307. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Arsenic. 2007;559.
- 308. Varma SK, Bloch E, Gondos B, et al. Reproductive toxicity of 2,4-toluenediamine in the rat. 3. Effects on androgen-binding protein levels, selected seminiferous tubule characteristics, and spermatogenesis. *J. Toxicol. Environ. Health*. 1988;25(4):435–451.
- 309. Heiden TCK, Struble CA, Rise ML, et al. Molecular targets of 2,3,7,8-tetrachlorodibenzo-p-dioxin (TCDD) within the zebrafish ovary: Insights into TCDD-induced endocrine disruption and reproductive toxicity. *Reprod. Toxicol.* 2008;25(1):47–57.
- 310. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Chromium. 2012;592.
- 311. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Toxaphene. 2014;291.
- 312. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Toluene. 2017;496.
- 313. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Cobalt. 2004;(https://www.atsdr.cdc.gov/ToxProfiles/tp33.pdf). (Accessed March 18, 2021)
- 314. Valkusz Z, Nagyéri G, Radács M, et al. Further analysis of behavioral and endocrine consequences of chronic exposure of male Wistar rats to subtoxic doses of endocrine disruptor chlorobenzenes. *Physiol. Behav.* 2011;103(5):421–430.
- 315. Nagyeri G, Valkusz Z, Radacs M, et al. Behavioral and endocrine effects of chronic exposure to low doses of chlorobenzenes in Wistar rats. *Neurotoxicol. Teratol.* 2012;34(1):9–19.
- 316. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Cyanide. 2006;341.
- 317. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Trichloroethylene. 2019;511.
- 318. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Mercury. 1999;(https://www.atsdr.cdc.gov/ToxProfiles/tp46.pdf). (Accessed March 18, 2021)

- 319. Waldbillig NCRSJCD. Effects of the Pesticides Carbofuran, Chlorpyrifos, Dimethoate, Lindane, Triallate, Trifluralin, 2,4-D, and Pentachlorophenol on the Metabolic Endocrine and Reproductive Endocrine System in Ewes. J. Toxicol. Environ. Health A. 1998;54(1):21–36.
- 320. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Lead. 2020;583.
- 321. van den Berg KJ. Interaction of chlorinated phenols with thyroxine binding sites of human transthyretin, albumin and thyroid binding globulin. *Chem. Biol. Interact.* 1990;76(1):63–75.
- 322. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Nickel. 2005;397.
- 323. Lijinsky W, Reuber MD. Chronic toxicity studies of vinyl acetate in Fischer rats. *Toxicol. Appl. Pharmacol.* 1983;68(1):43–53.
- 324. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Manganese. 2012;556.
- 325. Chen W, Hillyer JF. FlyNap (Triethylamine) Increases the Heart Rate of Mosquitoes and Eliminates the Cardioacceleratory Effect of the Neuropeptide CCAP. *PLoS ONE*. 2013;8(7):e70414.
- 326. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Selenium. 2003;(https://www.atsdr.cdc.gov/ToxProfiles/tp92.pdf). (Accessed March 18, 2021)
- 327. Agency for Toxic Substances and Disease Registry (ATSDR). Toxicological Profile for Vinyl Chloride. 2006;(https://www.atsdr.cdc.gov/ToxProfiles/tp20.pdf). (Accessed March 18, 2021)
- 328. Alofe O, Kisanga E, Inayat-Hussain SH, et al. Determining the endocrine disruption potential of industrial chemicals using an integrative approach: Public databases, in vitro exposure, and modeling receptor interactions. *Environ. Int.* 2019;131:104969.
- 329. Wallace ER, Ni Y, Loftus CT, et al. Prenatal urinary metabolites of polycyclic aromatic hydrocarbons and toddler cognition, language, and behavior. *Environ. Int.* 2022;159:107039.
- 330. Wheeler DC, Rustom S, Carli M, et al. Bayesian Group Index Regression for Modeling Chemical Mixtures and Cancer Risk. *Int. J. Environ. Res. Public. Health*. 2021;18(7):3486.
- Li H, Deng W, Small R, et al. Health effects of air pollutant mixtures on overall mortality among the elderly population using Bayesian kernel machine regression (BKMR). *Chemosphere*. 2022;286:131566.
- 332. Barkoski JM, Philippat C, Tancredi D, et al. In utero pyrethroid pesticide exposure in relation to autism spectrum disorder (ASD) and other neurodevelopmental outcomes at 3 years in the MARBLES longitudinal cohort. *Environ. Res.* 2021;194:110495.
- 333. Ozonoff S, Young GS, Carter A, et al. Recurrence Risk for Autism Spectrum Disorders: A Baby Siblings Research Consortium Study. *Pediatrics*. 2011;128(3):e488–e495.

- Georgiades S, Szatmari P, Zwaigenbaum L, et al. A Prospective Study of Autistic-Like Traits in Unaffected Siblings of Probands With Autism Spectrum Disorder. JAMA Psychiatry. 2013;70(1):42–48.
- 335. Li Q-Q, Huang J, Cai D, et al. Prenatal Exposure to Legacy and Alternative Per- and Polyfluoroalkyl Substances and Neuropsychological Development Trajectories over the First 3 Years of Life. *Environ. Sci. Technol.* 2023;57(9):3746–3757.
- 336. Gao Y, Luo J, Zhang Y, et al. Prenatal Exposure to Per- and Polyfluoroalkyl Substances and Child Growth Trajectories in the First Two Years. *Environ. Health Perspect.* 2022;130(3):037006.
- 337. Braun JM, Yolton K, Stacy SL, et al. Prenatal Environmental Chemical Exposures and Longitudinal Patterns of Child Neurobehavior. *Neurotoxicology*. 2017;62:192–199.
- 338. Dewey D, Martin JW, MacDonald AM, et al. Sex-specific associations between maternal phthalate exposure and neurodevelopmental outcomes in children at 2 years of age in the APrON cohort. *NeuroToxicology*. 2023;98:48–60.
- 339. Johns LE, Cooper GS, Galizia A, et al. Exposure assessment issues in epidemiology studies of phthalates. *Environ. Int.* 2015;85:27–39.
- Michelotti J, Charman T, Slonims V, et al. Follow-up of children with language delay and features of autism from preschool years to middle childhood. *Dev. Med. Child Neurol.* 2002;44(12):812–819.
- 341. Miniscalco C, Nygren G, Hagberg B, et al. Neuropsychiatric and neurodevelopmental outcome of children at age 6 and 7 years who screened positive for language problems at 30 months. *Dev. Med. Child Neurol.* 2006;48(5):361–366.
- 342. Haney JL. Autism, females, and the DSM-5: Gender bias in autism diagnosis. *Soc. Work Ment. Health*. 2016;14(4):396–407.
- 343. Barua A, Singla K, Narwat P, et al. Gender bias in autism spectrum disorders-a review. *J. Clin. Diagn. Res.* 2018;12(4):KE01–KE07.

Appendix

Supplemental Table 1: Median and Interquartile Ranges of Phthalate Metabolites (ng/mL) by Child Characteristics for Included Individuals – File Supplemental 1

Supplemental Table 2: AIC for Transformations of Phthalates – File Supplemental 2